UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 27, 2021

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from sition period from ______ to ___ Commission File number <u>1-9273</u>

PILGRIM'S PRIDE CORPORATION

Delaware (State or other jurisdiction of incorporation or organization)

X

П

1770 Promontory Circle Greeley CO

75-1285071 (I.R.S. Employer Identification No.)

> 80634-9038 (Zip code)

(Address of principal executive offices)

Registrant's telephone number, including area code: (970) 506-8000

Securities registered pursuant to Section 12(b) of the Act: Trading Symbol
PPC Name of Exchange on which Registered
The Nasdaq Stock Market LLC Title of each class on Stock, Par Value \$0.01

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes $\ oxin{2}$ No $\ oxin{2}$

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \boxtimes No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "smaller growth Rule "accelerated reporting company," "emerging company" Exchange X

Large Accelerated Filer Accelerated Filer Non-accelerated Filer Smaller reporting company Emerging growth company

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

 $Number of shares outstanding of the issuer's common stock, \$0.01 \ par value \ per share, as of July 28, 2021, was 243,675,522.$

INDEX

PILGRIM'S PRIDE CORPORATION

<u>PART I. FINANCIAL INFOR</u>	<u>MATION</u>
tem 1.	Condensed Consolidated Financial Statements
	Condensed Consolidated Balance Sheets June 27, 2021 and December 27, 2020
	Condensed Consolidated Statements of Income Three and Six Months Ended June 27, 2021 and June 28, 2020
	Condensed Consolidated Statements of Comprehensive Income Three and Six Months Ended June 27, 2021 and June 28, 2020
	Condensed Consolidated Statements of Stockholders' Equity. Three and Six Months Ended June 27, 2021 and June 28, 2020
	Condensed Consolidated Statements of Cash Flows Six Months Ended June 27, 2021 and June 28, 2020
	Notes to Condensed Consolidated Financial Statements
tem 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations
tem 3.	Quantitative and Qualitative Disclosures about Market Risk
tem 4.	Controls and Procedures
PART II. OTHER INFORMA	<u>TION</u>
tem 1.	<u>Legal Proceedings</u>
tem 1A.	Risk Factors
tem 2.	Unregistered Sales of Equity Securities and Use of Proceeds
tem 6.	Exhibits
SIGNATURES	

PART I. FINANCIAL INFORMATION ITEM 1. CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

PILGRIM'S PRIDE CORPORATION CONDENSED CONSOLIDATED BALANCE SHEETS

	 (Unaudited) June 27, 2021	December 27, 2020
	 (In tho	isands)
Cash and cash equivalents	\$ 391,805	\$ 547,624
Restricted cash and cash equivalents	98,212	782
Trade accounts and other receivables, less allowance for doubtful accounts	866,476	741,992
Accounts receivable from related parties	868	1,084
Inventories	1,530,014	1,358,793
Income taxes receivable	51,351	69,397
Prepaid expenses and other current assets	190,574	183,039
Total current assets	3,129,300	2,902,711
Deferred tax assets	5,494	5,471
Other long-lived assets	26,837	24,780
Operating lease assets, net	295,391	288,886
Identified intangible assets, net	589,536	589,913
Goodwill	1,024,900	1,005,245
Property, plant and equipment, net	2,677,387	2,657,491
Total assets	\$ 7,748,845	\$ 7,474,497
Accounts payable	\$ 1,092,164	\$ 1,028,710
Accounts payable to related parties	8,595	9,650
Revenue contract liabilities	36,275	65,918
Accrued expenses and other current liabilities	1,051,546	807,847
Income taxes payable	30,681	_
Current maturities of long-term debt	25,453	25,455
Total current liabilities	2,244,714	1,937,580
Noncurrent operating lease liability, less current maturities	221,345	217,432
Long-term debt, less current maturities	2,270,298	2,255,546
Deferred tax liabilities	318,159	339,831
Other long-term liabilities	99,817	148,761
Total liabilities	5,154,333	4,899,150
Common stock	2,614	2,612
Treasury stock	(345,134)	(345,134)
Additional paid-in capital	1,959,558	1,954,334
Retained earnings	906,090	972,569
Accumulated other comprehensive income (loss)	59,354	(20,620)
Total Pilgrim's Pride Corporation stockholders' equity	2,582,482	2,563,761
Noncontrolling interest	12,030	11,586
Total stockholders' equity	2,594,512	2,575,347
Total liabilities and stockholders' equity	\$ 7,748,845	\$ 7,474,497

PILGRIM'S PRIDE CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF INCOME (Unaudited)

	Three Mo	nths Ended	Six Months Ended			
	June 27, 2021	June 28, 2020	June 27, 2021	June 28, 2020		
		(in thousands, ex	cept per share data)			
Net sales	\$ 3,637,698	\$ 2,824,023	\$ 6,911,123	\$ 5,898,951		
Cost of sales	3,257,457	2,704,164	6,269,639	5,601,993		
Gross profit	380,241	119,859	641,484	296,958		
Selling, general and administrative expense	503,372	92,570	606,151	185,283		
Operating income (loss)	(123,131)	27,289	35,333	111,675		
Interest expense, net of capitalized interest	50,651	32,323	80,985	65,011		
Interest income	(842)	(1,158)	(3,208)	(2,848)		
Foreign currency transaction loss (gain)	4,145	5,525	6,659	(12,860)		
Miscellaneous, net	(770)	(45)	(8,614)	(34,233)		
Income (loss) before income taxes	(176,315)	(9,356)	(40,489)	96,605		
Income tax expense (benefit)	(9,812)	(2,956)	25,546	35,556		
Net income (loss)	(166,503)	(6,400)	(66,035)	61,049		
Less: Net income (loss) attributable to noncontrolling interests	184	(364)	444	(183)		
Net income (loss) attributable to Pilgrim's Pride Corporation	\$ (166,687)	\$ (6,036)	\$ (66,479)	\$ 61,232		
Weighted average shares of Pilgrim's Pride Corporation common stock						
outstanding:						
Basic	243,675	246,687	243,627	248,017		
Effect of dilutive common stock equivalents	_	_	_	291		
Diluted	243,675	246,687	243,627	248,308		
Net income (loss) attributable to Pilgrim's Pride Corporation per share of common stock outstanding:						
Basic	\$ (0.68)	\$ (0.02)	\$ (0.27)	\$ 0.25		
Diluted	\$ (0.68)	\$ (0.02)	\$ (0.27)	\$ 0.25		

PILGRIM'S PRIDE CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (Unaudited)

	Three	Months Ended	Six Mon	ths Ended
	June 27, 2021	June 28, 2020	June 27, 2021	June 28, 2020
			ousands)	
Net income (loss)	\$ (166,50	3) \$ (6,400)	\$ (66,035)	\$ 61,049
Other comprehensive income (loss):				
Foreign currency translation adjustment:				
Gains (losses) arising during the period	15,84	7 (18,782)	49,138	(115,547)
Derivative financial instruments designated as cash flow hedges:				
Gains (losses) arising during the period	7-	1 (2,147)	2,209	1,901
Income tax effect	1	5 —	32	_
Reclassification to net earnings for losses (gains) realized	(1,08	0) (162)	(1,248)	580
Income tax effect	(4	0) —	(72)	_
Available-for-sale securities:				
Gains arising during the period	-	_ 2	_	14
Income tax effect	-	- (1)	_	(4)
Reclassification to net earnings for gains realized	-	- (12)	_	(12)
Income tax effect	-	- 3	_	3
Defined benefit plans:				
Gains (losses) arising during the period	9,8	6 (34,151)	39,103	(44,961)
Income tax effect	(3,58	8) 6,459	(9,918)	9,164
Reclassification to net earnings of losses realized	38	4 375	955	751
Income tax effect	(9	0) (94)	(225)	(187)
Total other comprehensive income (loss), net of tax	22,07	5 (48,510)	79,974	(148,298)
Comprehensive income (loss)	(144,42	8) (54,910)	13,939	(87,249)
Less: Comprehensive income (loss) attributable to noncontrolling interests	18	4 (364)	444	(183)
Comprehensive income (loss) attributable to Pilgrim's Pride Corporation	\$ (144,61	2) \$ (54,546)	\$ 13,495	\$ (87,066)

PILGRIM'S PRIDE CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (Unaudited)

Six Months Ended June 27, 2021	Commo	Common Stock		Treasury Stock		Additional Paid-in		Accumulated Other Comprehensive Noncontrolling	
	Shares	Amount	Shares	Amount	Capital	Retained Earnings	Income (Loss)		
					(In t	housands)			
Balance at December 27, 2020	261,185	\$ 2,612	(17,673) \$	(345,134) \$	1,954,334	\$ 972,569	\$ (20,620)	\$ 11,586	\$ 2,575,347
Net loss	_	_	_	_	_	(66,479)	_	444	(66,035)
Other comprehensive income, net of tax	_	_	_	_	_	_	79,974	_	79,974
Stock-based compensation plans:									
Common stock issued under compensation plans	162	2	_	_	(2)	_	_	_	_
Requisite service period recognition	_	_	_	_	5,226	_	_	_	5,226
Balance at June 27, 2021	261,347	\$ 2,614	(17,673) \$	(345,134) \$	1,959,558	\$ 906,090	\$ 59,354	\$ 12,030	\$ 2,594,512

Three Months Ended June 27, 2021	Common Stock		Treasury Stock		Additional Paid-in		Accumulated Other Comprehensive Noncontrolling			
	Shares	Amount	Shares	Amount	Capital	Retained Earnings	Income	Interest	7	Total
					(In tl	ousands)				
Balance at March 28, 2021	261,338	\$ 2,613	(17,673) \$	(345,134) \$	1,956,375	\$ 1,072,777	\$ 37,279	\$ 11,846	\$	2,735,756
Net loss	_	_	_	_	_	(166,687)	_	184		(166,503)
Other comprehensive income, net of tax	_	_	_	_	_	_	22,075	_		22,075
Stock-based compensation plans:										
Common stock issued under compensation plans	9	1	_	_	(1)	_	_	_		_
Requisite service period recognition	_	_	_	_	3,184	_	_	_		3,184
Balance at June 27, 2021	261,347	\$ 2,614	(17,673) \$	(345,134) \$	1,959,558	\$ 906,090	\$ 59,354	\$ 12,030	\$	2,594,512

Six Months Ended June 28, 2020	Common Stock		Treasury Stock		Additional Paid-in		Other Comprehensive Noncontrolling		
	Shares	Amount	Shares	Amount	Capital	Retained Earnings		Interest	Total
					(In t	nousands)			
Balance at December 29, 2019	261,119	2,611	(11,547) \$	(234,892)	\$ 1,955,261	\$ 877,812	\$ (75,129)	\$ 10,397	\$ 2,536,060
Net income	_	_	_	_	_	61,232	_	(183)	61,049
Other comprehensive loss, net of tax	_	_	_	_	_	_	(148,298)	_	(148,298)
Stock-based compensation plans:									
Common stock issued under compensation plans	66	1	_	_	(1)	_	_	_	_
Requisite service period recognition	_	_	_	_	3,467	_	_	_	3,467
Common stock purchased under share repurchase program	_	_	(4,121)	(77,879)	_	_	_	_	(77,879)
Balance at June 28, 2020	261,185	2,612	(15,668) \$	(312,771)	\$ 1,958,727	\$ 939,044	\$ (223,427)	\$ 10,214	\$ 2,374,399

Three Months Ended June 28, 2020	Common	Common Stock		Treasury Stock		Additional Paid-in		Noncontrolling	p	
	Shares	Amount	Shares	Amount	Capital	Retained Earnings	Comprehensive Loss	Interest	Total	
					(In th	nousands)				
Balance at March 29, 2020	261,185	\$ 2,612	(13,013) \$	(262,798)	\$ 1,955,936	\$ 945,080	\$ (174,917)	\$ 10,578	\$ 2,476,491	
Net loss	_	_	_	_	_	(6,036)	_	(364)	(6,400)	
Other comprehensive loss, net of tax	_	_	_	_	_	_	(48,510)	_	(48,510)	
Stock-based compensation plans:										
Common stock issued under compensation plans	_	_	_	_	_	_	_	_	_	
Requisite service period recognition	_	_	_	_	2,791	_	_	_	2,791	
Common stock purchased under share repurchase program			(2,655)	(49,973)					(49,973)	
Balance at June 28, 2020	261,185	\$ 2,612	(15,668) \$	(312,771)	\$ 1,958,727	\$ 939,044	\$ (223,427)	\$ 10,214	\$ 2,374,399	

PILGRIM'S PRIDE CORPORATION CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

June 27, 2021 June 28, 2020 (In thousands) Cash flows from operating activities: Net income (loss) \$ (66,035) \$ 61.049 Adjustments to reconcile net income to cash provided by operating activities: Depreciation and amortization 182,260 164,376 Deferred income tax expense (benefit)

Loss on early extinguishment of debt recognized as a component of interest expense (32,809)25,255 24,254 3,467 Stock-based compensation 5,168 Gain on property disposals (5.057)(1.587)Loan cost amortization 2,279 2,422 Accretion of discount related to Senior Notes 675 491 Amortization of premium related to Senior Notes (167) (334) Loss (gain) on equity-method investments Negative adjustment to previously recognized gain on bargain purchase (8) 304 1,740 Changes in operating assets and liabilities: Trade accounts and other receivables (117,610) 29,920 (173,947) (6,027) Inventories 16,350 Prepaid expenses and other current assets (22.072)Accounts payable, accrued expenses and other current liabilities 266,487 (122,191) Income taxes 46,638 (27,350)Long-term pension and other postretirement obligations (1,908) (9,507) Other operating assets and liabilities (1,642) 10,794 Cash provided by operating activities Cash flows from investing activities: 114,952 140,726 Acquisitions of property, plant and equipment Proceeds from property disposals (148,175) (183,744) 21,385 9,894 (4,216) (142,497) Purchase of acquired business, net of cash acquired Cash used in investing activities (162,359) Cash flows from financing activities: Proceeds from revolving line of credit and long-term borrowings 1,540,133 356.547 Payments on revolving line of credit, long-term borrowings and finance lease obligations (1,522,416) (20,105) Payment on early extinguishment of debt (21,258) Payment of capitalized loan costs (8,650)Payment of equity distribution under Tax Sharing Agreement between JBS USA Food Company Holdings and Pilgrim's Pride Corporation (650) Purchase of common stock under share repurchase program (77,879) Cash provided by (used in) financing activities (12,841) 258,563 Effect of exchange rate changes on cash and cash equivalents 1,859 (2,896) Increase (decrease) in cash, cash equivalents and restricted cash (58,389) 253,896 Cash, cash equivalents and restricted cash, beginning of period 548,406 280,577 Cash, cash equivalents and restricted cash, end of period 490,017 534,473

NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. GENERAL

Business

Pilgrim's Pride Corporation (referred to herein as "Pilgrim's," "PPC," "the Company," "we," "us," "our," or similar terms) is one of the largest chicken producers in the world, with operations in the United States ("U.S."), the United Kingdom ("U.K."), Mexico, France, Puerto Rico and the Netherlands. Pilgrim's products are sold to foodservice, retail and frozen entrée customers. The Company's primary distribution is through retailers, foodservice distributors and restaurants throughout the countries listed above. Additionally, the Company exports chicken and pork products to approximately 120 countries. Pilgrim's fresh products consist of refrigerated (nonfrozen) whole chickens, whole cut-up chickens, selected chicken parts that are either marinated or non-marinated, primary pork cuts, added value pork and pork ribs. The Company's prepared products include fully cooked, ready-to-cook and individually frozen chicken parts, strips, nuggets and patties, some of which are either breaded or non-breaded and either marinated or non-marinated, processed sausages, bacon, slow-cooked, smoked meat and gammon joints. The Company's other products include ready-to-eat meals, multi-protein frozen foods, vegetarian foods and desserts, pre-packed meats, sandwich, deli counter meats, pulled pork balls, meat balls and coated foods. As a vertically integrated company, we control every phase of the production of our products. We operate feed mills, hatcheries, processing plants and distribution centers in 14 U.S. States, the U.K., Mexico, France, Puerto Rico and the Netherlands. As of June 27, 2021, PPC had the capacity to process approximately 45.8 million birds per work week for a total of more than 13.3 billion pounds of live chicken annually. Approximately 45,000 contract growers supply chicken for the Company's operations. As of June 27, 2021, PPC had the capacity to process approximately 45,000 pigs per week for a total of 449.4 million pounds of live pork annually and approximately 275 contract growers supply pork for the Co

Condensed Consolidated Financial Statements

The accompanying unaudited condensed consolidated financial statements of the Company have been prepared in accordance with accounting principles generally accepted in the U.S. ("U.S. GAAP") for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X of the U.S. Securities and Exchange Commission ("SEC"). Accordingly, they do not include all of the information and footnotes required by U.S. GAAP for complete financial statements. In the opinion of management, all adjustments (consisting of normal and recurring adjustments unless otherwise disclosed) considered necessary for a fair presentation have been included. Operating results for the six months ended June 27, 2021 are not necessarily indicative of the results that may be expected for the year ending December 26, 2021. For further information, refer to the consolidated financial statements and notes thereto included in the Company's annual report on Form 10-K for the year ended December 27, 2020.

The Company operates on the basis of a 52/53 week fiscal year ending on the Sunday falling on or before December 31. Any reference we make to a particular year (for example, 2021) in the notes to these Condensed Consolidated Financial Statements applies to our fiscal year and not the calendar year. The three months ended June 27, 2021 represents the period from March 29, 2021 through June 27, 2021. The six months ended June 27, 2021 represents the period from March 30, 2020 through June 28, 2020. The six months ended June 28, 2020 represents the period from December 30, 2019 through June 28, 2020.

The Condensed Consolidated Financial Statements include the accounts of the Company and its majority-owned subsidiaries. We eliminate all significant affiliate accounts and transactions upon consolidation.

The Condensed Consolidated Financial Statements have been prepared in conformity with U.S. GAAP using management's best estimates and judgments. These estimates and judgments affect the reported amounts of assets and liabilities at the date of the financial statements. The estimates and judgments will also affect the reported amounts for certain revenues and expenses during the reporting period. Actual results could differ materially from these estimates and judgments. Significant estimates made by the Company include the allowance for doubtful accounts, reserves related to inventory obsolescence or valuation, useful lives of long-lived assets, goodwill, valuation of deferred tax assets, insurance accruals, valuation of pension and other postretirement benefits obligations, income tax accruals, certain derivative positions, certain litigation reserves and valuations of acquired businesses.

The functional currency of the Company's U.S. and Mexico operations and certain holding-company subsidiaries in Luxembourg, the U.K., Malta and Ireland is the U.S. dollar. The functional currency of its U.K. operations is the British pound. The functional currency of the Company's operations in France and the Netherlands is the euro. For foreign currency-denominated entities other than the Company's Mexico operations, translation from local currencies into U.S. dollars is performed for most assets and liabilities using the exchange rates for the balance sheet date. Income and expense accounts are remeasured using average exchange rates for the Period. Adjustments resulting from translation of these financial records are reflected as a separate component of Accumulated other comprehensive income (loss) in the Condensed Consolidated Balance Sheets. For the Company's Mexico operations, remeasurement from the Mexican peso to U.S. dollars is performed for monetary assets and liabilities using the exchange rate in effect as of the balance sheet date. Remeasurement is performed for non-monetary assets using the historical exchange rate in effect on the date of each asset's acquisition. Income and expense accounts are remeasured using average exchange rates for the period. Net adjustments resulting from remeasurement of these financial records, as well as foreign currency transaction gains and losses, are reflected in Foreign currency transaction loss (gain) in the Condensed Consolidated Statements of Income.

Sale and purchase agreement

On June 17, 2021, the Company entered into a sale and purchase agreement (the "SPA") with Kerry Group plc, a public limited company incorporated in the Republic of Ireland and listed on the Irish Stock Exchange (the "Seller") and certain wholly-owned subsidiaries of the Company, Onix Investments UK Limited, a company incorporated in England and Wales, and Arkose Investments ULC, a company incorporated in the Republic of Ireland (together, the "Designated Buyers"), pursuant to which the Designated Buyers shall acquire the entire issued share capital of certain target companies which are wholly-owned indirect subsidiaries of the Seller, that conduct the meats and meals businesses of the Seller ("Kerry Meats" and "Kerry Meals" respectively) in the United Kingdom and Ireland (the "Target Companies") for an aggregate purchase price of £680,000,000 (or approximately \$952,000,000 based on a USD to GBP exchange rate of 1.40 as of June 16, 2021) (the "Acquisition") to be paid by the Designated Buyers (or, failing which, by the Company) in cash upon the Closing (the "Closing"), subject to certain routine closing adjustments in respect of working capital and net indebtedness.

Restricted Cash

The Company is required to maintain cash balances with a broker as collateral for exchange traded futures contracts. These balances are classified as restricted cash as they are not available for use by the Company to fund daily operations. The balance of restricted cash may also include investments in U.S. Treasury Bills that qualify as cash equivalents, as required by the broker, to offset the obligation to return cash collateral.

The following table reconciles cash, cash equivalents and restricted cash as reported in the Condensed Consolidated Balance Sheets to the total of the same amounts shown in the Condensed Consolidated Statements of Cash

	June	June 27, 2021		
	·	(In t	nousands)	<u> </u>
Cash and cash equivalents	\$	391,805	\$	547,624
Restricted cash		98,212		782
Total cash, cash equivalents and restricted cash shown in the Condensed Consolidated Statements of Cash Flows	\$	490,017	\$	548,406

Recent Accounting Pronouncements Adopted as of December 28, 2020

In December 2019, the FASB issued ASU 2019-12, Simplifying the Accounting for Income Taxes, which is intended to improve consistency and simplify several areas of existing guidance. ASU 2019-12 removes certain exceptions to the general principles related to the approach for intraperiod tax allocation, the methodology for calculating income taxes in an interim period and the recognition of deferred tax liabilities for outside basis differences. The new guidance also clarifies the accounting for transactions that result in a step-up in the tax basis of goodwill. The adoption of this guidance did not have a material impact on our financial statements.

In January 2020, the FASB issued ASU 2020-01, Clarifying the Interactions between Topic 321, Topic 323, and Topic 815, which clarifies that an entity should consider observable transactions that require it to either apply or discontinue the equity method of accounting for the purposes of applying the measurement alternative in accordance with Topic 321 immediately before applying or upon discontinuing the equity method. This update also specifies that for the purpose of applying paragraph 815-10-15-1419(a) an entity should not consider whether, upon the settlement of the forward contract or exercise of the purchased option, individually or with existing investments, the underlying securities would be accounted for under the equity method in Topic 323 or the fair value option in accordance with the financial instruments guidance in Topic 825. The entity also would evaluate the remaining characteristics in paragraph 815-10-15-141 to determine the accounting for those forward contracts and purchased options. The adoption of this guidance did not have a material impact on our financial statements.

In October 2020, the FASB issued ASU 2020-10, *Codification Improvements*, which provided codification updates for technical corrections such as conforming amendments, clarifications to guidance, simplifications to wording or structure of guidance, and other minor improvements. Additionally, changes to clarify the codification or correct unintended application of guidance that are not expected to have a significant effect on current accounting practice or create a significant administrative cost to most entities were also included in this update. The adoption of this guidance did not have a material impact on our financial statements.

Recent Accounting Pronouncements Not Yet Adopted as of June 27, 2021

In March 2020, the FASB issued ASU 2020-04, Reference Rate Reform (Topic 848): Facilitation of the Effects of Reference Rate Reform on Financial Reporting, which provides optional expedients and exceptions to the application of current GAAP to existing contracts, hedging relationships and other transactions affected by reference rate reform. The new guidance will ease the transition to new reference rates by allowing entities to update contracts and hedging relationships without applying many of the contract modification requirements specific to those contracts. The provisions of the new guidance will be effective beginning March 12, 2020, extending through December 31, 2022 with the option to apply the guidance at any point during that time period. In January 2021, the FASB issued ASU 2021-01, Reference Rate Reform (Topic 848), which provides further clarification on the scope of Topic 848 so that derivatives affected by the discounting transition are explicitly eligible for certain optional expedients and exceptions in Topic 848. Once an entity elects an expedient or exception it must be applied to all eligible contracts or transactions. We currently have hedging transactions and debt agreements that reference LIBOR and will apply the new guidance as these contracts are modified to reference other rates.

2. REVENUE RECOGNITION

The vast majority of the Company's revenue is derived from contracts which are based upon a customer ordering our products. While there may be master agreements, the contract is only established when the customer's order is accepted by the Company. The Company accounts for a contract, which may be verbal or written, when it is approved and committed by both parties, the rights of the parties are identified along with payment terms, the contract has commercial substance and collectability is probable.

The Company evaluates the transaction for distinct performance obligations, which are the sale of its products to customers. Since its products are commodity market-priced, the sales price is representative of the observable, standalone selling price. Each performance obligation is recognized based upon a pattern of recognition that reflects the transfer of control to the customer at a point in time, which is upon destination (customer location or port of destination), which faithfully depicts the transfer of control and recognition of revenue. There are instances of customer pick-up at the Company's facility, in which case control transfers to the customer at that point and the Company recognizes revenue. The Company's performance obligations are typically fulfilled within days to weeks of the acceptance of the order.

The Company makes judgments regarding the nature, amount, timing and uncertainty of revenue and cash flows arising from revenue and cash flows with customers. Determination of a contract requires evaluation and judgment along with the estimation of the total contract value is constrained. Due to the nature of our business, there is minimal variable consideration, as the contract is established at the acceptance of the order from the customer. When applicable, variable consideration is estimated at contract inception and updated on a regular basis until the contract is completed. Allocating the transaction price to a specific performance obligation based upon the relative standalone selling prices including discounts and variable consideration.

Disaggregated Revenue

Revenue has been	n disaggregated	into the categories	below to	show how economic		offect the nature,	amount, timin	g and uncertainty	of revenue	and cash flows:	
		(In thousands)									
		Fresh		Prepared		Export		Other		Total	
U.S. chicken	\$	1,807,640	\$	207,309	\$	115,844	\$	117,676	\$	2,248,469	
U.K. and Europe chicker	n	218,002		245,397		76,167		22,126		561,692	
Mexico chicken		402,295		29,738		_		21,350		453,383	
Total chicken		2,427,937		482,444		192,011		161,152		3,263,544	
U.K. and Europe pork		61,966		280,372		18,646		13,170		374,154	
Total net sales	\$	2,489,903	\$	762,816	\$	210,657	\$	174,322	\$	3,637,698	

				Three Months Ended June 28, 2020									
	-			(In thousands)									
		Fresh	Prepared	Export	Other	Total							
U.S. chicken	\$	1,452,519	\$ 156,268	\$ 97,470	\$ 92,432	1,798,689							
U.K. and Europe chicken		224,554	137,573	42,761	16,083	420,971							
Mexico chicken		236,690	18,514	_	12,929	268,133							
Total chicken		1,913,763	312,355	140,231	121,444	2,487,793							
U.K. and Europe pork		168,315	138,907	16,093	12,915	336,230							
Total net sales	\$	2,082,078	\$ 451,262	\$ 156,324	\$ 134,359	\$ 2,824,023							
	====	Six Months Ended June 27, 2021											
	_			(In thousands)									
		Fresh	Prepared	Export	Other	Total							
U.S. chicken	\$	3,405,063	\$ 401,581	\$ 229,815	\$ 211,570	\$ 4,248,029							
U.K. and Europe chicken		428,448	461,709	139,717	33,728	1,063,602							
Mexico chicken		775,016	57,143		40,356	872,515							
Total chicken		4,608,527	920,433	369,532		6,184,146							
U.K. and Europe pork		238,407	440,180	34,477	13,913	726,977							
Total net sales	\$	4,846,934	\$ 1,360,613	\$ 404,009	\$ 299,567	\$ 6,911,123							
				Six Months Ended June 28, 2020									
		Fresh		(In thousands)	Other	Total							
***	Φ.		Prepared	Export 156.323									
U.S. chicken	\$	2,994,159			. , .								
U.K. and Europe chicken		456,507	331,978	101,873	31,786	922,144							
Mexico chicken		529,602	43,485		20,832	593,919							
Total chicken		3,980,268	747,818	258,196		5,241,631							
U.K. and Europe pork		332,997	256,537	34,144	33,642	657,320							
Total net sales	\$	4,313,265	\$ 1,004,355	\$ 292,340	\$ 288,991	\$ 5,898,951							

Shipping and Handling Costs

In the rare case when shipping and handling activities are performed after a customer obtains control of the good, the Company has elected to account for shipping and handling as activities to fulfill the promise to transfer the good. When revenue is recognized for the related good before the shipping and handling activities occur, the related costs of those shipping and handling activities are accrued. Shipping and handling costs are recorded within cost of sales.

Taxes

The Company excludes all taxes assessed by a governmental authority that are both imposed on and concurrent with a specific revenue-producing transaction and collected by the entity from a customer (for example, sales, use, value added and some excise taxes) from the transaction price.

Contract Balances

The Company receives payment from customers based on terms established with the customer. Payments are typically due within two weeks of delivery. Revenue contract liabilities relate to payments received in advance of satisfying the performance under the customer contract. The revenue contract liability relates to customer prepayments and the advanced consideration, such as cash, received from governmental agency contracts for which performance obligations to the end customer have not been satisfied.

Changes	in	the	revenue	contract	liabilities	balance	are	as	follows:
								June 27, 2021	
						·	(In thousands)	
Balance, beginning of pe	eriod					\$			65,918
Revenue recognized									(54,875)
Cash received, excluding	g amounts recogn	ized as revenue during the	period						25,232
Balance, end of period	i					\$			36,275

Accounts Receivable

The Company records accounts receivable when revenue is recognized. The Company records an allowance for doubtful accounts to reduce the receivables balance to an amount it estimates is collectible from customers. Estimates used in determining the allowance for doubtful accounts are based on historical collection experience, current trends, aging of accounts receivable and periodic credit evaluations of customers' financial condition. The Company writes off accounts receivable when it becomes apparent, based upon age or customer circumstances, that such amounts will not be collected. Generally, the Company does not require collateral for its accounts receivable.

3. DERIVATIVE FINANCIAL INSTRUMENTS

The Company utilizes various raw materials in its operations, including corn, soybean meal, soybean oil, wheat, natural gas, electricity and diesel fuel, which are all considered commodities. The Company considers these raw materials generally available from a number of different sources and believes it can obtain them to meet its requirements. These commodities are subject to price fluctuations and related price risk due to factors beyond our control, such as economic and political conditions, supply and demand, weather, governmental regulation and other circumstances. Generally, the Company purchases derivative financial instruments, specifically exchange-traded futures and options, in an attempt to mitigate price risk related to its anticipated consumption of commodity inputs for approximately the next twelve months. The Company may purchase longer-term derivative financial instruments on particular commodities if deemed appropriate.

The Company has operations in Mexico, the U.K., France and the Netherlands. Therefore, it has exposure to translational foreign exchange risk when the financial results of those operations are remeasured in U.S. dollars. The Company has purchased foreign currency forward contracts to manage this translational foreign exchange risk.

The Company has exposure to variability in cash flows from interest payments due to the use of variable interest rates on certain long-term debt arrangements in the U.S. reportable segment. The Company has purchased an interest rate swap contract to convert the variable interest rate to a fixed interest rate on a portion of its outstanding long-term debt arrangements in order to manage this interest rate risk and add stability to interest expense and cash flows

The fair value of derivative assets is included in the line item *Prepaid expenses and other current assets* on the Condensed Consolidated Balance Sheets while the fair value of derivative liabilities is included in the line item *Accrued expenses and other current liabilities* on the same statements. The Company's counterparties require that it post collateral for changes in the net fair value of the derivative contracts. This cash collateral is reported in the line item *Restricted cash and cash equivalents* on the Condensed Consolidated Balance Sheets.

Undesignated contracts may include contracts not designated as a hedge or for which the normal purchase normal sales ("NPNS") exception was not elected, contracts that do not qualify for hedge accounting and derivatives that do not or no longer qualify for the NPNS scope exception. The fair value of each of these derivatives is recognized in the Condensed Consolidated Balance Sheets within Prepaid expenses and other current assets or Accrued expenses and other current liabilities. Changes in fair value of each derivative are recognized immediately in the Condensed Consolidated Statements of Income within Net sales, Cost of sales, Selling, general and administrative expense, or Foreign currency transaction loss (gain) depending on the risk the derivative is intended to mitigate. While management believes these instruments help mitigate various market risks, they are not designated and accounted for as hedges as a result of the extensive record keeping requirements.

The Company has elected not to apply the NPNS exemption to a fixed-price product sales contract with a certain customer in order to mitigate various risk exposures and to try to achieve an accounting result that aligns the accounting for the derivative with the economics achieved through the use of the derivative. Transactions originating from this contact are accounted for as undesignated derivatives and recognized at fair value

The Company does not apply hedge accounting treatment to certain derivative financial instruments that it has purchased to mitigate commodity purchase exposures in the U.S. and Mexico or foreign currency transaction exposures on our Mexico operations. Therefore, the Company recognized changes in the fair value of these derivative financial instruments immediately in earnings. Gains or losses related to the commodity derivative financial instruments are included in the line item Cost of sales in the Condensed Consolidated Statements of Income. Gains or losses related to the foreign currency derivative financial instruments are included in the line item Foreign currency transaction loss (gain) and Cost of sales in the Condensed Consolidated Statements of Income.

The Company does apply hedge accounting to certain derivative financial instruments related to its U.K. and Europe reportable segment that it has purchased to mitigate foreign currency transaction exposures. Before the settlement date of the financial derivative instruments, the Company recognizes changes in the fair value of the cash flow hedge into accumulated other comprehensive income ("AOCI"). When the derivative financial instruments are settled, the amount in AOCI is then reclassified to earnings. Gains or losses related to these derivative financial instruments are included in the line item Net sales and Cost of sales in the Condensed Consolidated Statements of Income.

The Company does apply hedge accounting to a derivative financial instrument related to its U.S. reportable segment that it has purchased to mitigate variable interest rate exposures. The interest rate swap has monthly settlement dates. Upon each settlement date, the Company recognizes changes in the fair value of the cash flow hedge into AOCI. Upon settlement of the derivative instrument, the amount in AOCI is then reclassified to earnings. Gains or losses related to the interest rate swap derivative financial instrument are included in the line item Interest expense, net of capitalized interest in the Condensed Consolidated Statements of Income

Information	regarding	the	Company's	outstanding	derivative	instruments	and	cash	collateral	posted	with	brokers	is	included	in	the	following	table:
										June 27, 2021					De	cember 2	7, 2020	
												(In	thousan	ds)				
Fair values:																		
Commodity	derivative assets							\$			28,838			\$			24,059	
Commodity	derivative liabili	ties									(86,806)						(6,531)	
Foreign curre	ency derivative a	issets									2,957						2,204	
Foreign curre	ency derivative l	iabilities									(135)						(428)	
Interest rate :	wap derivative	liabilities									(478)						(640)	
Sales contrac	t derivative asse	ts									5,133						_	
Cash collater	al posted with b	rokers ^(a)									98,210						782	
Derivatives cov	erage ^(b) :																	
Corn											28.6	%					16.0	%
Soybean mea	ıl										31.8	%					24.0	%
Period through	gh which stated p	percent of	f needs are covere	d:														
Corn												May 2022					Decem	ber 2021
Soybean n	neal											July 2022					Decem	ber 2021

Collateral posted with brokers consists primarily of cash, short-term treasury bills, or other cash equivalents Derivatives coverage is the percent of anticipated commodity needs covered by outstanding derivative instru

The following table presents the	gains and losses Three Mor		nstrument held by the Six Months		esignated or qualifyir	
Gains (Losses) by Type of Contract	June 27, 2021	June 28, 2020	June 27, 2021	June 28, 2020	Affected Line Item in the	Condensed Consolidated Statements of Income
		(In thou	sands)			
Foreign currency derivatives gain (loss)	\$ (8,822)	\$ (5,475)	\$ (3,482)	\$ 27,556	Foreign currency transaction	on loss (gain)
Commodity derivative gain (loss)	1,420	(5,154)	18,798	(10,425)	Cost of sales	
Sales contract derivative gain	23,237	_	5,133	_	Net sales	
Total	\$ 15,835	\$ (10,629)	\$ 20,449	\$ 17,131	•	
The following tables present	the components	of the gain	or loss on Gain (Loss) Recognized in Oth	derivatives tha er Comprehensive Income on	1	cash flow hedges:
		Three Months End	ed		Six Months	Ended .
	June 27, 2	021	June 28, 2020	Jui	ne 27, 2021	June 28, 2020
			(Iı	n thousands)		
Foreign currency derivatives	\$	824 \$	(1,42	3) \$	2,309	\$ 2,700
Interest rate swap derivatives		(98)	(92)	9)	(127)	(929)
Total	\$	726 \$	(2,35	2) \$	2,182	\$ 1,771

			Three Mon	ths End	ded		
		June 27, 2021				June 28, 2020	
	Net sales	Cost of sales	Interest expense, net of capitalized interest		Net sales	Cost of sales	 Interest expense, net of capitalized interest
			(In thou	sands)			
Total amounts of income and expense line items presented in the Condensed Consolidated Statements of Income in which the effects of cash flow hedges are recorded	\$ 3,637,698	\$ 3,257,457	\$ 50,651	\$	2,824,023 \$	2,704,164	\$ 32,323
Gain (loss) on cash flow hedging relationships:							
Interest rate swaps	_	_	(158)		_	_	2
Foreign currency contracts	1,582	(344)	_		(199)	359	_

			Six Mont	hs En	nded			
		June 27, 2021				June 28, 2020		
	Net sales	Cost of sales	Interest expense, net of capitalized interest		Net sales	Cost of sales	Inter ca	rest expense, net of pitalized interest
			(In thousands)			<u> </u>		
Total amounts of income and expense line items presented in the Condensed Consolidated Statements of Income in which the effects of cash flow hedges are recorded	\$ 6,911,123	\$ 6,269,639	\$ 80,985	\$	5,898,951 \$	5,601,993	\$	65,011
Gain (loss) on cash flow hedging relationships:								
Interest rate swaps	_	_	(290)		_	_		2
Foreign currency contracts	2,408	(870)	_		(450)	(132)		_

At June 27, 2021, the pre-tax deferred net gains on foreign currency derivatives recorded in AOCI that are expected to be reclassified to the Condensed Consolidated Statements of Income during the next twelve months are \$1.4 million. This expectation is based on the anticipated settlements on the hedged investments in foreign currencies that will occur over the next twelve months, at which time the Company will recognize the deferred losses to earnings.

At June 27, 2021, the pre-tax deferred net losses on interest rate swap derivatives recorded in AOCI that are expected to be reclassified to the Condensed Consolidated Statements of Income during the next twelve months are \$0.5 million. This expectation is based on the anticipated settlements on the hedged interest rate that will occur over the next twelve months, at which time the Company will recognize the deferred losses or gains to earnings.

4. TRADE ACCOUNTS AND OTHER RECEIVABLES

Trade accounts and other receivables, less allowance for doubtful accounts, consisted of the following:

		June 27, 2021	1	December 27, 2020
	·	(In the	usands)	
Trade accounts receivable	\$	800,959	\$	691,499
Notes receivable		29,441		25,712
Other receivables		43,817		31,954
Receivables, gross	·	874,217		749,165
Allowance for doubtful accounts		(7,741)		(7,173)
Receivables, net	\$	866,476	\$	741,992
Accounts receivable from related parties ^(a)	\$	868	\$	1,084

(a) Additional information regarding accounts receivable from related parties is included in "Note 16. Related Party Transactions."

Activity in the allowance for doubtful accounts was as follows:

	June 27, 2021
	(In thousands)
Balance, beginning of period	\$ (7,173)
Provision charged to operating results	(517)
Account write-offs and recoveries	(2)
Effect of exchange rate	(49)
Balance, end of period	\$ (7,741)

5. INVENTORIES

Inventories consisted of the following:

	Jı	une 27, 2021		December 27, 2020
		(In	thousands)	<u> </u>
Raw materials and work-in-process	\$	1,044,299	\$	868,369
Finished products		354,015		356,052
Operating supplies		62,640		66,495
Maintenance materials and parts		69,060		67,877
Total inventories	\$	1,530,014	\$	1,358,793

6. INVESTMENTS IN SECURITIES

The Company recognizes investments in available-for-sale securities as cash equivalents, current investments or long-term investments depending upon each security's length to maturity. The following table summarizes our investments

in available-for-sale securities:

	June 2	27, 2021			Decemb	er 27, 2020	
-	Cost	Fa	air Value		Cost		Fair Value
-			(In th	ousands)			
Cash equivalents:							
Fixed income securities	\$ 34,123	\$	34,123	\$	178,677	\$	178,677

Gross realized gains during the three months ended and six months ended June 27, 2021 related to the Company's available-for-sale securities totaled \$0.8 million and \$2.9 million while gross realized losses were immaterial. Gross realized gains during the three months ended and six months ended June 28, 2020 related to the Company's available-for-sale securities totaled \$1.0 million and \$2.4 million while gross realized losses were immaterial. Proceeds received from the sale or maturity of available-for-sale securities investments are historically disclosed in the Condensed Consolidated Statements of Cash Flows. Net unrealized holding gains and losses on the Company's available-for-sale securities recognized during the six months ended June 27, 2021 and June 28, 2020 that have been included in accumulated other comprehensive loss and the net amount of gains and losses reclassified out of accumulated other comprehensive loss to earnings during the six months ended June 27, 2021 and June 28, 2020 are disclosed in "Note 12. Stockholders' Equity".

7. GOODWILL AND INTANGIBLE ASSETS

The activity in goodwill by segment for the six months ended June 27, 2021 was as follows:

	D	ecember 27, 2020	Currency Translation	June 27, 2021
			(In thousands)	
U.S.	\$	41,936	\$ _	\$ 41,936
U.K. and Europe		835,505	19,655	855,160
Mexico		127,804	_	127,804
Total	\$	1,005,245	\$ 19,655	\$ 1,024,900

Identified intangible assets consisted of the following:

	Dece	mber 27, 2020	A	mortization	Curre	ncy Translation	J	une 27, 2021
				(In tho	usands)			
Cost:								
Trade names	\$	78,343	\$	_	\$	_	\$	78,343
Customer relationships		297,062		_		3,209		300,271
Non-compete agreements		320		_		_		320
Trade names not subject to amortization		405,240		_		9,262		414,502
Accumulated amortization:								
Trade names		(47,486)		(984)		_		(48,470)
Customer relationships		(143,246)		(10,677)		(1,187)		(155,110)
Non-compete agreements		(320)		_		_		(320)
Identified intangible assets, net	\$	589,913	\$	(11,661)	\$	11,284	\$	589,536

Intangible assets are amortized over the estimated useful lives of the assets as follows:

Customer relationships	3-16 years
Trade names	20 years
Non-compete agreements	3 years

At June 27, 2021, the Company assessed if events or changes in circumstances indicated that the asset group-level carrying amounts of its identified intangible assets subject to amortization might not be recoverable. There were no indicators present that required the Company to test the recoverability of the asset group-level carrying amounts of its identified intangible assets subject to amortization at that date.

8. PROPERTY, PLANT AND EQUIPMENT

Property, plant and equipment ("PP&E"), net consisted of the following:

 June 27, 2021	December 27, 2020
 (In thou	sands)
\$ 251,602	\$ 255,171
1,972,410	1,983,823
3,305,901	3,230,199
73,227	73,647
2,182	2,182
254,811	199,161
5,860,133	5,744,183
(3,182,746)	(3,086,692)
\$ 2,677,387	\$ 2,657,491
\$	\$ 251,602 1,972,410 3,305,901 73,227 2,182 254,811 5,860,133 (3,182,746)

The Company recognized depreciation expense of \$89.9 million and \$79.4 million during the three months ended June 27, 2021 and June 28, 2020, respectively. The Company recognized depreciation expense of \$170.6 million and \$153.9 million during the six months ended June 27, 2021 and June 28, 2020, respectively.

During the six months ended June 27, 2021, Pilgrim's spent \$183.7 million on capital projects and transferred \$129.4 million of completed projects from construction-in-progress to depreciable assets. Capital expenditures were primarily incurred during the six months ended June 27, 2021 to improve efficiencies and reduce costs. During the six months ended June 28, 2020, the Company spent \$148.2 million on capital projects and transferred \$153.6 million of completed projects from construction-in-progress to depreciable assets.

During the three and six months ended June 27, 2021, the Company sold certain PP&E for \$8.3 million and \$21.4 million, respectively, in cash and recognized net gains on these sales of \$2.7 million and \$5.1 million, respectively. PP&E sold during the six months ended June 27, 2021 consisted of a broiler farm in Mexico, two processing plants within the U.K. and other miscellaneous equipment. During the three and six months ended June 28, 2020, the Company sold miscellaneous equipment for cash of \$9.3 million and \$9.9 million, respectively, and recognized net gains on these sales of \$1.1 million and \$1.6 million, respectively.

The Company has closed or idled various facilities in the U.S. and in the U.K. The Board of Directors has not determined if it would be in the best interest of the Company to divest any of these idled assets. Management is therefore not certain that it can or will divest any of these assets within one year, is not actively marketing these assets and, accordingly, has not classified them as assets held for sale. The Company continues to depreciate these assets. As of June 27, 2021, the carrying amounts of these idled assets totaled \$43.3 million based on depreciable value of \$206.2 million and accumulated depreciation of \$162.9 million.

As of June 27, 2021, the Company assessed if events or changes in circumstances indicated that the asset group-level carrying amounts of its property, plant and equipment held for use might not be recoverable. There were no indicators present that required the Company to test the recoverability of the asset group-level carrying amounts of its property, plant and equipment held for use at that date.

9. CURRENT LIABILITIES

Current liabilities, other than current notes payable to banks, income taxes and current maturities of long-term debt, consisted of the following components:

	 June 27, 2021	December 27, 2020
	 (In thou	isands)
Accounts payable:		
Trade accounts	\$ 1,010,576	\$ 904,674
Book overdrafts	64,056	106,435
Other payables	17,532	17,601
Total accounts payable	 1,092,164	1,028,710
Accounts payable to related parties(a)	8,595	9,650
Revenue contract liabilities ^(b)	36,275	65,918
Accrued expenses and other current liabilities:		
Legal settlements	391,000	75,000
Compensation and benefits	171,905	189,767
Derivative liabilities	87,419	7,599
Taxes	74,901	67,812
Current maturities of operating lease liabilities	73,643	71,592
Insurance and self-insured claims	66,120	61,212
Accrued sales rebates	25,561	44,708
Interest and debt-related fees	22,736	29,559
U.S. Department of Justice agreement	_	110,524
Other accrued expenses	138,261	150,074
Total accrued expenses and other current liabilities	 1,051,546	807,847
Total accounts payable, accrued expenses and other current liabilities	\$ 2,188,580	\$ 1,912,125

⁽a) Additional information regarding accounts payable to related parties is included in "Note 16. Related Party Transactions."

⁽b) Additional information regarding revenue contract liabilities is included in "Note 2. Revenue Recognition."

\10. INCOME TAXES

The Company recorded income tax expense of \$25.5 million, a (63.1)% effective tax rate, for the six months ended June 27, 2021 compared to income tax expense of \$35.6 million, a 36.8% effective tax rate, for the six months ended June 28, 2020. The decrease in income tax expense in 2021 resulted primarily from a decrease in pre-tax income partially offset by the recognition of deferred tax expense of \$32.2 million related to enactment of the U.K. tax rate change to 25% effective April 1, 2023.

In assessing the realizability of deferred tax assets, management considers whether it is more likely than not that some portion or all of the deferred tax assets will not be realized. The ultimate realization of deferred tax assets is dependent upon the generation of future taxable income during the periods in which those temporary differences become deductible. Management considers the scheduled reversal of deferred tax liabilities (including the impact of available carry back and carry forward periods), projected future taxable income and tax-planning strategies in making this assessment. As of June 27, 2021, the Company did not believe it had sufficient positive evidence to conclude that realization of a portion of its foreign net deferred tax assets are more likely than not to be realized.

For the six months ended June 27, 2021 and June 28, 2020, there is a tax effect of \$(10.2) million and \$9.0 million, respectively, reflected in other comprehensive income.

For the six months ended June 27, 2021 and June 28, 2020, there are immaterial tax effects reflected in income tax expense due to excess tax benefits and shortfalls related to stock-based compensation.

The Company and its subsidiaries file a variety of consolidated and standalone income tax returns in various jurisdictions. In the normal course of business, our income tax filings are subject to review by various taxing authorities. In general, tax returns filed by the Company and its subsidiaries for years prior to 2011 are no longer subject to examination by tax authorities.

As of July 27, 2020, JBS owns in excess of 80% of Pilgrim's. JBS has a federal tax election to file a consolidated tax return with subsidiaries in which it holds an ownership of at least 80%. The Company is currently analyzing the related impacts to our federal and state tax return filings.

11. DEBT

Long-term debt and other borrowing arrangements, including current notes payable to banks, consisted of the following components:

	Maturity	June 27, 2021	December 27, 2020
		 (In thousands)	
Senior notes payable, net of discount at 4.25%	2031	\$ 990,190 \$	_
Senior notes payable, net of discount at 5.875%	2027	845,508	845,149
Senior notes payable, net of premium and discount at 5.75%	2025	_	1,001,693
U.S. Credit Facility (defined below):			
Term note payable at 1.34%	2023	437,500	450,000
Revolving note payable at 3.50%	2023	40,500	_
Moy Park Bank of Ireland Revolving Facility with notes payable at LIBOR or EURIBOR plus 1.25% to 2.00%	2023	_	_
Mexico Credit Facility (defined below) with notes payable at TIIE plus 1.50%	2023	_	_
Secured loans with payables at weighted average of 3.34%	2022	23	38
Finance lease obligations	Various	1,456	1,664
Long-term debt		2,315,177	2,298,544
Less: Current maturities of long-term debt		(25,453)	(25,455)
Long-term debt, less current maturities		2,289,724	2,273,089
Less: Capitalized financing costs		(19,426)	(17,543)
Long-term debt, less current maturities, net of capitalized financing costs		\$ 2,270,298 \$	2,255,546

U.S. Senior Notes

On March 11, 2015, the Company completed a sale of \$500.0 million aggregate principal amount of its 5.75% senior notes due 2025. On September 29, 2017, the Company completed an add-on offering of \$250.0 million of these senior notes. The issuance price of this add-on offering was 102.0%, which created gross proceeds of \$255.0 million. The additional \$5.0 million will be amortized over the remaining life of the senior notes. On March 7, 2018, the Company completed another add-on offering of \$250.0 million of these senior notes (together with the senior notes issued in March 2015 and September 2017, the "Senior Notes due 2025"). The issuance price of this add-on offering was 99.25%, which created gross proceeds of \$248.1 million. The \$1.9 million discount will be amortized over the remaining life of the senior notes. Each issuance of the Senior Notes due 2025 is treated as a single class for all purposes under the 2015 Indenture (defined below) and have the same terms.

The Senior Notes due 2025 are governed by, and were issued pursuant to, an indenture dated as of March 11, 2015 by and among the Company, its guarantor subsidiaries and Regions Bank, as trustee (the "2015 Indenture"). The 2015 Indenture provides, among other things, that the Senior Notes due 2025 bear interest at a rate of 5.75% per annum from the date of issuance until maturity, payable semi-annually in cash in arrears, beginning on September 15, 2015 for the Senior Notes due 2025 that were issued in March 2015 and beginning on March 15, 2018 for the Senior Notes due 2025 that were issued in September 2017 and March 2018.

On April 8, 2021, the Company announced the early tender results in connection with its previously announced offer to purchase for cash any and all of the \$1.0 billion aggregate principal amount of the Senior Notes due 2025. Outstanding principal totaling \$896.1 million, representing 89,6% of the Senior Notes due 2025, was validly tendered. On April 14, 2021, the Company redeemed \$103.9 million, which represented the remaining outstanding principal balance of the Senior Notes due 2025. Tender and call premium of \$21.3 million, capitalized financing costs of \$4.6 million, remaining original issue premium of \$2.6 million from the add-on offering in March 2018 were recognized in earnings during the second quarter of 2021.

On September 29, 2017, the Company completed a sale of \$600.0 million aggregate principal amount of its 5.875% senior notes due 2027. On March 7, 2018, the Company completed an add-on offering of \$250.0 million of these senior notes (together with the senior notes issued in September 2017, the "Senior Notes due 2027"). The issuance price of this add-on offering was 97.25%, which created gross proceeds of \$243.1 million. The \$6.9 million discount will be amortized over the remaining life of the Senior Notes due 2027. Each issuance of the Senior Notes due 2027 is treated as a single class for all purposes under the 2017 Indenture (defined below) and have the same terms.

The Senior Notes due 2027 are governed by, and were issued pursuant to, an indenture dated as of September 29, 2017 by and among the Company, its guarantor subsidiaries and Regions Bank, as trustee (the "2017 Indenture"). The 2017 Indenture provides, among other things, that the Senior Notes due 2027 bear interest at a rate of 5.875% per annum from the date of issuance until maturity, payable semi-annually in cash in arrears, beginning on March 30, 2018 for the Senior Notes due 2027 that were issued in September 2017 and beginning on March 15, 2018 for the Senior Notes due 2027 that were issued in March 2018.

On April 8, 2021, the Company completed a sale of \$1.0 billion aggregate principal amount of its 4.25% sustainability-linked senior notes due 2031 ("Senior Notes due 2031"). The Company used the net proceeds, together with cash on hand, to redeem the Senior Notes due 2025. The issuance price of this offering was 98.994%, which created gross proceeds of \$989.9 million. The \$10.1 million discount will be amortized over the remaining life of the Senior Notes due 2031. Each issuance of the Senior Notes due 2031 is treated as a single class for all purposes under the 2021 Indenture (defined below) and have the same terms.

The Senior Notes due 2031 are governed by, and were issued pursuant to, an indenture dated as of April 8, 2021 by and among the Company, its guarantor subsidiaries and Regions Bank, as trustee (the "2021 Indenture"). The 2021 Indenture provides, among other things, that the Senior Notes due 2031 bear interest at a rate of 4.25% per annum payable semi-annually on April 15 and October 15 of each year, beginning on October 15, 2021. From and including October 15, 2026, the interest rate payable on the notes shall be increased to 4.50% per annum unless the Company has notified the trustee at least 30 days prior to October 15, 2026 that in respect of the year ended December 31, 2025, (1) the Company's greenhouse gas emissions intensity reduction target of 17.679% by December 31, 2025 from a 2019 baseline (the "Sustainability Performance Target") has been satisfied and (2) the satisfaction of the Sustainability Performance Target has been confirmed by a qualified provider of third-party assurance or attestation services appointed by the Company to review the Company's statement of the greenhouse gas emissions intensity in accordance with its customary procedures.

The Senior Notes due 2025, the Senior Notes due 2027 and the Senior Notes due 2031 were and are each guaranteed on a senior unsecured basis by the Company's guarantor subsidiaries. In addition, any of the Company's other existing or future domestic restricted subsidiaries that incur or guarantee any other indebtedness (with limited exceptions) must also guarantee the Senior Notes due 2027 and the Senior Notes due 2031. The Senior Notes due 2025, the Senior Notes due 2027 and the Senior Notes due 2031 and related guarantees were and are unsecured senior obligations of the Company and its guarantor subsidiaries and rank equally with all of the Company's and its guarantor subsidiaries' other unsubordinated indebtedness. The Senior Notes due 2027, the 2017 Indenture, the Senior Notes due 2031 and the 2021 Indenture also contain customary covenants and events of default, including failure to pay principal or interest on the Senior Notes due 2027 and the Senior Notes due 2031, respectively, when due, among others.

U.S. Credit Facility

On July 20, 2018, the Company, and certain of the Company's subsidiaries entered into a Fourth Amended and Restated Credit Agreement (the "U.S. Credit Facility") with CoBank, ACB, as administrative agent and collateral agent, and the other lenders party thereto. The U.S. Credit Facility provides for a \$750.0 million revolving credit commitment and a term loan commitment of up to \$500.0 million (the "Term Loans"). The Company used the proceeds from the term loan commitment under the U.S. Credit Facility, together with cash on hand, to repay the outstanding loans under the Company's previous credit agreement with Coöperatieve Rabobank U.A., New York Branch, as administrative agent, and the other lenders and financial institutions party thereto.

The U.S. Credit Facility includes an accordion feature that allows the Company, at any time, to increase the aggregate revolving loan and term loan commitments by up to an additional \$1.25 billion, subject to the satisfaction of certain conditions, including obtaining the lenders' agreement to participate in the increase.

The revolving loan commitment under the U.S. Credit Facility matures on July 20, 2023. All principal on the Term Loans is due at maturity on July 20, 2023. Installments of principal are required to be made, in an amount equal to 1.25% of the original principal amount of the Term Loans, on a quarterly basis prior to the maturity date of the Term Loans. Covenants in the U.S. Credit Facility also require the Company to use the proceeds it receives from certain asset sales and specified debt or equity issuances and upon the occurrence of other events to repay outstanding borrowings under the U.S. Credit Facility. As of June 27, 2021, the Company had outstanding borrowings under the term loan commitment of \$437.5 million. As of June 27, 2021, the Company had outstanding borrowings, outstanding letters of credit and available borrowings under the revolving credit commitment of \$40.5 million, \$38.5 million and \$671.0 million, respectively.

The U.S. Credit Facility includes a \$75.0 million sub-limit for swingline loans and a \$125.0 million sub-limit for letters of credit. Outstanding borrowings under the revolving loan commitment and the Term Loans bear interest at a per annum rate equal to (1) in the case of LIBOR loans, LIBOR plus 1.25% through August 2, 2018 and, thereafter, based on the Company's net senior secured leverage ratio, between LIBOR plus 1.25% and LIBOR plus 2.75% and (2) in the case of alternate base rate plus 0.25% through August 2, 2018 and, based on the Company's net senior secured leverage ratio, between the base rate plus 0.25% and base rate plus 1.75% thereafter.

The U.S. Credit Facility contains customary financial and other various covenants for transactions of this type, including restrictions on the Company's ability to incur additional indebtedness, incur liens, pay dividends, make certain restricted payments, consummate certain asset sales, enter into certain transactions with the Company's affiliates, or merge, consolidate and/or sell or dispose of all or substantially all of its assets, among other things. The U.S. Credit Facility requires the Company to comply with a minimum level of tangible net worth covenant. The U.S. Credit Facility also provides that the Company may not incur capital expenditures in excess of \$500.0 million in any fiscal year.

All obligations under the U.S. Credit Facility continue to be unconditionally guaranteed by certain of the Company's subsidiaries and continue to be secured by a first priority lien on (1) the accounts receivable and inventory of the Company and its domestic subsidiaries, To-Ricos, Ltd. and To-Ricos Distribution, Ltd., (2) 100% of the equity interests in the Company's domestic subsidiaries, To-Ricos, Ltd. and To-Ricos Distribution, Ltd., and 65% of the equity interests in its direct foreign subsidiaries and (3) substantially all of the assets of the Company and the guarantors under the U.S. Credit Facility. The Company is currently in compliance with the covenants under the U.S. Credit Facility.

Moy Park Bank of Ireland Revolving Facility Agreement

On June 2, 2018, Moy Park Holdings (Europe) Ltd. and its subsidiaries entered into an unsecured multicurrency revolving facility agreement (the "Bank of Ireland Facility Agreement") with the Governor and Company of the Bank of Ireland, as agent, and the other lenders party thereto. The Bank of Ireland Facility Agreement provides for a multicurrency revolving loan commitment of up to £100.0 million. The multicurrency revolving loan commitments under the Bank of Ireland Facility Agreement mature on June 2, 2023. Outstanding borrowings under the Bank of Ireland Facility Agreement bear interest at a rate per annum equal to the sum of (1) LIBOR or, in relation to any loan in euros, EURIBOR, plus (2) a margin, ranging from 1.25% to 2.00% based on Leverage (as defined in the Bank of Ireland Facility Agreement). All obligations under the Bank of Ireland Facility Agreement are guaranteed by certain of Moy Park's subsidiaries. As of June 27, 2021, the U.S. dollar-equivalent loan commitment and borrowing availability were both \$138.8 million. As of June 27, 2021, there were no outstanding borrowings under the Bank of Ireland Facility Agreement.

The Bank of Ireland Facility Agreement contains representations and warranties, covenants, indemnities and conditions that the Company believes are customary for transactions of this type. Pursuant to the terms of the Bank of Ireland Facility Agreement, Moy Park is required to meet certain financial and other restrictive covenants. Additionally, Moy Park is prohibited from taking certain actions without consent of the lenders, including, without limitation, incurring additional indebtedness, entering into certain mergers or other business combination transactions, permitting liens or other encumbrances on its assets and making restricted payments, including dividends, in each case except as expressly permitted under the Bank of Ireland Facility Agreement. The Bank of Ireland Facility Agreement contains events of default that the Company believes are customary for transactions of this type. If a default occurs, any outstanding obligations under the Bank of Ireland Facility Agreement may be accelerated.

Mexico Credit Facility

On December 14, 2018, certain of the Company's Mexican subsidiaries entered into an unsecured credit agreement (the "Mexico Credit Facility") with Banco del Bajio, Sociedad Anónima, Institución de Banca Múltiple, as lender. The loan commitment under the Mexico Credit Facility is \$1.5 billion Mexican pesos and can be borrowed on a revolving basis. Outstanding borrowings under the Mexico Credit Facility accrue interest at a rate equal to the 28-Day Interbank Equilibrium Interest Rate plus 1.5%. The Mexico Credit Facility contains covenants and defaults that the Company believes are customary for transactions of this type. The Mexico Credit Facility will be used for general corporate and working capital purposes. The Mexico Credit Facility will mature on December 14, 2023. As of June 27, 2021, the U.S. dollar-equivalent of the loan commitment and borrowing availability was \$75.7 million. As of June 27, 2021, there were no outstanding borrowings under the Mexico Credit Facility.

12. STOCKHOLDERS' EQUITY

Accumulated Other Comprehensive Income

	The	following	tables	provide	infor	mation	regarding	the	changes	in	accumulated	other	comprehensive	income:
									Six M	onths Ended June 27	2021 ^(a)			
					Gains Re Currency Tr	elated to Foreign anslation	Financial Instr	osses on Derivative ruments Classified as Flow Hedges		Losses Related to Pen Postretirement Bene	sion fits Availa	Gains (losses) on ble-for-Sale Securities		Total
										(In thousands)				
	Balance, be	eginning of period			\$	82,782	\$	(1,191)	\$	(102	211)	\$	\$	(20,620)
Other	comprehensive	income before reclas	ssifications			49,138		2,214		29	185	_		80,537
	nts reclassified et income	from accumulated ot	ther comprehensive	e income		_		(1,320)			730	_		(590)
	Currency to	ranslation				_		27			_	_		27
	Net current	period other compre	hensive income			49,138		921		29	915	=		79,974
	Balance, er	nd of period		_	\$	131,920	\$	(270)	\$	(72	296)	\$	\$	59,354

(a) All amounts are net of tax. Amounts in parentheses represent income (expenses) related to results of operations.

			Six	Months Ended June 28, 2020 ^(a)			
	Losses Related to Foreign Currency Translation	ains on Derivative Financial struments Classified as Cash Flow Hedges		Losses Related to Pension and Other Postretirement Benefits	G	ains on Available-for-Sale Securities	Total
				(In thousands)			
Balance, beginning of period	\$ (1,108)	\$ (2,406)	\$	(71,615)	\$	- :	\$ (75,129)
Other comprehensive income (loss) before reclassifications	(115,547)	2,003		(35,797)		10	(149,331)
Amounts reclassified from accumulated other comprehensive loss to net income	_	580		564		(9)	1,135
Currency translation	_	(102)		_		_	(102)
Net current period other comprehensive income (loss)	(115,547)	2,481		(35,233)		1	(148,298)
Balance, end of period	\$ (116,655)	\$ 75	\$	(106,848)	\$	1 5	\$ (223,427)

(a) All amounts are net of tax. Amounts in parentheses represent income (expenses) related to results of operations.

	Amount R	eclassified from Accumul	lated O	Other Comprehensive Income ^(a)	
Details about Accumulated Other Comprehensive Income Components	Six Months	Ended June 27, 2021	S	Six Months Ended June 28, 2020	Affected Line Item in the Condensed Consolidated Statements of Income
		(In the	ousands	ls)	·
Realized gain (loss) on settlement of foreign currency derivatives classified as cash flow hedges	\$	1,282	\$	(582)	Net sales
Realized gain on settlement of foreign currency derivatives classified as cash flow hedges		255		_	Cost of sales
Realized loss on settlement of interest rate swap derivatives classified as cash flow hedges		(289)		2	Interest expense, net of capitalized interest
Realized gain on sale of securities		_		12	Interest income
Amortization of pension and other postretirement plan actuarial losses(b)		(955)		(751)	Miscellaneous, net
Total before tax		293		(1,319)	
Tax expense		297		184	
Total reclassification for the period	\$	590	\$	(1,135)	

- (a) Positive amounts represent income to the results of operations while amounts in parentheses represent expenses to the results of operations
- by These accumulated other comprehensive loss components are included in the computation of net periodic pension cost. See "Note 13. Pension and Other Postretirement Benefits."

Share Repurchase Program and Treasury Stock

On October 31, 2018, the Company's Board of Directors approved a \$200.0 million share repurchase authorization. The Company repurchased shares through open market purchases. As of June 27, 2021, the Company repurchased approximately 6.3 million shares under this program with a market value of approximately \$113.4 million. The Company accounted for the shares repurchased using the cost method. The Company currently plans to maintain these shares as treasury stock. This program expired on February 16, 2021.

Restrictions on Dividends

Both the U.S. Credit Facility and the indentures governing the Company's senior notes restrict, but do not prohibit, the Company from declaring dividends. Additionally, the Moy Park Multicurrency Revolving Facility Agreement restricts Moy Park's ability and the ability of certain of Moy Park's subsidiaries to, among other things, make payments and distributions to the Company.

13. PENSION AND OTHER POSTRETIREMENT BENEFITS

The Company sponsors programs that provide retirement benefits to most of its employees. These programs include qualified defined benefit pension plans such as the Pilgrim's Pride Retirement Plan for Union Employees (the "Union Plan") the Pilgrim's Pride Pension Plan for Legacy Gold Kist Employees (the "GK Pension Plan"), the Tulip Limited Pension Plan and the Geo Adams Group Pension Fund (together, the "U.K. Plans"), nonqualified defined benefit retirement plans, a defined benefit postretirement life insurance plan and defined contribution retirement savings plan. Expenses recognized under all retirement plans totaled \$4.0 million and \$3.5 million in the three months ended June 27, 2021 and June 28, 2020, respectively, and \$9.3 million in the six months ended June 27, 2021 and June 28, 2020, respectively.

Defined Benefit Plans Obligations and Assets

The change in benefit obligation, change in fair value of plan assets, funded status and amounts recognized in the Condensed Consolidated Balance Sheets for the defined benefit plans were as follows:

Six Months Ended

	SIA MORIUS LIRUCU												
		June 2	27, 2021		June 28, 2020								
	Per	nsion Benefits	Ot	her Benefits	Pen	sion Benefits	Oth	er Benefits					
	'			(In th	ousands)								
Change in projected benefit obligation													
Projected benefit obligation, beginning of period	\$	404,194	\$	1,593	\$	369,066	\$	1,527					
Interest cost		2,437		8		4,050		18					
Actuarial loss (gain)		(28,552)		(35)		28,806		64					
Benefits paid		(5,978)		(70)		(9,965)		(80)					
Curtailments and settlements		(2,689)		_		_		_					
Prior service cost		_		_		11		_					
Currency translation loss (gain)		10,196		_		(11,001)		_					
Projected benefit obligation, end of period	\$	379,608	\$	1,496	\$	380,967	\$	1,529					

Pens	June 27, 2021 sion Benefits	Other Benefits (In thousands)	June 28, 2020 Pension Benefits	Other Benefits
Pens			Pension Benefits	Other Benefits
\$	20E 092 ¢	(In thousands)		
\$	20E 002 - ¢			
\$	20E 002 ¢			
	202,962 \$	— \$	294,589 \$	_
	14,564	_	(9,948)	_
	6,832	70	5,173	80
	(5,978)	(70)	(9,965)	(80)
	(2,689)	_	_	_
	(169)	_	(526)	_
	8,764	_	(9,849)	_
\$	327,307 \$	_ \$	269,474 \$	_
	June 27, 2021		December 27, 207	20
	\$	14,564 6,832 (5,978) (2,689) (169) 8,764 \$ 327,307 \$ June 27, 2021	14,564 — 6,832 70 (5,978) (70) (2,669) — (169) — 8,764 — \$ 327,307 \$ —	14,564 — (9,948) 6,832 70 5,173 (5,978) (70) (9,965) (2,689) — — (169) — (526) 8,764 — (9,849) \$ 327,307 \$ \$ 269,474 \$

		June 27, 2021		December 27, 2020			
	Pension Benefits		Other Benefits	Pension Benefits	Other Benefits		
			(In thou	sands)			
Funded status							
Unfunded benefit obligation, end of period	\$	(52,301) \$	(1,496)	\$ (98,211)	\$ (1,593)		

	June 2			0			
	 Pension Benefits		Other Benefits		Pension Benefits		Other Benefits
			(In tho	usands)			
Amounts recognized in the Condensed Consolidated Balance Sheets at end of period							
Current liability	\$ (819)	\$	(167)	\$	(7,510)	\$	(169)
Long-term liability	(51,482)		(1,329)		(90,701)		(1,424)
Recognized liability	\$ (52,301)	\$	(1,496)	\$	(98,211)	\$	(1,593)
	June 2	7, 2021			Decembe	r 27, 202	0
	 Pension Benefits		Other Benefits		Pension Benefits		Other Benefits
			(In tho	usands)			
Amounts recognized in accumulated other comprehensive loss at end of period							
Net actuarial loss	\$ 55,846	\$	138	\$	95,522	\$	174

The accumulated benefit obligation for the Company's defined benefit pension plans was \$379.6 million and \$404.2 million at June 27, 2021 and December 27, 2020, respectively. Each of the Company's defined benefit pension plans had accumulated benefit obligations that exceeded the fair value of plan assets at both June 27, 2021 and December 27, 2020. As of June 27, 2021, the weighted average duration of the Company's defined benefit pension obligation is 19.67 years.

Net Periodic Benefit Costs

	Net defin	ied	ber	nefit	pension		and	other	post	retirement	C	osts	included	tl	he	following	con	nponents:
			Th	ree Months	Ended June 27,	2021		Three Months	Ended June 28,	2020		Six Months E	nded June 27, 2021			Six Months E	nded June 28, 20	120
			Pension	1 Benefits	Othe	r Benefits		Pension Benefits	Othe	er Benefits	Pens	sion Benefits	Other I	Benefits	Pe	nsion Benefits	Othe	er Benefits
										(1	n thousands)							
	Interest cost		\$	987	\$	3	5	3 2,011	\$	9	\$	2,437	\$	8	\$	4,050	\$	18
plan as	Estimated return on sets			(1,815)		_		(3,215)		_		(4,456)		_		(6,498)		_
	Settlement loss			837		_		_		_		837		_		_		_
assets	Expenses paid from			91		_		93		_		169		_		537		_
loss	Amortization of net			381		_		375		_		946		1		751		_
service	Amortization of past cost			3		_		_		_		8		_		_		_
	Net costs(a)		\$	484	\$	3	9	(736)	\$	9	\$	(59)	\$	9	\$	(1,160)	\$	18

 $(a) \quad \text{Net costs are included in the line item } \textit{Miscellaneous}, \textit{net} \text{ on the Condensed Consolidated Statements of Income.}$

Economic Assumptions

	The	weighted	average	assumptions	used	in	determining	pension	and	other	postre	etirement	plan	information	were	as	follows:
								June 27, 2	021					December	27, 2020		
					_		Pension Benefits			Other Benefits			Pension Benef	fits		Other Benefits	3
	Assumpti	ions used to measu	re benefit obligati	on at end of period	_												
	Discount	rate					2.31	%		2.20	%		1.83	%		1.80	%
												Six Months En	nded				
									June 27, 2	021				June 28	8, 2020		
							Pe	nsion Benefits		Other B	enefits		Pension B	enefits	(Other Benefits	
I	Assumptions used t	to measure net pen	sion and other po	stretirement cost													

The discount rate represents the interest rate used to determine the present value of future cash flows currently expected to be required to settle the Company's pension and other benefit obligations. The discount rate assumptions used to determine future pension obligations at June 27, 2021 and December 27, 2020 were based on Prudential Financial, Inc.'s ("Prudential") Pru Above Mean yield curve, which was designed by Prudential to provide a means for plan sponsors to value the liabilities of their postretirement benefit plans. The Pru Above Mean yield curve represents a series of annual discount rates from bonds with an AA minimum average credit quality rating as rated by Moody's Investor Service, Standard & Poor's and Fitch Ratings. The expected benefit payments were discounted by each corresponding discount rate on the yield curve. For payments beyond 30 years, the Company extended the curve assuming the discount rate derived in year 30 is extended to the end of the plan's payment expectations. Once the present value of the string of benefit payments was established, the Company determined the single rate on the yield curve, that when applied to all obligations of the plan, would exactly match the previously determined present value. The discount rate assumptions used to determine future pension obligations for the U.K. pension plans at June 27, 2020 were based on corporate bond spot yield curves provided by Merrill Lynch. Merrill Lynch bases this calculation entirely on AA1-AA3 rated bonds. As part of the evaluation of pension and other postretirement assumptions, the Company applied assumptions for mortality that incorporate generational white and blue collar mortality trends. In determining its benefit obligations, the Company used generational tables that take into consideration increases in plan participant longevity. As of June 27, 2021, the U.K. pension plans used variations of the Pri-2012 mortality improvement scale. As of June 27, 2021 and December 27, 2020, the U.K. pension plans used variations

The sensitivity of the projected benefit obligation for pension benefits to changes in the discount rate is set out below. The impact of a change in the discount rate of 0.25% on the projected benefit obligation for other benefits is less than \$1,000. This sensitivity analysis is based on changing one assumption while holding all other assumptions constant. In practice, this is unlikely to occur, and changes in some of the assumptions may be correlated. When calculating the sensitivity of the defined benefit obligation to variations in significant actuarial assumptions, the same method (present value of the defined benefit obligation calculated with the projected unit credit method at the end of period) has been applied as for calculating the liability recognized in the Condensed Consolidated Balance Increase in Discount Rate of 0.25% Decrease in Discount Rate of 0.25% (In thousands) Impact on projected benefit obligation for pension benefits (10,162)

The expected rate of return on plan assets was primarily based on the determination of an expected return and for each plan's current asset portfolio that the Company believes are likely to prevail over long periods. This determination was made using assumptions for return and volatility of the portfolio. Asset class assumptions were set using a combination of empirical and forward-looking analysis. To the extent historical results were affected by unsustainable trends or events, the effects of those trends or events were quantified and removed. The Company also considered anticipated asset allocations, investment strategies and the views of various investment professionals when developing this rate.

Plan Assets

The	following	table	reflects	the	pension	plans'	actual	asset	allocations:
					-	June 27, 2021		December 27, 2020	
Cash and cash	equivalents				· · · · · · · · · · · · · · · · · · ·	2	2 %	1	%
Pooled separat	e accounts for the Union Plan	(a):							
Equity secur	ities					2	2 %	2	2 %
Fixed incom	e securities					2	2 %	2	2 %
Pooled separat	e accounts and common colle	ctive trust funds for the G	K Pension Plan ^(a) :						
Equity secur	ities					20) %	20) %
Fixed incom	e securities					12	2 %	13	8 %
Real estate							l %	1	%
Pooled separat	e accounts for the U.K. Plans	a):							
Equity secur	ities					36	6 %	35	%
Fixed incom	e securities					19	9 %	20) %
Real estate						(6 %	6	6 %
Total assets						100) %	100) %

(a) Pooled separate accounts ("PSAs") and common collective trust funds ("CCTs") are two of the most common types of alternative vehicles in which benefit plans invest. These investments are pooled funds that look like mutual funds, but they are not registered with the SEC. Often times, they will be invested in mutual funds or other marketable securities, but the unit price generally will be different from the value of the underlying securities because the fund may also hold cash for liquidity purposes, and the fees imposed by the fund are deducted from the fund value rather than charged separately to investors. Some PSAs and CCTs have no restrictions as to their investment strategy and can invest in riskier investments, such as derivatives, hedge funds, private equity funds, or similar investments.

Absent regulatory or statutory limitations, the target asset allocation for the investment of pension assets in the PSAs for the Union Plan is 50% in each of fixed income securities and equity securities, the target asset allocation for the investment of pension assets in the PSAs and/or CCTs for the GK Pension Plan is 35% in fixed income securities, 60% in equity securities and 5% in real estate and investment of pension assets in the PSAs for the U.K. Plans is 28% in fixed income securities, 62% in equity securities and 10% in real estate. The plans only invest in fixed income and equity instruments for which there is a readily available public market. The Company develops its expected long-term rate of return assumptions based on the historical rates of returns for equity and fixed income securities of the type in which its plans invest.

The fair val	lue	measurements	of plan		ell into une 27, 2021	the following	ng levels	of the	fair va	lue hierarchy	as of		2021 ar er 27, 2020	nd De	cember 27	7, 2020:
_		Level 1 ^(a)		Level 2 ^(b)		Level 3 ^(c)		Total	a 15	Level 1 ^(a)		Level 2 ^(b)	Lev	el 3 ^(c)		Total
Cash and cash equivalents	\$	6,170	\$	_		s —	\$	6,170	thousands)	1,487	\$	_	\$	_	\$	1,48
PSAs for the Union Plan:																
Large U.S. equity funds ^(d)		_		2,776		_		2,776		_		3,100		_		3,100
Small/Mid U.S. equity funds ^(e)		_		1,116		_		1,116		_		392		_		39:
International equity funds ^(f)		_		2,175		_		2,175		_		1,874		_		1,87
Fixed income funds ^(g)		_		5,042		_		5,042		_		5,365		_		5,36
PSAs and CCTs for the GK Pension Plan:																
Large U.S. equity funds ^(d)		_		32,409		_		32,409		_		29,602		_		29,602
Small/Mid U.S. equity funds ^(e)		_		16,636		_		16,636		_		17,569		_		17,569
International equity funds ^(f)		_		16,600		_		16,600		_		16,320		_		16,320
Fixed income funds(g)		_		39,229		_		39,229		_		38,944		_		38,94
Real estate ^(h) PSAs for the U.K. Plans:		_		4,852		_		4,852		_		5,677		_		5,67
Large U.S. equity funds ^(d)		_		46,814		_		46,814		_		39,002		_		39,00:
equity funds ^(f)		_		73,672		_		73,672		_		69,251		_		69,25
Fixed income funds ^(g)		_		61,276		_		61,276		_		60,212		_		60,212
Real estate(h)				18,540				18,540				17,188				17,18
Total assets	\$	6,170	\$	321,137		s –	\$	327,307	\$	1,487	\$	304,496	\$	_	\$	305,983

(a) Unadjusted quoted prices in active markets for identical assets are used to determine fair value.
(b) Quoted prices in active markets for similar assets and inputs that are observable for the asset are used to determine fair value.
(c) Unobservable inputs, such as discounted cash flow models or valuations, are used to determine fair value.
(d) This category is comprised of investment options that invest in stocks, or shares of ownership, in large, well-established U.S. companies. These investment options typically carry more risk than fixed income options but have the potential for higher returns over longer time periods.
(e) This category is comprised of investment options that invest in stocks, or shares of ownership, in small to medium-sized U.S. companies. These investment options typically carry more risk than larger U.S. equity investment options but have the potential for higher returns.
(f) This category is comprised of investment options that invest in stocks, or shares of ownership, in companies or office outside of the U.S.
(g) This category is comprised of investment options that invest in bonds, or debt of a company or government entity (including U.S. and non-U.S. entities). These investment options typically carry more risk than short-term fixed income investment options, but less overall risk than equities.

This category is comprised of investment options that invest in real scatate investment trusts or private equity pools that own real estate. These long-term investments are primarily in office buildings, industrial parks, apartments or retail complexes. These investment options typically carry more risk, including liquidity risk, than fixed income investment options.

Benefit Payments

The following table reflects the benefits as of June 27, 2021 expected to be paid through 2030 from the Company's pension and other postretirement plans. The Company's pension plans are primarily funded plans. Therefore, anticipated benefits with respect to these plans will come primarily from the trusts established for these plans. The Company's other postretirement plans are unfunded. Therefore, anticipated benefits with respect to these plans will come from the Company's other postretirement plans are unfunded. Therefore, anticipated benefits with respect to these plans will come from assets.

COII	IC.	nom	uic		Company 3		OWII		assets.
				<u></u>		Pension Benefits		Other Benefits	
				·		(Ir	thousands)		
	2021				\$	17,363	\$		99
	2022					16,792			163
	2023					16,348			156
	2024					15,962			149
	2025					15,607			140
	2026-2030					71,821			555
	Total			_	\$	153,893	\$		1,262

As required by funding regulations or laws, the Company anticipates contributing \$4.8 million and \$0.2 million to its pension plans and other postretirement plans, respectively, during the remainder of 2021.

Unrecognized Benefit Amounts in Accumulated Other Comprehensive Income

The amounts in accumulated other comprehensive income that were not recognized as components of net periodic benefits cost and the changes in those amounts are as follows:

		Six Months Ended							
		Jun	e 27, 2021		June 28, 2020				
	Per	Pension Benefits Other Benefits				Pension Benefits		Other Benefits	
				(Iı	thousands)				
Net actuarial loss, beginning of period	\$	95,522	\$	174	\$	58,239	\$	91	
Amortization		(954)		(1)		(751)		_	
Curtailment and settlement adjustments		(837)		_		_		_	
Actuarial loss (gain)		(28,552)		(35)		28,806		64	
Asset loss (gain)		(10,109)		_		16,438		_	
Currency translation loss (gain)		776		_		(202)		_	
Net actuarial loss, end of period	\$	55,846	\$	138	\$	102,530	\$	155	

Risk Management

Through its defined benefit plans, the Company is exposed to a number of risks, the most significant of which are detailed below:

Asset volatility. The plan liabilities are calculated using a discount rate set with reference to corporate bond yields; if plan assets under perform this yield, this will create a deficit. The pension plans hold a significant proportion of equities, which are expected to outperform corporate bonds in the long-term while contributing volatility and risk in the short-term. The Company monitors the level of investment risk but has no current plan to significantly modify the mixture of investments. The investment position is discussed more below.

Changes in bond yields. A decrease in corporate bond yields will increase plan liabilities, although this will be partially offset by an increase in the value of the plans' bond holdings.

The investment position is managed and monitored by a committee of individuals from various departments. This group actively monitors how the duration and the expected yield of the investments are matching the expected cash outflows arising from the pension obligations. The group has not changed the processes used to manage its risks from previous periods. The group does not use derivatives to manage its risk. Investments are well diversified, such that the failure of any single investment would not have a material impact on the overall level of assets. The majority of equities are in U.S. large and small cap companies with some global diversification into international entities.

Remeasurement

The Company remeasures both plan assets and obligations on a quarterly basis.

Defined Contribution Plans

The Company sponsors two defined contribution retirement savings plans in the U.S. reportable segment for eligible U.S. and Puerto Rico employees. The Company maintains three postretirement plans for eligible employees in the Mexico reportable segment, as required by Mexico law, which primarily cover termination benefits. The Company maintains two defined contribution retirement savings plans in the U.K. and Europe reportable segment for eligible U.K. and Europe employees, as required by U.K. and Europe law. The Company's expenses related to its defined contribution plans totaled \$3.5 million in the three months ended June 27, 2021 and \$8.1 million in the six months ended June 27, 2021.

14. STOCK-BASED COMPENSATION

For the three months ended June 27, 2021 and June 28, 2020, we recognized total stock-based compensation expense of \$3.2 million and \$3.2 million, respectively. For the three months ended June 27, 2021 and June 28, 2020, the total income tax benefit recognized for stock-based compensation arrangements was \$0.8 million and \$0.8 million, respectively.

For the six months ended June 27, 2021 and June 28, 2020, we recognized total stock-based compensation expense of \$5.2 million and \$3.9 million, respectively. For the six months ended June 27, 2021 and June 28, 2020, the total income tax benefit recognized for stock-based compensation arrangements was \$1.3 million and \$1.0 million, respectively.

During the six months ended June 27, 2021, we granted 145,000 time-vesting restricted stock units at a grant date price of \$19.73 per unit. These awards will vest equally on July 1, 2022, July 1, 2023 and July 1, 2024. We also granted 561,236 performance-based restricted stock units at a grant date price of \$22.39 per unit. These awards will convert to time-vesting restricted stock units in the first quarter of 2022 if or when the Compensation Committee of the Company's Board of Directors certifies the achievement of 2021 performance targets. Once converted to time-vesting restricted stock units, the awards will vest equally on December 31, 2022, December 31, 2023 and December 31, 2024. During the same period we also granted 400,000 performance-based restricted stock units at a grant date price of \$19.73 per unit. These awards will convert to time-vesting restricted stock units in the first quarter of 2024 if or when the Compensation Committee of the Company's Board of Directors certifies the achievement of performance targets over a three-year period of January 1, 2021 through December 31, 2023. Once converted to time-vesting restricted stock units, the awards will vest equally on Duly 1, 2024, July 1, 2025 and July 1, 2026. We also granted 63,000 time-vesting restricted stock units on May 3, 2021 at a grant date price of \$24.16.

15. FAIR VALUE MEASUREMENTS

Fair value is an exit price, representing the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants. Assets and liabilities measured at fair value must be categorized into one of three different levels depending on the assumptions (i.e., inputs) used in the valuation:

Level 1 Unadjusted quoted prices in active markets for identical assets or liabilities;

Level 2 Quoted prices in active markets for similar assets and liabilities and inputs that are observable for the asset or liability; or

Level 3 Unobservable inputs, such as discounted cash flow models or valuations.

The determination of where assets and liabilities fall within this hierarchy is based upon the lowest level of input that is significant to the fair value measurement in its entirety.

As of June 27, 2021 and December 27, 2020, the Company held derivative assets and liabilities that were required to be measured at fair value on a recurring basis. Derivative assets and liabilities consist of long and short positions on exchange-traded commodity futures instruments, commodity options instruments, sales contracts instruments, foreign currency instruments to manage translation and remeasurement risk and interest rate swap instruments.

The	following	items	W	ere	measured	at	fair	value	on	a	recurring	basis:
					June 27, 2021					December 27, 2020		
			L	evel 1	Level 2		Total		Level 1	Level 2		Total
								(In thousands)				
Assets:												
Commodity	derivative assets		\$	28,838	_		28,83	8 \$	24,059	_	\$	24,059
Foreign curr	ency derivative assets			2,957	_		2,95	7	2,204	_		2,204
Liabilities:												
Commodity	derivative liabilities			(86,806)	_		(86,80	5)	(6,531)	_		(6,531)
Foreign curr	ency derivative liabilities			(135)	_		(13	5)	(428)	_		(428)
Interest rate	swap derivative liabilities			_	(478)	(47)	3)	_	(640)		(640)
Sales contract derivative liabilities				_	5,133		5,13	3	_	_		_

See "Note 3. Derivative Financial Instruments" for additional information.

The valuation of financial assets and liabilities classified in Level 1 is determined using a market approach, taking into account current interest rates, creditworthiness, and liquidity risks in relation to current market conditions, and is based upon unadjusted quoted prices for identical assets in active markets. The valuation of financial assets and liabilities in Level 2 is determined using a market approach based upon quoted prices for similar assets and liabilities in active markets or other inputs that are observable for substantially the full term of the financial instrument. The valuation of financial assets in Level 3 is determined using an income approach based on unobservable inputs such as discounted cash flow models or valuations. For each class of assets and liabilities not measured at fair value in the Condensed Consolidated Balance Sheets but for which fair value is disclosed, the Company is not required to provide the quantitative disclosure about significant unobservable inputs used in fair value measurements categorized within Level 3 of the fair value hierarchy.

In addition to the fair value disclosure requirements related to financial instruments carried at fair value, accounting standards require interim disclosures regarding the fair value of all of the Company's financial instruments. The methods and significant assumptions used to estimate the fair value of financial instruments and any changes in methods or significant assumptions from prior periods are also required to be disclosed.

The	carrying	amounts	and	estimated	fair	values	of	our	fixed-rate	debt	obligation	recorded	in	the	Condensed	Consolidated	Balance	Sheets	consisted	of	the	following:
												June 27	, 2021			_		De	cember 27, 202	20		
											Carrying Amou	nt		Va	Fair lue		Carrying A	mount		V	Fai alue	r
																(In thousands)						
Fixed-	rate senior n	otes payable	at 5.75%	6, at Level 2 ii	nputs					\$		_	\$	3	_	- \$		(1,001,693)	· \$			(1,024,510)
Fixed-	rate senior n	otes payable	at 5.875	%, at Level 2	inputs						(84	5,508)			(907,73	2)		(845,149)				(911,957)
Fixed-	rate senior n	otes payable	at 4.25%	6, at Level 2 ii	nputs						(99	0,190)			(1,036,250	0)		_				_
Secure	ed loans, at L	evel 3 inputs										(23)			(23	3)		(38)				(38)

See "Note 11. Debt" for additional information.

The carrying amounts of our cash and cash equivalents, derivative trading accounts' margin cash, restricted cash and cash equivalents, accounts receivable, accounts payable and certain other liabilities approximate their fair values due to their relatively short maturities. Derivative assets were recorded at fair value based on quoted market prices and are included in the line item *Prepaid expenses and other current assets* on the Condensed Consolidated Balance Sheets. Derivative liabilities were recorded at fair value based on quoted market prices and are included in the line item *Accrued expenses and other current liabilities* on the Condensed Consolidated Balance Sheets. The fair value of the Company's Level 2 fixed-rate debt obligations was based on the quoted market price at June 27, 2021 or December 27, 2020, as applicable. The fair value of the Company's Level 3 fixed-rate debt obligation was based on discounted cash flows using weighted average cost of debt of 0.5% for the periods ending June 27, 2021 and December 27, 2020.

In addition to assets and liabilities that are recorded at fair value on a recurring basis, the Company records certain assets and liabilities at fair value on a nonrecurring basis. Generally, assets are recorded at fair value on a nonrecurring basis as a result of impairment charges when required by U.S. GAAP. There were no significant fair value measurement losses recognized for such assets and liabilities in the periods reported.

16. RELATED PARTY TRANSACTIONS

Pilgrim's	has	been	and,	in	some	cases,	continues	to Three	be Months Ended	a	party to	ce	rtain	transactions	with Six Mont		liated	companies.
							June 27, 2021			June	28, 2020		J	une 27, 2021			June 28, 20	20
											(I	n thousand	ls)					
Sales to relate																		
JBS USA F						\$		4,017	\$		3,094		\$		082	\$		6,547
	lia Pty. Ltd.							939			495			, ,	322			1,281
	ercado de Co	ongelados						326			414				777			487
JBS Chile I					_			(1)			(44)				87			(44)
Total sales to	related parti	es			=	\$		5,281	\$		3,959		\$	9,7	768	\$		8,271
									Three Mor	nths Ended	1				Six Mont	hs Ended		
							Jı	une 27, 2021		inio Enaca	June 28, 2020			June 27, 2021	OLI MONE	no Linucu	June 28, 2020	
												(In tho	usands)	·			-	
Cost of goods purchased	d from relate	ed parties																
JBS USA Food Comp	pany ^(a)						\$		- ,-	\$		35,913	\$		105,796	\$		72,810
Penasul UK LTD									2,838			_			5,156			_
Seara Meats B.V.									643			1,080			2,344			3,723
Planterra Food Comp									150			_			150			_
JBS Global (U.K.) Lt	td.								126			219			495			445
JBS Food Trading (Sl	hanghai) Lir	mited							42			_			42			_
JBS Toledo NV									_			93			_			156
JBS Asia Co Limited															5			
Total cost of goods pure	chased from	related parties	5				\$		53,346	\$		37,305	\$		113,988	\$		77,134
									Three Mor	nths Ended	l				Six Mont	ths Ended		
							Jı	une 27, 2021			June 28, 2020			June 27, 2021			June 28, 2020	1
												(In tho	usands)					
Expenditures paid by re																		
JBS USA Food Comp							\$		- ,	\$		13,892	\$		40,732	\$		21,973
Seara Food Europe H	-						.		2			2			12			2
Total expenditures paid	by related p	arties					\$		16,989	\$		13,894	\$		40,744	\$		21,975
									Three Mor	nths Ended	ı				Six Mont	ths Ended		
							Jı	ine 27, 2021		inio Enaca	June 28, 2020			June 27, 2021	OLI MOII	uis Enucu	June 28, 2020	1
							-				·	(In tho	usands)	-				
Expenditures paid on be	ehalf of relat	ted parties																
JBS USA Food Comp	pany ^(b)						\$		11,072	\$		4,206	\$		27,446	\$		6,626
												June 27, 20	21			Docombo	r 27, 2020	
										-		June 27, 20	21	(In thousand	s)	Decembe	1 27, 2020	
Accounts receivable fro	m related pa	arties												,	,			
JBS USA Food Comp										\$				803 \$				714
Combo, Mercado de														35				_
JBS Australia Pty. Ltd	d.													29				370
Seara International L	td.													1				_
Total accounts receivable	le from relat	ed parties								\$				868 \$				1,084
										_	·	•	·		·	·		

	Jun	27, 2021	December 27, 2020
	·	(In thousands)	
Accounts payable to related parties			
JBS USA Food Company ^(a)	\$	6,844 \$	8,562
Penasul UK LTD		1,144	_
Seara Meats B.V.		429	1,075
JBS Global (U.K.) Ltd.		125	5
JBS Food Trading (Shanghai) Limited		42	_
JBS Chile Ltda.		11	8
Total accounts payable to related parties	\$	8,595 \$	9,650

- (a) The Company routinely executes transactions to both purchase products from JBS USA Food Company ("JBS USA") and sell products to them. As of June 27, 2021, approximately \$1.4 million of goods purchased from JBS USA were in transit and not reflected on our Condensed Consolidated Balance Sheet.

 (b) The Company has an agreement with JBS USA to allocate costs associated with JBS USA's procurement of SAP licenses and maintenance services for its combined companies. Under this agreement, the fees associated with procuring SAP licenses and maintenance services are allocated between the Company and JBS USA in proportion to the percentage of licenses used by each company. The agreement expires on the date of expiration, or earlier termination, of the underlying SAP license agreement. The Company also has an agreement with JBS USA to allocate the costs of supporting the business operations by one consolidated corporate team, which have historically been supported by their respective corporate teams. Expenditures paid by JBS USA on behalf of the Company will be reimbursed by the Company and expenditures paid by the Company on behalf of JBS USA will be reimbursed by JBS USA. This agreement expires on December 31, 2021.

17. REPORTABLE SEGMENTS

The Company operates in three reportable segments: U.S., U.K. and Europe, and Mexico. The Company measures segment profit as operating income. Corporate expenses are allocated to the Mexico and U.K. and Europe reportable segments based upon various apportionment methods for specific expenditures incurred related thereto with the remaining amounts allocated to the U.S. reportable segment.

We conduct separate operations in the continental U.S. and in Puerto Rico. For segment reporting purposes, the Puerto Rico operations are included in the U.S. reportable segment. The chicken products processed by the U.S. reportable segment are sold to foodservice, retail and frozen entrée customers. The segment's primary distribution is through retailers, foodservice distributors and restaurants.

The U.K. and Europe reportable segment processes primarily chicken and pork products that are sold to foodservice, retail and frozen entrée customers. The segment's primary distribution is through retailers, foodservice distributors and restaurants.

The chicken products processed by the Mexico reportable segment are sold to foodservice, retail and frozen entrée customers. The segment's primary distribution is through retailers, foodservice distributors and restaurants.

Additional information regarding reportable segments is as follows:

	Three Months Ended				Six Months Ended				
	Jı	me 27, 2021 ^(a)		June 28, 2020 ^(b)		June 27, 2021 ^(c)		June 28, 2020 ^(d)	
				(In					
Net sales									
U.S.	\$	2,248,470	\$	1,798,689	\$	4,248,029	\$	3,725,569	
U.K. and Europe		935,845		757,201		1,790,579		1,579,463	
Mexico		453,383		268,133		872,515		593,919	
Total	\$	3,637,698	\$	2,824,023	\$	6,911,123	\$	5,898,951	

For the three months ended June 27, 2021, the U.S. reportable segment had intercompany sales to the Mexico reportable segment of \$82.8 million. These sales consisted of fresh products, prepared products and grain. For the three months ended June 28, 2020, the U.S. reportable segment had intercompany sales to the Mexico reportable segment of \$57.1 million. These sales consisted of fresh products, prepared products and grain. For the six months ended June 27, 2021, the U.S. reportable segment had intercompany sales to the Mexico reportable segment of \$10.8 million. These sales consisted of fresh products, prepared products and grain. For the six months ended June 28, 2020, the U.S. reportable segment had intercompany sales to the Mexico reportable segment of \$113.7 million. These sales consisted of fresh products, prepared products and grain.

		Three Mor	nths Ended	Six Mon	ths Ended
	J	une 27, 2021	June 28, 2020	June 27, 2021	June 28, 2020
			(In th	ousands)	
Reportable segment profit (loss)					
U.S.	\$	(224,171)	\$ 39,448	\$ (156,046)	\$ 124,500
U.K. and Europe		21,831	23,185	32,326	46,375
Mexico		79,195	(35,544)	159,025	(59,424)
Eliminations		14	200	28	224
Total operating income (loss)		(123,131)	27,289	35,333	111,675
Interest expense, net of capitalized interest		50,651	32,323	80,985	65,011
Interest income		(842)	(1,158)	(3,208)	(2,848)
Foreign currency transaction loss (gain)		4,145	5,525	6,659	(12,860)
Miscellaneous, net		(770)	(45)	(8,614)	(34,233)
Income (loss) before income taxes		(176,315)	(9,356)	(40,489)	96,605
Income tax expense (benefit)		(9,812)	(2,956)	25,546	35,556
Net income (loss)	\$	(166,503)	\$ (6,400)	\$ (66,035)	\$ 61,049

	 June 27, 2021		December 27, 2020
	(In tho	usands)	
Total assets			
U.S.	\$ 5,502,300	\$	5,189,021
U.K. and Europe	3,108,881		3,034,219
Mexico	1,099,050		1,212,428
Eliminations	(1,961,386)		(1,961,171)
Total assets	\$ 7,748,845	\$	7,474,497

	J	une 27, 2021	December 27, 2020
		(In thousa	nds)
Long-lived assets ^(a)			
U.S.	\$	1,843,412 \$	1,815,460
U.K. and Europe		852,059	842,049
Mexico		281,063	292,651
Eliminations		(3,756)	(3,783)
Total long-lived assets	\$	2,972,778 \$	2,946,377

⁽a) For this disclosure, we exclude financial instruments, deferred tax assets and intangible assets in accordance with ASC 280-10-50-41, Segment Reporting. Long-lived assets, as used in ASC 280-10-50-41, implies hard assets that cannot be readily removed.

18. COMMITMENTS AND CONTINGENCIES

General

The Company is a party to many routine contracts in which it provides general indemnities in the normal course of business to third parties for various risks. Among other considerations, the Company has not recorded a liability for any of these indemnities because, based upon the likelihood of payment, the fair value of such indemnities would not have a material impact on its financial condition, results of operations and cash flows.

Financial Instruments

The Company's loan agreements generally obligate the Company to reimburse the applicable lender for incremental increased costs due to a change in law that imposes (1) any reserve or special deposit requirement against assets of, deposits with or credit extended by such lender related to the loan, (2) any tax, duty or other charge with respect to the loan (except standard income tax) or (3) capital adequacy requirements. In addition, some of the Company's loan agreements contain a withholding tax provision that requires the Company to pay additional amounts to the applicable lender or other financing party, generally if withholding taxes are imposed on such lender or other financing party as a result of a change in the applicable tax law. These increased cost and withholding tax provisions continue for the entire term of the applicable transaction, and there is no limitation on the maximum additional amounts the Company could be obligated to pay under such provisions. Any failure to pay amounts due under such provisions generally would trigger an event of default, and, in a secured financing transaction, would entitle the lender to foreclose upon the collateral to realize the amount due.

Litigation

The Company is subject to various legal proceedings and claims which arise in the ordinary course of business. In the Company's opinion, it has made appropriate and adequate accruals for claims where necessary; however, the ultimate liability for these matters is uncertain, and if significantly different than the amounts accrued, the ultimate outcome could have a material effect on the financial condition or results of operations of the Company.

Tax Claims and Proceedings

During 2014 and 2015 the Mexican Tax Authorities opened a review of Avicola Pilgrim's Pride de Mexico, S.A. de C.V. ("APPM") in regards to tax years 2009 and 2010, respectively. In both instances, the Mexican Tax Authorities claim that controlled company status did not exist for certain subsidiaries because APPM did not own 50% of the shares in voting rights of Incubadora Hidalgo, S. de R.L de C.V. and Commercializadora de Carnes de México S. de R.L de C.V. (both in 2009) and Pilgrim's Pride, S. de R. L. de C.V. (in 2010). As a result, APPM should have considered dividends paid out of these subsidiaries partially taxable since a portion of the dividend amount was not paid from the net tax profit account (*CUFIN*). APPM is currently appealing. Amounts under appeal are \$30.8 million and \$18.6 million for tax years 2009 and 2010, respectively. No loss has been recorded for these amounts at this time.

In re Broiler Chicken Antitrust Litiaation

Between September 2, 2016 and October 13, 2016, a series of purported federal class action lawsuits styled as *In re Broiler Chicken Antitrust Litigation*, Case No. 1:16-cv-08637 were filed with the U.S. District Court for the Northern District of Illinois (the "Illinois Court") against PPC and 19 other producers by and on behalf of direct and indirect purchasers of broiler chickens alleging violations of federal and state antitrust and unfair competition laws. The complaints seek, among other relief, treble damages for an alleged conspiracy among defendants to reduce output and increase prices of broiler chickens from the period of January 2008 to the present. The class plaintiffs have filed three consolidated amended complaints: one on behalf of direct purchasers ("the Direct Purchaser Plaintiff Class") and two on behalf of distinct groups of indirect purchasers. Between December 8, 2017 and July 9, 2021, 77 individual direct action complaints were filed with the Illinois Court by individual direct purchaser entities ("DAPs") naming PPC as a defendant, the allegations of which largely mirror those in the class action complaints. Subsequent amendments to certain complaints added allegations of price fixing and bid rigging on certain sales, which have been stayed by the Illinois Court pending resolution of the original supply reduction conspiracy. On June 17, 2021, the Illinois Court issued a revised scheduling order through trial, which contemplates the close of merits fact discovery for defendants and most plaintiffs on July 31, 2021, with additional discovery of subsequent DAPs proceeding in six month increments following consolidation of each DAP complaint. Expert discovery will proceed from August 31, 2021 through May 13, 2022 with summary judgment briefing beginning on June 10, 2022 and concluding on November 21, 2022. The Court has not yet set a trial date.

On January 11, 2021, PPC announced that it had entered into an agreement to settle all claims made by the putative Direct Purchaser Plaintiff Class ("DPPs"), which the Illinois Court preliminarily approved February 25, 2021. As a result of this agreement, PPC recognized an expense of \$75.0 million within *Selling*, *general and administrative expense* in the Consolidated Statements of Income for the year ended December 27, 2020. Pursuant to this agreement, PPC paid the DPPs this amount during the first quarter of 2021.

On July 16, 2021, PPC and the putative End-User Consumer Indirect Purchaser Plaintiff Class ("EUCPs") jointly informed the Court that they had reached an agreement to settle all claims, subject to Court approval under Rule 23. The EUCPs will file a motion for preliminary, and ultimately final, approval of the proposed settlement in the near future. In addition, PPC has agreed in principle to settle all claims made by the putative Commercial and Institutional Indirect Purchaser Plaintiff Class ("CIIPPs"). PPC and CIIPPs are negotiating a long-form settlement agreement, which the Illinois Court will need to approve. Under the terms of these settlements, PPC has agreed to pay the EUCPs and CIIPPs an aggregate amount of \$120.5 million to release all outstanding claims brought by such Classes. These settlements are subject to the execution of long-form settlement agreements with the respective parties and Illinois Court approval thereafter. As a result of these agreements, PPC recognized this expense within Selling, general and administrative expense in the Consolidated Statements of Income for the three and six months ended June 27, 2021

The settlements with the DPPs, EUCPs and CIIPPs do not cover the claims of the DAPs or other parties who have or will opt out of such settlements (collectively, the "Opt Outs"). PPC will therefore continue to litigate against such Opt Outs and will seek reasonable settlements where they are available. PPC has recognized an expense of \$251.4 million to cover potential settlements with various Opt Outs. The amount accrued is an estimate that is subject to change. PPC recognized this expense within *Selling, general and administrative expense* in the Consolidated Statements of Income for the three and six months ended June 27, 2021.

On February 21, 2017, the Attorney General of Florida ("Florida AG") issued a civil investigative demand ("CID") regarding the broiler chicken market. The CID requests, among other things, data and information related to the acquisition and processing of broiler chickens and the sale of chicken products. PPC is cooperating with the Florida AG in producing documents pursuant to the CID.

On August 6, 2020, the Attorney General of Washington ("Washington AG") issued a CID regarding similar broiler chicken matters that are the subject of the Florida CID. The CID requests, among other things, data and information related to the acquisition and processing of broiler chickens and the sale of chicken products. PPC is cooperating with the Washington AG in producing documents pursuant to the CID.

On each of February 24, 2021 and May 4, 2021, the Attorney General of Louisiana ("Louisiana AG") issued a CID regarding similar broiler chicken matters that are the subject of Florida CID. The CID requests, among other things, data and information related to the acquisition and processing of broiler chickens and the sale of chicken products. PPC is cooperating with the Louisiana AG in producing documents pursuant to the CID.

On September 1, 2020, the Attorney General of New Mexico filed a complaint in the First Judicial District Court in the County of Santa Fe, New Mexico. The complaint alleges the same claims as those made in the *In re Broiler Chicken Antitrust Litigation* under New Mexico state law. PPC filed its answer to the complaint on February 1, 2021.

On February 22, 2021, the Attorney General of Alaska filed a complaint in Superior Court in the Third Judicial District in Anchorage, Alaska. The complaint alleges the same claims as those made in the *In re Broiler Chicken Antitrust Litiqation* under Alaska state law. PPC answered the complaint on June 14, 2021.

Other Claims and Proceedings

On October 10, 2016, Patrick Hogan, acting on behalf of himself and a putative class of persons who purchased shares of PPC's stock between February 21, 2014 and October 6, 2016, filed a class action complaint in the U.S. District Court for the District of Colorado (ne "Colorado Court") against PPC and its named executive officers (the "Hogan Litigation"). The complaint alleges, among other things, that PPC's SEC filings contained statements that were rendered materially false and misleading by PPC's failure to disclose that (1) PPC colluded with several of its industry peers to fix prices in the broiler-chicken market as alleged in the In re Broiler Chicken Antitrust Litigation, (2) its conduct constituted a violation of federal antitrust laws, (3) PPC's revenues during the class period were the result of illegal conduct and (4) that PPC lacked effective internal control over financial reporting. The complaint also states that PPC's industry was anticompetitive and seeks compensatory damages. On April 4, 2017, the Colorado Court appointed another stockholder, George James Fuller, as lead plaintiff. On May 11, 2017, the plaintiff filed an amended complaint, which extended the end date of the putative class period to November 17, 2017. PPC and the other defendants moved to dismiss on June 12, 2017, and the plaintiff filed its opposition on July 12, 2017. On March 14, 2018, the Colorado Court dismissed the plaintiff's complaint without prejudice and issued final judgment in favor of PPC and the other defendants. On April 11, 2018, the plaintiff moved for reconsideration of the Colorado Court's decision and for permission to file a Second Amended Complaint. PPC and the other defendants filed a response to the plaintiff's motion on April 25, 2018. On November 19, 2018, the Colorado Court denied the plaintiff's motion for reconsideration and granted plaintiff leave to file a Second Amended Complaint. On June 8, 2020, the plaintiff filed a Second Amended Complaint against the same defendants.

Table of Contents

based in part on the Indictment (defined below). On July 31, 2020, defendants filed a motion to dismiss the Second Amended Complaint pursuant to Rule 12(b)(6) of the Federal Rules of Civil Procedure. The Colorado Court granted the motion to dismiss on April 19, 2021 and issued judgment in favor of defendants. On May 17, 2021, the plaintiff filed a motion for amended judgment. PPC and the other defendants filed their opposition to the motion for amended judgment on June 7, 2021. The Colorado Court's decision on the motion for the amended judgment is currently pending.

On January 27, 2017, a purported class action on behalf of broiler chicken farmers was brought against PPC and four other producers in the U.S. District Court for the Eastern District of Oklahoma (the "Oklahoma Court") alleging, among other things, a conspiracy to reduce competition for grower services and depress the price paid to growers. Plaintiffs allege violations of the Sherman Antitrust Act and the Packers and Stockyards Act and seek, among other relief, treble damages. The complaint was consolidated with a subsequently filed consolidated amended class action complaint styled as In re Broiler Chicken Grower Litigation, Case No. CIV-17-033-RJS (the "Grower Litigation"). The defendants (including PPC) jointly moved to dismiss the consolidated amended complaint on September 9, 2017 for failure to state a claim under Rule 12(b)(6) of the Federal Rules of Civil Procedure. The Oklahoma Court granted only certain other defendants' motions challenging jurisdiction. On January 6, 2020, the Oklahoma Court denied the pending Rule 12 motion, and lifted the stay on discovery. The case is currently in discovery. On October 6, 2020, the Oklahoma plaintiffs filed a motion with the U.S. Judicial Panel on Multidistrict Litigation (the "JPML") seeking consolidation of a series of copycat complaints filed in September and October 2020 in the U.S. District Courts for the District of Colorado, the District of Kansas, and the Northern District of California. On December 15, 2020, the JPML ordered the transfer of all cases to the Oklahoma Court for consolidated pretrial proceedings. On February 12, 2020, the Oklahoma Court entered a case management order in the multi-district litigation setting a February 1, 2022 deadline for the close of fact discovery. That order also set a deadline of February 1, 2022 for reply briefing. Under the order, motions for summary judgment are to be filed on February 1, 2023, with oppositions and replies due March 22, 2023, and April 12, 2023, respectively.

On March 9, 2017, a stockholder derivative action, DiSalvio v. Lovette, et al., No. 2017 cv. 30207, was brought against all of PPC's directors and its Chief Financial Officer, Fabio Sandri, in the Nineteenth Judicial District Court for the County of Weld in Colorado (the "Weld County Court"). The complaint alleges, among other things, that the named defendants breached their fiduciary duties by failing to prevent PPC and its officers from engaging in an antitrust conspiracy as alleged in the Ino 2017 cv. 30308, was brought against all of PPC's directors and its Chief Financial Officer in the Weld County Court. The Brima complaint contains largely the same allegations as the DiSalvio complaint. The DiSalvio and Brima litigations ("the Derivative Litigation") have been consolidated, and on October 14, 2020, an amended shareholder derivative complaint was filed which alleges, among other things, that the named defendants breached their fiduciary duties by failing to prevent PPC from engaging in an antitrust conspiracy as alleged in the Broiler litigation, the Indictment (as defined below), and other related proceedings; and by failing to prevent the issuance of false and misleading statements as alleged in the Hogan securities litigation and the UFCW securities litigation (as defined below). The consolidated case was stayed, pending the resolution of the motion to dismiss in the Hogan Litigation described above. Following the Colorado Court granting defendants' motion to continue the stay pending the Colorado Court's decision on the motion for amended judgment. The Weld County Court granted the motion to continue the stay on June 22, 2021.

Between August 30, 2019 and October 16, 2019, four purported class action lawsuits were filed in the U.S. District Court for the District of Maryland (the "Maryland Court") against PPC and a number of other chicken producers, as well as WMS Webber, Meng, Sahl and Company and Agri Stats. The plaintiffs seek to represent a nationwide class of processing plant production and maintenance workers ("Plant Workers"). They allege that the consolidation of the Sherman Act and seek damages from January 1, 2009 to the present. On November 12, 2019, the Maryland Court ordered the consolidation of the four cases for pretrial purposes. The defendants (including PPC) jointly moved to dismiss the consolidated complaint on November 22, 2019. Shortly thereafter, the plaintiffs informed the defendants and the Maryland Court that they would be amending their complaint, which they did on December 20, 2019. The consolidated amended complaint asserts largely similar allegations to the pleadings in the consolidated complaint, but was extended to include more class members and turkey processors as well as chicken processors. The defendants filed motions to dismiss the consolidated amended complaint on March 2, 2020. The Maryland Court dismissed PPC and a number of other defendants on September 16, 2020 without prejudice. The plaintiffs subsequently filed amended complaints on November 2, 2020 re-naming PPC and the other dismissed defendants. On June 14, 2021, PPC entered into a binding Settlement Agreement to settle all claims with the putative class for \$29.0 million. As a result of this agreement, PPC recognized this expense within Selling, general and administrative expense in the Consolidated Statements of Income for the three and six months ended June 27, 2021. A motion for preliminary approval of the settlement was filed on July 16, 2021, which the Maryland Court granted on July 20, 2021. The settlement is still subject to final approval by the Maryland Court.

On July 6, 2020, United Food and Commercial Workers International Union Local 464A ("UFCW"), acting on behalf of itself and a putative class of persons who purchased shares of PPC stock between February 9, 2017 and June 3, 2020, filed a class action complaint in the Colorado Court against PPC, and Messrs. Lovette, Penn, and Sandri. The complaint alleges, among other things, that PPC's public statements regarding its business and the drivers behind its financial results were false and misleading due to the defendants' purported failure to disclose its participation in an antitrust conspiracy as alleged in the Broiler litigation and the Indictment (defined below). On September 4, 2020, UFCW and the New Mexico State Investment Council ("NMSIC") filed competing motions to be appointed lead plaintiff under the Private Litigation Securities Reform Act. On March 17, 2021, the court appointed NMSIC as lead plaintiff. On May 26, 2021, NMSIC filed an amended complaint, which shortened the end date of the putative class period to June 3, 2020. PPC and the other defendants moved to dismiss the amended complaint on July 19, 2021.

PPC cannot predict the outcome of these pending litigations nor when they will be resolved. The consequences of the pending litigation matters are inherently uncertain, and adverse actions, judgments or settlements in some or all of these matters may result in materially adverse monetary damages, fines, penalties or injunctive relief against PPC. Any claims or litigation, even if fully indemnified or insured, could damage PPC's reputation and make it more difficult to compete effectively or to obtain adequate insurance in the future.

DO I Antitrust Matter

On July 1, 2019, the U.S. Department of Justice (the "DOJ") issued a subpoena to PPC in connection with its investigation arising from the In re Broiler Chicken Antitrust Litigation. The Company has been cooperating with the DOJ investigation.

On June 3, 2020, PPC learned of an indictment by a Grand Jury in the Colorado Court against Jayson Penn, the chief executive officer and president of PPC at that time, in addition to two former employees of PPC and a former employee of a different company (the "Indictment"). The Indictment alleges that the defendants entered into and engaged in a conspiracy to suppress and eliminate competition by rigging bids and fixing prices and other price-related terms for broiler chicken products sold in the U.S., in violation of Section 1 of the Sherman Antitrust Act, 15 U.S.C. Section 1. On June 4, 2020, PPC learned that Mr. Penn pleaded not guilty to the charges. Effective June 15, 2020, Mr. Penn began a paid leave of absence from PPC. In connection with Mr. Penn's leave of absence, PPC's Board of Directors appointed the chief financial officer of PPC, Fabio Sandri, to serve in the additional role of PPC's interim president and chief executive officer. On September 22, 2020, PPC's Board of Directors appointed Fabio Sandri as PPC's President and Chief Executive Officer in addition to his role as Chief Financial Officer. On September 22, 2020, PPC disclosed that Mr. Penn was no longer with the Company, On February 10, 2021, the Company appointed Matthew Galvanoni as its new Chief Financial Officer, effective March 15, 2021.

On October 6, 2020, PPC learned of a superseding indictment by a Grand Jury in the Colorado Court against former Chief Executive Officer of PPC, William Lovette, one additional former employee of PPC, and four employees of different companies. The superseding indictment alleges similar claims to the Indictment.

On October 13, 2020, the Company announced that it had entered into a plea agreement (the "Plea Agreement") with the DOJ pursuant to which the Company agreed to (1) plead guilty to one count of competition involving sales of broiler chicken products in the U.S. in violation of the Sherman Antitrust Act, 15 U.S.C. § 1, and (2) pay a fine of \$110.5 million. The Company recognized the fine as expense which is included in Selling, general and administrative expense in the Condensed Consolidated Statements of Income for the three months ended September 27, 2020. Under the Plea Agreement, the DOJ agreed not to bring further charges against the Company for any antitrust violation involving the sale of broiler chicken products in the U.S. occurring prior to the date of the Plea Agreement. On February 23, 2021, the Colorado Court approved the Plea Agreement and assessed a fine of \$107.9 million. The Company continues to cooperate with the DOJ in connection with the ongoing federal antitrust investigation into alleged price fixing and other anticompetitive conduct in the broiler chicken industry.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Executive Summary

Overview

We are one of the largest chicken producers in the world, and as a vertically integrated company, we are able to control every phase of the production process, which helps us manage food safety and quality, control margins and improve customer service. This gives us the opportunity to continue to create growth and development opportunities, further increasing our position as a leading domestic and global protein company. With the acquisition of Pilgrim's Pride Limited ("PPL") and Moy Park in 2019 and 2017, respectively, we solidified ourselves as a leading European food company while diversifying our product mix with introduction into the pork market. With the acquisition of GNP in 2017, we further solidified ourselves as a leading poultry company within the U.S.

We reported net loss attributable to Pilgrim's of \$66.5 million, or \$0.27 per diluted common share, and loss before tax totaling \$40.5 million, for the six months ended June 27, 2021. These operating results included net sales of \$6.9 billion, gross profit of \$641.5 million and \$115.0 million of cash used in operating activities. We generated a consolidated operating margin of 0.5% with operating margins of (3.7)%, 1.8% and 18.2% in our U.S., U.K. and Europe, and Mexico reportable segments, respectively. For the six months ended June 27, 2021, we generated EBITDA and Adjusted EBITDA is included below.

Cybersecurity Attack

On May 30, 2021, we determined that we were the target of an organized cybersecurity attack (the "Cyberattack") affecting some of the servers supporting our global IT systems. Upon learning of the intrusion, we contacted federal officials and activated our cybersecurity protocols, including voluntarily shutting down all affected systems to isolate the intrusion, limit the potential infection and preserve core systems. Restoring systems critical to production was prioritized. In addition, encrypted backup servers, which were not affected by the Cyberattack, allowed for a return to full operations within two days. We incurred a loss of approximately \$10.0 million related to the Cyberattack, which included an allocation of \$2.4 million of the total \$11.0 million ransom paid by our parent company.

Our response, IT systems and encrypted backup servers allowed for a rapid recovery from the Cyberattack. As a result, the loss of food produced was limited to less than one day of production. We continue to cooperate with government officials regarding this incident. We are not aware of any evidence that any customer, supplier, employee or financial data has been compromised or misused as a result of the Cyberattack.

Impact of COVID-19

The extensive impact of the pandemic caused by the novel coronavirus ("COVID-19") has resulted and will likely continue to result in significant disruptions to the global economy, as well as businesses and capital markets around the world. In an effort to halt the outbreak of COVID-19, a number of countries, states, counties and other jurisdictions have imposed various measures, including but not limited to, voluntary and mandatory quarantines, stay-athome orders, travel restrictions, limitations on gatherings of people, reduced operations and extended closures of businesses. On April 28, 2020, former President Trump signed an executive order directing the Department of Agriculture to ensure meat and poultry processors in the U.S. continue operations uninterrupted to the maximum extent possible and designating meat and poultry processing plants as critical infrastructure.

As the global spread of the virus began to accelerate late in March of 2020, we began to experience adverse impacts to our business and financial results. The impact of the COVID-19 pandemic included disruptions in supply chain, an increase in both broiler and chick costs and an increase in payroll and benefits costs. With the uncertainty surrounding COVID-19, we believe that we will continue to experience certain disruptions to our business for the remainder of 2021.

During 2021, COVID-19 vaccinations have increased while daily COVID-19 case rates decreased, leading to gradual relaxations of COVID-19 restrictions, such as those directly affecting restaurants' indoor dining capacities and increased consumer mobility. The delivery of the second and third COVID-19 direct relief packages to taxpayers, in addition to extended unemployment benefits, were supportive of consumer income. These same relief packages have been a factor in labor shortages and higher absenteeism at our facilities, which has caused a reduction in chicken production.

The impact of COVID-19 and measures to prevent its spread have affected and continue to affect our business in a number of ways.

- Our workforce. Employee health and safety is our priority. As an essential business in a critical infrastructure industry, we continue to produce chicken and pork products, while coordinating with and implementing guidance from the U.S. Centers for Disease Control and Prevention, the National Institute of Occupational Safety and Health, and local and regional Departments of Health in an effort to keep our employees safe and healthy. Measures we have implemented include, but are not limited to: increasing physical distancing of our employees, where possible, by staggering start and shift breaks, placing on-site tents to create more space for employees at break and at meal times, and installing physical barriers to distance employees while working on production lines; adding temperature and symptom screening stations for employees prior to entering our facilities; increasing personal hygiene practices and providing our employees additional personal protective equipment and sanitation stations; and increasing sanitation of our facilities.
- · Our operations. All of our 60 production facilities are operating. To date, we have not experienced a material impact from a plant closure and our facilities have largely been exempt from government closure orders.
- Demand for our products. COVID-19 and the implementation of restricted living have led to a shift in demand from restaurants to retail grocery stores, with consumers eating more at home due to pandemic restrictions. In our U.S. and Mexico businesses, demand for parts and whole-birds (typically bound for restaurants) and prepared foods (distributed, in part, to schools) has declined, while our U.K. and Europe business, which is more retail focused, has generally seen less of an impact. However, two PPL plants had their export liceness to China suspended due to pandemic issues. In an effort to counter the adverse effects of COVID-19, we have transitioned, where commercially reasonable and possible to do so, our business operations to be in the best position to supply COVID-19 market demands. These efforts have included transferring live supply to case ready, shifting production form and mix from foodservice to retail, increasing capacity utilization of retail packaging equipment, and analyzing export positions.
- Foreign currency exchange rates and commodity prices. During the six months ended June 27, 2021, we experienced increased volatility in foreign currency exchange rates and commodity prices, in part related to the uncertainty from COVID-19, as well as actions taken by governments and central banks in response to COVID-19. We expect continued volatility in foreign currency exchange rates and commodity prices during 2021, though we cannot reasonably estimate the duration, extent or impact of that volatility.
- CARES Act. On March 27, 2020, the U.S. government enacted the CARES Act, which includes modifications to the limitation on business interest expense and net operating loss provisions, and provides a payment delay of employer payroll taxes during 2020 after the date of enactment. As of the June 27, 2021, we have delayed approximately \$52.3 million of employer payroll taxes with 50% due by December 31, 2021 and the remaining 50% by December 31, 2022.

Senior Notes due 2031

On April 8, 2021, we completed a sale of \$1.0 billion aggregate principal amount of 4.25% sustainability-linked senior notes due 2031 ("Senior Notes due 2031"). We used the net proceeds of the sale, together with cash on hand, to redeem our 5.75% senior notes due 2025 ("Senior Notes due 2025"). From and including October 15, 2026, the interest rate payable on the notes will increase to 4.50% per annum unless we timely notify the related indenture trustee that our greenhouse gas emissions intensity reduction target has been satisfaction of the target has been confirmed by a qualified provider of third-party assurance or attestation services appointed by us to review our statement of the greenhouse gas emissions intensity in accordance with its customary procedures.

Additional information regarding the Senior Notes due 2031 is included in "Note 11. Debt."

Announced Acquisition

On June 17, 2021, the Company announced that it has executed an acquisition agreement to acquire the Meats and Meals business of Kerry Consumer Foods in the United Kingdom and Ireland (the "Acquisition"). The Acquisition, which was unanimously approved by Pilgrim's Board of Directors, values the acquired businesses at a £680 million (or approximately \$952 million based on a 1.40 USD/GBP exchange rate as of June 16, 2021) enterprise value.

Kerry Meats is a leading manufacturer of branded and private label meats, meat snacks and food-to-go products in the United Kingdom and Ireland. Kerry Meals is a leading ethnic chilled and frozen ready meals business in the United Kingdom. The combined businesses produced over £725 million in annual sales during the year ended December 31, 2020 and have more than 4,500 team members.

The Acquisition is expected to close in the fourth quarter of 2021 and is subject to routine closing purchase price adjustments (including working capital and net debt adjustments), customer closing conditions and regulatory approvals.

Raw Materials and Pricing

Our U.S. and Mexico segments use corn and soybean meal as the main ingredients for feed production, while our U.K. and Europe segment uses wheat, soybean meal and barley as the main ingredients for feed production.

Market prices for chicken products during the three months ended June 27, 2021 began the period slightly above the 5-year average and maintained levels well above historical norms throughout the period. During the second quarter of 2021, the industry saw higher levels of egg sets, chick placements and production than second quarter of 2020 due to the reductions in prior year resulting from the onset of the COVID-19 pandemic. The result of these increases drove an increase in broiler production relative to levels in the prior year. The continued easing of COVID-19 restrictions and increased vaccinations against COVID-19 throughout the U.S. drove a significantly improved foodservice demand environment for chicken. The retail environment maintained consistent levels as consumers continue to utilize chicken as a staple in their at-home meal preparation. Increased chicken demand coincided with mild production growth and already pressured cold storage inventory levels, which entered the quarter well below the 5-year average, resulting in the continued strength of market prices for chicken products in the three months ended June 27, 2021.

While market prices for chicken products have improved thus far in 2021, prices for the remainder of the year will depend on the recovery of the foodservice industry and the evolution of retail meat demand, influenced by factors such as the COVID-19 pandemic, government regulation and uncertainty surrounding both the general economy and protein supply.

Reportable Segments

We operate in three reportable segments: U.S., U.K. and Europe, and Mexico. We measure segment profit as operating income. Corporate expenses are allocated to the Mexico and U.K. and Europe reportable segments based upon various apportionment methods for specific expenditures incurred related thereto with the remaining amounts allocated to the U.S. For additional information, see "Note 17. Reportable Segments" of our Condensed Consolidated Financial Statements included in this quarterly report.

Results of Operations

Three Months Ended June 27, 2021 Compared to the Three Months Ended June 28, 2020

Net sales. Net sales generated in the three months ended June 27, 2021 increased \$813.7 million, or 28.8%, from net sales generated in the three months ended June 28, 2020. The following table provides net sales information:

				Change from Three Months Ended June 28, 2020				
Sources of net sales		Three Months Ended June 27, 2021 Amount			Amount	Percent		
				ept percent data)				
U.S.		\$	2,248,470	\$	449,781	25.0	%	
U.K. and Europe			935,845		178,644	23.6	%	
Mexico			453,383		185,250	69.1	%	
Total net sales	_	\$	3,637,698	\$	813,675	28.8	%	

U.S. Reportable Segment. U.S. net sales generated in the three months ended June 27, 2021 increased \$449.8 million, or 25.0%, from U.S. net sales generated in the three months ended June 28, 2020 primarily due to an increase in net sales per pound which contributed \$447.7 million, or 24.9 percentage points, to the increase in net sales. The increase in net sales per pound was driven primarily from higher than average chicken commodity prices in the U.S. during the three months ended June 27, 2021. Also contributing to the increase in net sales was an increase from sales volume of \$2.1 million, or 0.1 percentage points.

U.K. and Europe Reportable Segment. U.K. and Europe net sales generated in the three months ended June 27, 2021 increased \$178.6 million, or 23.6%, from U.K. and Europe net sales generated in the three months ended June 28, 2020 primarily due to a favorable impact of foreign currency translation, an increase in net sales per pound and an increase in sales volume. The favorable impact of foreign currency translation contributed \$103.6 million, or 13.7 percentage points, to the increase in net sales and was primarily driven by market recoveries in foodservice from the lessening of restrictions due to the COVID-19 pandemic from prior year. The increase in net sales per pound contributed \$13.5 million, or 1.8 percentage points, to the increase in net sales and was primarily driven by changes to feed and other production input costs.

Mexico Reportable Segment. Mexico net sales generated in the three months ended June 27, 2021 increased \$185.3 million, or 69.1%, from Mexico net sales generated in the three months ended June 28, 2020 primarily due to an increase in net sales per pound of \$119.9 million, or 44.7 percentage points, and a favorable impact of foreign currency remeasurement of \$63.6 million, or 23.7 percentage points. This increase in net sales per pound was driven primarily by higher live chicken commodity prices in Mexico during the three months ended June 27, 2021 in comparison to the three months ended June 28, 2020. Also contributing to the increase in net sales was an increase from sales volume of \$1.8 million, or 0.7 percentage points.

Gross profit. Gross profit increased by \$260.4 million, or 217.2%, from \$119.9 million generated in the three months ended June 28, 2020 to \$380.2 million generated in the three months ended June 27, 2021. The following tables provide information regarding gross profit and cost of sales information:

Three Months Ended June 27,			Change from Three Months Ended June 28, 2020				Three Months Ended			
Components of gross profit		2021		Amount Percent		June 27, 2021		June 28, 2020		
					(In thousands, except p	ercent data)				
Net sales	\$	3,637,698	\$	813,675	28.8	%	100.0	%	100.0	
Cost of sales		3,257,457		553,293	20.5	%	89.5	%	95.8 9	
Gross profit	\$	380,241	\$	260,382	217.2	%	10.5	%	4.2 9	

		Change from Three Months Ended June 28, 2020					
Sources of gross profit	Three Months Ended June 27, 2021	Amount	Percent				
	(In thousands, except percent data)						
U.S.	\$ 240,348	\$ 152,327	173.1 %				
U.K. and Europe	50,045	(6,603)	(11.7)%				
Mexico	89,834	114,844	459.2 %				
Elimination	14	(186)	(93.0)%				
Total gross profit	\$ 380,241	\$ 260,382	217.2 %				

			Change from Three Months Ended June 28, 2020					
Sources of cost of sales			Amount	Percent				
		(In thousands, except percent data)						
U.S.	\$	2,008,122	\$ 297,454	17.4 %				
U.K. and Europe		885,800	185,247	26.4 %				
Mexico		363,549	70,406	24.0 %				
Elimination		(14)	186	93.0 %				
Total cost of sales	\$	3,257,457	\$ 553,293	20.5 %				

U.S. Reportable Segment. Cost of sales incurred by our U.S. operations during the three months ended June 27, 2021 increased \$297.5 million, or 17.4%, from cost of sales incurred by our U.S. segment during the three months ended March 29, 2020. Cost of sales increased primarily because of the impact of increased cost per pound sold of \$281.0 million, or 16.4 percentage points, and increased cost from sales volume of \$16.5 million, or 1.0 percentage point. Included in the increased cost of sales was an \$248.9 million increase in live operations costs and a \$14.0 million driven primarily from higher corn and soy commodity prices, one of our main ingredients in feed, an increase of \$20.6 million in chick costs and an increase of \$17.1 million in contract grower costs. Other factors affecting cost of sales were individually immaterial.

U.K. and Europe Reportable Segment. Cost of sales incurred by our U.K. and Europe operations during the three months ended June 27, 2021 increased \$185.2 million, or 26.4%, from cost of sales incurred by our U.K. and Europe segment during the three months ended June 28, 2020. The increase in cost of sales was primarily from the unfavorable impact of foreign currency translation, increased sales volume and increased cost per pound sold, contributing \$98.0 million, or 14.0 percentage points, \$56.9 million, or 8.1 percentage points, and \$30.3 million, or 4.3 percentage points, respectively, to the increase in cost of sales. The increase in sales volume is primarily from market recoveries in foodservice from the lessening of restrictions due to the COVID-19 pandemic from prior year. The increase in net sales per pound is primarily from changes to feed and other input costs. Other factors affecting cost of sales were individually immaterial.

Mexico Reportable Segment. Cost of sales incurred by our Mexico operations during the three months ended March 28, 2021 increased \$70.4 million, or 24.0%, from cost of sales incurred by our Mexico segment during the three months ended June 28, 2020. This increase was driven by an unfavorable impact of foreign currency remeasurement, increased cost per pound sold and an increase in sales volume of \$51.0 million, or 17.3 percentage points, \$17.5 million, or 6.0 percentage points, and \$1.9 million, or 0.7 percentage points, respectively. The increase in cost per pound sold was primarily driven by increases in com and soy costs, which are used as feed. Other factors affecting cost of sales were individually immaterial.

Operating income. Operating income decreased by \$150.4 million, or 551.2%, from \$27.3 million generated in the three months ended June 28, 2020 to \$123.1 million generated in the three months ended June 27, 2021. The following tables provide information regarding operating income and selling, general and administrative ("SG&A") expense:

	Three Mont	Three Months Ended June 27, Change from Three Months Ended June 28, 2020					Three Months Ended			
Components of operating income		2021		Amount		Percent		June 27, 2021		:0
					(In thousands, except perce	ent data)				
Gross profit	\$	380,241	\$	260,382	217.2	%	10.5	%	4.2	%
SG&A expense		503,372		410,802	443.8	%	13.8	%	3.3	%
Operating income	\$	(123,131)	\$	(150,420)	(551.2)	%	(3.4)	%	1.0	%

			Change from Three Months Ended June 28, 2020					
Sources of operating income	Three Months Ended June 27, 2021		Amount	Percent				
	(In thousands, except percent data)							
U.S.	\$	(224,171) \$	(263,619)	(668.3)%				
U.K. and Europe		21,831	(1,354)	(5.8)%				
Mexico		79,195	114,739	322.8 %				
Eliminations		14	(186)	(93.0)%				
Total operating income	\$	(123,131) \$	(150,420)	(551.2)%				

			Change from Three Months Ended June 28, 2020				
Sources of SG&A expense	Three Months Ended June 27, 2021		Amount	Percent			
	(In thousands, except percent data)						
U.S.	\$	464,519	\$ 415,946	856.3 %			
U.K. and Europe		28,214	(5,249)	(15.7)%			
Mexico		10,639	105	1.0 %			
Total SG&A expense	\$	503,372	\$ 410,802	443.8 %			

U.S. Reportable Segment. SG&A expense incurred by our U.S. reportable segment during the three months ended June 27, 2021 increased \$415.9 million, or 856.3%, from SG&A expense incurred by our U.S. reportable segment during the three months ended June 28, 2020. This increase in SG&A expense resulted primarily from an increase in legal fees of \$12.0 million and \$395.9 million recognized in anticipation of probable settlements of ongoing litigation. Other factors affecting SG&A expense were individually immaterial.

U.K. and Europe Reportable Segment. SG&A expense incurred by our U.K. and Europe reportable segment during the three months ended June 27, 2021 decreased \$5.2 million, or 15.7%, from SG&A expense incurred by our U.K. and Europe segment during the three months ended June 28, 2020. The decrease in SG&A expense was driven primarily by a reduction of \$2.8 million in third-party hardware and software support expenses. Other factors affecting SG&A expense were individually immaterial.

Mexico Reportable Segment. SG&A expense incurred by our Mexico reportable segment during the three months ended June 27, 2021 increased approximately \$105,000, or 1.0%, from SG&A expense incurred by our Mexico segment during the three months ended June 28, 2020. Factors affecting our Mexico segment's SG&A expense were individually immaterial.

Net interest expense. Net interest expense increased to \$49.8 million recognized in the three months ended June 27, 2021 from \$31.2 million recognized in the three months ended June 28, 2020. The increase in net interest expense resulted primarily from a loss on early extinguishment of debt recognized as a component of interest expense of \$24.3 million, partially offset by a decrease in interest expense on outstanding borrowings of \$5.7 million. Average borrowings decreased by \$337.8 million from \$2.66 billion during the three months ended June 28, 2020 to \$2.33 billion during the three months ended June 27, 2021. As a percent of net sales, interest expense in the three months ended June 28, 2020 was 1.4% and 1.1%, respectively.

Income taxes. Income tax benefit increased to \$9.8 million, a 5.6% effective tax rate, for the three months ended June 27, 2021 compared to an income tax benefit of \$3.0 million, a 31.6% effective tax rate, for the three months ended June 28, 2020. The increase in income tax benefit resulted primarily from the recognition of deferred tax expense of \$32.2 million related to enactment of the U.K. tax rate change to 25% effective April 1, 2023, partially offset by a decrease in pre-tax income.

Six Months Ended June 27, 2021 Compared to the Six Months Ended June 28, 2020

Net sales. Net sales generated in the six months ended June 27, 2021 increased \$1.0 billion, or 17,2%, from net sales generated in the six months ended June 28, 2020. The following table provides net sales information:

		Change from Six Months Ended June 28, 2020						
Sources of net sales	Six Months Ended June 27, 2021	Amount	Percent					
	 (In thousands, except percent data)							
U.S.	\$ 4,248,029	\$ 522,460	14.0 %					
U.K. and Europe	1,790,579	211,116	13.4 %					
Mexico	872,515	278,596	6 46.9 %					
Total net sales	\$ 6,911,123	\$ 1,012,172	17.2 %					

U.S. Reportable Segment. U.S. net sales generated in the six months ended June 27, 2021 increased \$522.5 million, or 14.0%, from U.S. net sales generated in the six months ended June 28, 2020 primarily because of an increase in net sales per pound and an increase in sales volume. The increase in net sales per pound contributed \$511.3 million, or 13.7 percentage points, to the increase in net sales. This increase in net sales per pound was driven primarily from higher than average chicken commodity prices in the U.S. during the six months ended June 27, 2021. The increase in sales volume contributed \$11.2 million, or 0.3 percentage points to the increase in net sales.

U.K. and Europe Reportable Segment. U.K. and Europe net sales generated in the six months ended June 27, 2021 increased \$211.1 million, or 13.4%, from U.K. and Europe net sales generated in the six months ended June 28, 2020 primarily because of a favorable impact of foreign currency translation of \$163.6 million, or 10.4 percentage points, an increase in sales volume of \$41.0 million, or 2.6 percentage points, and an increase in net sales per pound of \$6.5 million, or 0.4 percentage points. The increase in sales volume was primarily driven by market recoveries in foodservice from the lessening of restrictions due to the COVID-19 pandemic from prior year.

Mexico Reportable Segment. Mexico net sales generated in the six months ended June 27, 2021 increased \$278.6 million, or 46.9%, from Mexico net sales generated in the six months ended June 28, 2020 primarily because of an increase in net sales per pound of \$260.7 million, or 43.9 percentage points, and the favorable impact of foreign currency remeasurement of \$58.8 million, or 9.9 percentage points. The increase in net sales per pound was driven primarily by higher live chicken commodity prices in Mexico during the six months ended June 27, 2021 in comparison to the six months ended June 28, 2020. The increases from foreign currency remeasurement and net sales per pound were partially offset by a decrease in sales volume of \$40.9 million, or 6.9 percentage points.

Gross profit. Gross profit increased by \$344.5 million, or 116.0%, from \$297.0 million generated in the six months ended June 28, 2020 to \$641.5 million generated in the six months ended June 27, 2021. The following tables provide information regarding gross profit and cost of sales information:

		_	-				Percent of	Net Sales	
			Change from Six Mo	onths Ended June 28, 2020		Six Months Ended			
Components of gross profit	Six Months	Ended June 27, 2021	Amount	Percent		June 27, 2021		June 28, 2020	
•				(In thousands, except pe	ercent data)				
Net sales	\$	6,911,123	\$ 1,012,172	17.2	%	100.0	%	100.0	%
Cost of sales		6,269,639	667,646	11.9	%	90.7	%	95.0	%
Gross profit	\$	641,484	\$ 344,526	116.0	%	9.3	%	5.0	%

			Cha	Change from Six Months Ended June 28, 2020					
Sources of gross profit	Six Mo	nths Ended June 27, 2021	A	Amount	Percent				
		(In thousands, except percent data)							
U.S.	\$	373,207	\$	147,083	65.0 %				
U.K. and Europe		87,853		(20,923)	(19.2)%				
Mexico		180,396		218,562	572.7 %				
Elimination		28		(196)	(87.5)%				
Total gross profit	\$	641,484	\$	344,526	116.0 %				

		Change from Six Months Ended June 28, 2020					
Sources of cost of sales	Six Months Ended June 27, 2021	Amount	Percent				
	 (In thousands, except percent data)						
U.S.	\$ 3,874,822	\$ 375,377	10.7 %				
U.K. and Europe	1,702,726	232,039	15.8 %				
Mexico	692,119	60,034	9.5 %				
Elimination	(28)	196	87.5 %				
Total cost of sales	\$ 6,269,639	\$ 667,646	11.9 %				

U.S. Reportable Segment. Cost of sales incurred by our U.S. operations during the six months ended June 27, 2021 increased \$375.4 million, or 10.7%, from cost of sales incurred by our U.S. segment during the six months ended June 28, 2020. Cost of sales increased primarily because of the impact of increased cost per pound sold of \$364.9 million, or 10.4 percentage points, and increased cost from sales volume of \$10.5 million, or 0.3 percentage points. Included in the increased cost of sales were a \$296.7 million increase in feed costs, \$30.2 million in chick costs and an increase in contract grower costs of \$21.2 million. The increase in feed costs is driven primarily from higher com and soy commodity prices, our main ingredients in feed. Other factors affecting cost of sales were individually immaterial.

U.K. and Europe Reportable Segment. Cost of sales incurred by our U.K. and Europe operations during the six months ended June 27, 2021 increased \$232.0 million, or 15.8%, from cost of sales incurred by our U.K. and Europe segment during the six months ended June 28, 2020. The increase in cost of sales was driven by the unfavorable impact of foreign currency translation, increased cost per pound sold and increased sales volume contributing \$155.6 million, or 10.6 percentage points, \$38.3 million, or 2.6 percentage points, respectively, to the increase in cost of sales. Other factors affecting cost of sales were individually immaterial.

Mexico Reportable Segment. Cost of sales incurred by our Mexico operations during the six months ended June 27, 2021 increased \$60.0 million, or 9.5%, from cost of sales incurred by our Mexico segment during the six months ended June 28, 2020. This increase was primarily because of an increase in cost per pound sold and the unfavorable impact of foreign currency remeasurement of \$57.0 million, or 9.0 percentage points, and \$46.6 million, or 7.4 percentage points, respectively. The increase in cost per pound sold was primarily driven by the cost of grains used as feed. These increases were partially offset by a decrease in sales volume of \$43.6 million, or 6.9 percentage points. Other factors affecting cost of sales were individually immaterial.

Operating income. Operating income decreased by \$76.3 million, or 68.4%, from \$111.7 million generated in the six months ended June 28, 2020 to \$35.3 million generated in the six months ended June 27, 2021. The following tables provide information regarding operating income and selling, general and administrative ("SG&A") expense:

						Percent of Net Sales			
			Change from Six Months Ended June 28, 2020				Six Montl	is Ended	
Components of operating income	Six Months I	Ended June 27, 2021	Amount	Percent		June 27, 2021		June 28, 2020	
 			((In thousands, except percen	t data)				
Gross profit	\$	641,484	\$ 344,526	116.0	%	9.3	%	5.0	%
SG&A expense		606,151	420,868	227.1	%	8.8	%	3.1	%
Operating income	\$	35,333	\$ (76,342)	(68.4)	%	0.5	%	1.9	%

	Change from Six Months Ended June 28, 2020					
Sources of operating income	Six Months Ended June 27, 2021	Amount	Percent			
	 (In thousands, except percent data)					
U.S.	\$ (156,046)	\$ (280,546)	(225.3)%			
U.K. and Europe	32,326	(14,049)	(30.3)%			
Mexico	159,025	218,449	367.6 %			
Eliminations	28	(196)	(87.5)%			
Total operating income	\$ 35,333	\$ (76,342)	(68.4)%			

	Change from Six Months Ended June 28, 2020						
Sources of SG&A expense	Six Months Ended June 27, 2021	Amount	Percent				
	 (In thousands, except percent data)						
U.S.	\$ 529,253	\$ 427,629	420.8 %				
U.K. and Europe	55,527	(6,874)	(11.0)%				
Mexico	21,371	113	0.5 %				
Total SG&A expense	\$ 606,151	\$ 420,868	227.1 %				

U.S. Reportable Segment. SG&A expense incurred by our U.S. reportable segment during the six months ended June 27, 2021 increased \$427.6 million, or 420.8%, from SG&A expense incurred by our U.S. reportable segment during the six months ended June 28, 2020. This increase in SG&A expense resulted primarily from an increase of \$18.1 million in legal fees and \$398.3 million recognized in anticipation of probable settlements in ongoing litigation. Other factors affecting SG&A expense were individually immaterial.

U.K. and Europe Reportable Segment. SG&A expense incurred by our U.K. and Europe reportable segment during the six months ended June 27, 2021 decreased \$6.9 million, or 11.0%, from SG&A expense incurred by our U.K. and Europe segment during the six months ended June 28, 2020. The decrease in SG&A expense was driven primarily by a reduction of \$3.4 million in third-party hardware and software support expenses. Other factors affecting SG&A expense in our U.K. and Europe operations were individually immaterial.

Mexico Reportable Segment. SG&A expense incurred by our Mexico reportable segment during the six months ended June 27, 2021 increased approximately \$113,000, or 0.5%, from SG&A expense incurred by our Mexico segment during the six months ended June 28, 2020. Factors affecting our Mexico segment's SG&A expense were individually immaterial.

Net interest expense. Net interest expense increased to \$77.8 million recognized in the six months ended June 27, 2021 from \$62.2 million recognized in the six months ended June 28, 2020. The increase in net interest expense resulted primarily from a loss on early extinguishment of debt recognized as a component of interest expense of \$24.3 million, partially offset by a decrease in interest expense on average outstanding borrowings of \$8.7 million. Average borrowings decreased by \$206.7 million from \$2.5 billion during the six months ended June 28, 2020 to \$2.3 billion during the six months ended June 27, 2021. As a percent of net sales, interest expense in the six months ended June 27, 2021 and June 28, 2020 was 1.2% and 1.1%, respectively.

Income taxes. Income tax expense decreased to \$25.5 million, a (63.1)% effective tax rate, for the six months ended June 27, 2021 compared to income tax expense of \$35.6 million, a 36.8% effective tax rate, for the six months ended June 28, 2020. The decrease in income tax expense resulted primarily from a decrease in pre-tax income partially offset by the recognition of deferred tax expense of \$32.2 million related to enactment of the U.K. tax rate change to 25% effective April 1, 2023.

Liquidity and Capital Resources

	The	following	table	presents	our	available	sources	of	liquidity	as	of	June	27,	2021:
Sources of Liquidity									Facility Amount		Amount Outstanding		Amount Available	
											(In millions)			
Cash and cash equivalents								\$	-	- \$		— \$		391.8
Borrowing arrangements:														
U.S. Credit Facility(a)									750.0)		40.5		671.0
Mexico Credit Facility(b)									75.7	7		_		75.7
U.K. and Europe Credit Facilities(c))								138.8	3		_		138.8

- Availability under the U.S. Credit Facility is also reduced by our outstanding standby letters of credit. Standby letters of credit outstanding at June 27, 2021 totaled \$38.5 million. The U.S. dollar-equivalent of the facility amount under the Mexico Credit Facility is 575,7 million (MXS1.5 billion). The U.S. dollar-equivalent of the facility amount under the Europe Credit Facilities is \$13.8 million (E100.0 million).

We expect cash flows from operations, combined with availability under our credit facilities, to provide sufficient liquidity to fund current obligations, projected working capital requirements, maturities of long-term debt and capital spending for at least the next twelve months.

Additionally, we have executed an agreement to acquire the meats and meals businesses of Kerry Consumer Foods in the U.K. and Ireland (the "Acquisition"). The Acquisition has valued the acquired businesses at a £680 million (or approximately \$952 million based on a 1.40 USD/GBP exchange rate as of June 16, 2021) enterprise value. The Acquisition is expected to close in the fourth quarter of 2021 and is subject to routine closing purchase price adjustments, customer closing conditions and regulatory approvals. We intend to fund the acquisition through a mix of cash on hand, existing credit facilities and debt issuance.

In July 2021, one of our Mexican subsidiaries received an observation letter from the Mexican Tax Authority asserting a withholding tax liability due in connection with our 2015 acquisition of Provemex Holding LLC and its subsidiaries. Although we do not expect any claims or assessments set forth in the observation letter to result in future cash outlays, we are currently evaluating the claims and assessments as set forth in the observation letter.

Historical Flow of Funds

Cash Flows from Operating Activities		Six Months Ended	ths Ended	
	June	27, 2021	June 28, 2020	
		(In millions)		
Net income (loss)	\$	(66.0) \$	61.0	
Net noncash expenses		176.6	196.1	
Changes in operating assets and liabilities:				
Trade accounts and other receivables		(117.6)	29.9	
Inventories		(173.9)	16.4	
Prepaid expenses and other current assets		(6.0)	(22.1)	
Accounts payable, accrued expenses and other current liabilities		266.5	(122.2)	
Income taxes		46.6	(27.4)	
Long-term pension and other postretirement obligations		(9.5)	(1.9)	
Other operating assets and liabilities		(1.7)	10.8	
Cash provided by operating activities	\$	115.0 \$	140.6	

Net Noncash Expenses

Items necessary to reconcile from net income to cash flow provided by operating activities included net noncash expenses of \$176.6 million for the six months ended June 27, 2021. Net noncash expense items included depreciation and amortization of \$182.3 million, loss on early extinguishment of debt of \$24.3 million, stock-based compensation of \$5.2 million, loan cost amortization of \$2.3 million and accretion of discounts related to Senior Notes of \$0.7 million. These expense items were partially offset by deferred income tax benefit of \$32.8 million, gains on property disposals of \$5.1 million and amortization of premiums related to Senior Notes of \$0.2 million.

Items necessary to reconcile from net income to cash flow provided by operating activities included net noncash expenses of \$196.1 million for the six months ended June 28, 2020. Net noncash expense items included depreciation and amortization of \$164.4 million, deferred income tax expense of \$25.3 million, stock-based compensation expense of \$3.5 million, loan cost amortization of \$2.4 million, an adjustment to a previously recognized gain on a bargain purchase of \$1.7 million, accretion of discounts related to Senior Notes and a loss in equity-method investments and of \$0.5 million and \$0.3 million, respectively. These expense items were partially offset by gains on property disposals and amortization of premiums related to Senior Notes of \$1.6 million and \$0.3 million, respectively.

Changes in Operating Assets and Liabilities

The change in trade accounts and other receivables, including accounts receivable from related parties, represented a \$117.6 million use of cash related to operating activities for the six months ended June 27, 2021. This change primarily resulted from an increase in trade accounts receivable due to customer payment timing and increased sales. The change in trade accounts and other receivables, including accounts receivable from related parties, represented a \$29.9 million source of cash related to operating activities for the six months ended June 28, 2020.

The change in inventories represented a \$173.9 million use of cash related to operating activities for the six months ended June 27, 2021. This change resulted primarily from an increase in our raw materials and work-in-process inventories due to increased feed and chick costs. The change in inventories represented a \$16.4 million source of cash related to operating activities for the six months ended June 28, 2020. This change resulted primarily from a decrease in our semi-processed and work-in-process inventories, partially offset by foreign currency fluctuations in our U.K. and Europe reporting segment.

The change in prepaid expenses and other current assets represented a \$6.0 million use of cash related to operating activities for the six months ended June 27, 2021. This change resulted primarily from a net increase in commodity derivative assets. The change in prepaid expenses and other current assets represented a \$22.1 million use of cash related to operating activities for the six months ended June 28, 2020. This change resulted primarily from a net increase in currency rate derivative assets and an increase in prepaid inventory in our Mexico reporting segment, partially offset by a decrease in value-added tax receivables.

The change in accounts payable, revenue contract liabilities, accrued expenses and other current liabilities, including accounts payable to related parties, represented a \$266.5 million source of cash related to operating activities for the six months ended June 27, 2021. This change resulted primarily from an accrual for probable settlement of ongoing litigation and a net increase in the liability position of commodity derivatives, partially offset by the payments of the previously accrued DOJ agreement and a \$75.0 million legal settlement, as well as payment of incentive compensation and revenue recognized on contract liabilities. The change in accounts payable, revenue contract liabilities, accrued expenses and other current liabilities, including accounts payable to related parties, represented a \$122.2 million use of cash related to operating activities for the six months ended June 28, 2020. This change resulted primarily from the timing of receipt of invoicing and payments, and cash payment of incentive compensation, partially offset by an increase in commodity derivative liabilities.

The change in income taxes, which includes income taxes receivable, income taxes payable, deferred tax assets, deferred tax liabilities, reserves for uncertain tax positions, and the tax components within accumulated other comprehensive loss, represented a \$46.6 million source of cash related to operating activities for the six months ended June 27, 2021. The change in income taxes, which includes income taxes receivable, income taxes payable, deferred tax assets, deferred tax liabilities, reserves for uncertain tax positions, and the tax components within accumulated other comprehensive loss, represented a \$27.00 million use of cash related to operating activities for the six months ended June 28, 2020. These changes were both a result of the timing of estimated tax payments.

Cash Flows from Investing Activities	· ·	• •		
	June 27, 2021			June 28, 2020
	(In millions)			
Acquisitions of property, plant and equipment	\$	(183.8)	\$	(148.2)
Proceeds from property disposals		21.4		9.9
Business acquisition		_		(4.2)
Cash used in investing activities	\$	(162.4)	\$	(142.5)

Capital expenditures were primarily incurred to improve operational efficiencies and reduce costs for the six months ended June 27, 2021 and June 28, 2020. Proceeds from property disposals were primarily from sales of two rocessing plants in our U.K. and Europe reportable segment and a broiler farm in our Mexico reportable segment during the six months ended June 27, 2021.

Cash Flows from Financing Activities	Six Months Ended			
	Jun	ne 27, 2021		June 28, 2020
	(In millions)			
Proceeds from revolving line of credit and long-term borrowings	\$	1,540.3	\$	356.5
Payments on revolving line of credit, long-term borrowings and finance lease obligations		(1,522.4)		(20.1)
Payment on early extinguishment of debt		(21.3)		_
Payment of capitalized loan costs		(8.7)		_
Distribution from Tax Sharing Agreement with JBS USA Food Company Holdings		(0.7)		_
Purchase of common stock under share repurchase program		_		(77.9)
Cash provided by (used in) financing activities	\$	(12.8)	\$	258.5

Proceeds from revolving line of credit and long-term borrowings include the sale of \$1.0 billion of sustainability-linked senior notes due 2031 with the remaining primarily from borrowings on our revolving loan commitments under our U.S. and Mexico Credit Facilities. The net proceeds of the senior notes due 2031 were used, together with cash on hand, to redeem the senior notes due 2025. This redemption of the senior notes due 2025 of \$1.0 billion is the primary driver of the payments on long-term borrowings with the remaining amount primarily from payments on our revolving loan commitments under our U.S. and Mexico Credit Facilities and finance lease obligations.

The payment on early extinguishment of debt is the early tender consideration paid as a result of the redemption of the senior notes due 2025. The payment of capitalized loan costs were those loan costs incurred as a part of the sale of the senior notes due 2031. The Distribution from Tax Sharing Agreement with JBS USA Food Company Holdings is payment of net tax incurred during the tax year 2020 under the tax sharing agreement. During the six months ended June 28, 2020, 4.1 million shares were repurchased under the share repurchase program. For further information relating to the share repurchase program, refer to "Note 12. Stockholders' Equity."

Debt

Our long-term debt and other borrowing arrangements consist of senior notes, revolving credit facilities and other term loan agreements. For a description, refer to "Note 11, Debt,"

Collateral

Substantially all of our domestic inventories and domestic fixed assets are pledged as collateral to secure the obligations under the U.S. Credit Facility.

Recent Accounting Pronouncements

See "Note 1. General" of our Condensed Consolidated Financial Statements included in this quarterly report for additional information relating to these recent accounting pronouncements.

Critical Accounting Policies and Estimates

For a description of our critical accounting policies and estimates, refer to "Part II, Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations-Critical Accounting Policies and Estimates" in our annual report on Form 10-K for the fiscal year ended December 27, 2020, filed with the Securities and Exchange Commission (the "SEC") on February 11, 2021 (the "2020 Annual Report").

Reconciliation of Net Income to EBITDA and Adjusted EBITDA

"EBITDA" is defined as the sum of net income (loss) plus interest, taxes, depreciation and amortization. "Adjusted EBITDA" is calculated by adding to EBITDA certain items of expense and deducting from EBITDA certain items of income that we believe are not indicative of our ongoing operating performance consisting of: (1) foreign currency transaction losses (gains), (2) transaction costs related to business acquisitions, (3) costs related to the DOJ agreement and legal settlements, (4) gains (losses) related to the deconsolidation of subsidiaries, and (5) net income (loss) attributable to noncontrolling interests. EBITDA is presented because it is used by us and we believe it is frequently used by securities analysts, investors and other interested parties, in addition to and not in lieu of results prepared in conformity with U.S. GAAP, to compare the performance of companies. We believe investors would be interested in our Adjusted EBITDA because this is how our management analyzes EBITDA applicable to continuing operations. We also believe that Adjusted EBITDA, in combination with our financial results calculated in accordance with U.S. GAAP, provides investors with additional perspective regarding the impact of certain significant items on EBITDA and facilitates a more direct comparison of its performance with its competitors. EBITDA and Adjusted EBITDA are not measurements of financial performance under U.S. GAAP, EBITDA and Adjusted EBITDA have limitations as analytical tools and should not be considered in isolation or as substitutes for an analysis of our results as reported under U.S. GAAP. Some of the limitations of these measures are:

- · They do not reflect our cash expenditures, future requirements for capital expenditures or contractual commitments;
- They do not reflect changes in, or cash requirements for, our working capital needs;
- They do not reflect the significant interest expense or the cash requirements necessary to service interest or principal payments on our debt;
- Although depreciation and amortization are noncash charges, the assets being depreciated and amortized will often have to be replaced in the future, and EBITDA and Adjusted EBITDA do not reflect any cash requirements for such replacements;
- They are not adjusted for all noncash income or expense items that are reflected in our statements of cash flows;
- EBITDA does not reflect the impact of earnings or charges attributable to noncontrolling interests;
- · They do not reflect the impact of earnings or charges resulting from matters we consider to not be indicative of our ongoing operations; and
- They do not reflect limitations on or costs related to transferring earnings from our subsidiaries to us.

In addition, other companies in our industry may calculate these measures differently than we do, limiting their usefulness as a comparative measure. Because of these limitations, EBITDA and Adjusted EBITDA should not be considered as an alternative to net income as indicators of our operating performance or any other measures of performance derived in

accordance with U.S. GAAP. You should compensate for these limitations by relying primarily on our U.S. GAAP results and using EBITDA and Adjusted EBITDA only on a supplemental basis.

		Six Months Ended June 27, 2021
		(In thousands)
Net loss	\$	(66,035)
Add:		
Interest expense, net		77,777
Income tax expense		25,546
Depreciation and amortization		182,260
EBITDA	·	219,548
Add:		
Foreign currency transaction loss (gain)		6,659
Transaction costs related to business acquisitions		2,545
DOJ agreement and litigation settlements		398,285
Minus:		
Deconsolidation of subsidiary		1,131
Net income attributable to noncontrolling interest		444
Adjusted EBITDA	\$	625,462

ITEM 3. OUANTITATIVE AND OUALITATIVE DISCLOSURES ABOUT MARKET RISK

Market Risk-Sensitive Instruments and Positions

The risk inherent in our market risk-sensitive instruments and positions is primarily the potential loss arising from adverse changes in commodity prices, foreign currency exchange rates, interest rates and the credit quality of available-for-sale securities as discussed below. The sensitivity analyses presented do not consider the effects that such adverse changes may have on overall economic activity, nor do they consider additional actions our management may take to mitigate our exposure to such changes. Actual results may differ.

Commodity Prices

We purchase certain commodities, primarily corn, soybean meal, soybean oil, and wheat, for use as ingredients in the feed we either sell commercially or consume in our live operations. As a result, our earnings are affected by changes in the price and availability of such feed ingredients. We have from time to time attempted to minimize our exposure to the changing price and availability of such feed ingredients using various techniques, including, but not limited to, (1) executing purchase agreements with suppliers for future physical delivery of feed ingredients at established prices and (2) purchasing or selling derivative financial instruments such as futures and options.

For this sensitivity analysis, market risk is estimated as a hypothetical 10% increase in the weighted-average cost of our primary feed ingredients as of the periods presented.

	Three Months Ended June 27, 2021				
		Amount	Impact of 10% Increase in Feed Ingredient Prices		
		(In thousands)			
Feed purchases(a)	\$	1,067,773 \$		106,777	
Feed inventory ^(b)		200,320		20,032	

- a) Based on our feed consumption, a 10% increase in the price of our feed purchases would have increased cost of sales for the three months ended June 27, 2021.
- A 10% increase in ending feed ingredient prices would have increased inventories as of June 27, 2021.

The impact of this fluctuation, if realized, could be mitigated by related commodity hedging activity. However, fluctuations greater than 10% could occur.

	June 27, 2021	
	Amount	Impact of 10% Increase in Commodity Prices
	(In thousands)	
Net commodity derivative assets ^(a)	\$ 40,242 \$	4,024

We purchase commodity derivative financial instruments, specifically exchange-traded futures and options, in an attempt to mitigate price risk related to our anticipated consumption of commodity inputs for the next 12 months. A 10% increase in corn, soybean meal, soybean oil and wheat prices would have resulted in an increase in the fair value of our net commodity derivative asset position, including margin cash, as of June 27, 2021.

Interest Rates

Fixed-rate debt. Market risk for fixed-rate debt is estimated as the potential decrease in fair value resulting from a hypothetical increase in interest rates of 10%. Using a discounted cash flow analysis, a hypothetical 10% increase in interest rates would have decreased the fair value of our fixed-rate debt by \$58.2 million as of June 27, 2021.

Variable-rate debt. Our variable-rate debt instruments represent approximately 21.1% of our total debt as of June 27, 2021. Holding other variables constant, including levels of indebtedness, an increase in interest rates of 25 basis points would have increased our interest expense by an immaterial amount for the three months ended June 27, 2021.

Foreign Currency

Mexico Subsidiaries

Our earnings are also affected by foreign exchange rate fluctuations related to the Mexican peso net monetary position of our Mexico subsidiaries. We manage this exposure primarily by attempting to minimize our Mexican peso net monetary position. We are also exposed to the effect of potential currency exchange rate fluctuations to the extent that amounts are repatriated from Mexico to the U.S. We currently anticipate that the future cash flows of our Mexico subsidiaries will be reinvested in our Mexico operations.

The Mexican peso exchange rate can directly and indirectly impact our financial condition and results of operations. For this sensitivity analysis, market risk is estimated as a hypothetical 10% change in the current exchange rate used to convert Mexican pesos to U.S. dollars as of June 27, 2021. However, fluctuations greater than 10% could occur. No assurance can be given as to how future movements in the Mexican peso could affect our future financial condition

or results

of operations.

		Three Months Ended June 27, 2021					
		oact of 10% Deterioration Exchange Rate		npact of 10% Appreciation n Exchange Rate			
	(In thousands, except for exchange rate data)						
Foreign currency remeasurement gain (loss)	\$	(2,870)	\$	3,508			
Exchange rate of Mexican peso to the U.S. dollar:							
As reported		19.82		19.82			
Hypothetical 10% change		21.81		17.84			

U.K. and Europe Foreign Investments

We are exposed to foreign exchange-related variability of investments and earnings from our U.K. and Europe subsidiaries. Foreign currency market risk is the possibility that our financial results or financial position could be better or worse than planned because of changes in foreign currency exchange rates. For this sensitivity analysis, market risk is estimated as a hypothetical 10% change in exchange rates used to convert U.S. dollars to British pound and to euro, and the effect of this change on our U.K. and Europe foreign investments.

Net Assets. As of June 27, 2021, our U.K. and Europe subsidiaries that are denominated in British pound had net assets of \$2.3 billion. A 10% deterioration in the British pound to U.S. dollar exchange rate would cause a decrease in the net assets of our U.K. and Europe subsidiaries of \$208.2 million. A 10% appreciation in the British pound to U.S. dollar exchange rate would cause an increase in the net assets of our U.K. and Europe subsidiaries of \$254.5 million.

Cash flow hedging transactions. We periodically enter into foreign currency forward contracts, which are designated and qualify as cash flow hedges, to hedge foreign currency risk on a portion of sales generated and purchases made by our U.K. and Europe subsidiary. A 10% weakening or strengthening of the U.S. dollar against the British pound and U.S. dollar against the euro would result in immaterial changes in the fair values of these derivative instruments. No assurance can be given as to how future movements in currency rates could affect our future financial condition or results of operations.

Quality of Investments

Certain retirement plans that we sponsor invest in a variety of financial instruments. We have analyzed our portfolios of investments and, to the best of our knowledge, none of our investments, including money market funds units, commercial paper and municipal securities, have been downgraded, and neither we nor any fund in which we participate hold significant amounts of structured investment vehicles, auction rate securities, collateralized debt obligations, credit derivatives, hedge funds investments, fund of funds investments or perpetual preferred securities. Certain postretirement funds in which we participate hold significant amounts of mortgage-backed securities. However, none of the mortgages collateralizing these securities are considered subprime.

Impact of Inflation

Due to low to moderate inflation in the U.S., the U.K. and Europe, and Mexico and our rapid inventory turnover rate, the results of operations have not been significantly affected by inflation.

Forward Looking Statements

Certain written and oral statements made by our Company and subsidiaries of our Company may constitute "forward-looking statements" as defined under the Private Securities Litigation Reform Act of 1995. This includes statements made herein, in our other filings with the SEC, in press releases, and in certain other oral and written presentations. Statements of our intentions, beliefs, expectations or predictions for the future, denoted by the words "anticipate," "believe," "estimate," "expect," "project," "plan," "imply," "intend," "foresee" and similar expressions, are forward-looking statements that reflect our current views about future events and are subject to risks, uncertainties and assumptions. Such risks, uncertainties and assumptions include the following:

- · The impact of the COVID-19 pandemic, efforts to contain the pandemic and resulting economic downturn on our operations and financial condition;
- Matters affecting the chicken industry generally, including fluctuations in the commodity prices of feed ingredients and chicken;
- Our ability to obtain and maintain commercially reasonable terms with vendors and service providers;
- Our ability to maintain contracts that are critical to our operations;
- · Our ability to retain management and other key individuals;
- · Outbreaks of avian influenza or other diseases, either in our own flock or elsewhere, affecting our ability to conduct our operations and/or demand for our poultry products;
- · Contamination of our products, which has previously and can in the future lead to product liability claims and product recalls;
- Exposure to risks related to product liability, product recalls, property damage and injuries to persons, for which insurance coverage is expensive, limited and potentially inadequate;
- · Changes in laws or regulations affecting our operations or the application thereof;
- Our ability to ensure that our directors, officers, employees, agents, third-party intermediaries and the companies to which we outsource certain of our business operations will comply with anti-corruption laws or other laws governing the conduct of business with government entities;
- New immigration legislation or increased enforcement efforts in connection with existing immigration legislation that cause our costs of business to increase, cause us to change the way in which we do business or otherwise disrupt our operations;
- · Competitive factors and pricing pressures or the loss of one or more of our largest customers;
- Inability to consummate, or effectively integrate, any acquisition or to realize the associated anticipated cost savings and operating synergies;
- Currency exchange rate fluctuations, trade barriers, exchange controls, expropriation and other risks associated with foreign segments, including risks associated with Brexit;
- Restrictions imposed by, and as a result of, Pilgrim's leverage;
- · Disruptions in international markets and distribution channels;
- · The impact of cyber-attacks, natural disasters, power losses, unauthorized access, telecommunication failures, and other problems on our information systems;

- Our ability to maintain favorable labor relations with our employees and our compliance with labor laws;
- · Extreme weather or natural disasters;
- The impact of uncertainties in litigation; and
- Other risks described herein and under "Risk Factors" in our annual report on Form 10-K for the year ended December 27, 2020 as filed with the SEC.

Actual results could differ materially from those projected in these forward-looking statements as a result of these factors, among others, many of which are beyond our control.

In making these statements, we are not undertaking, and specifically decline to undertake, any obligation to address or update each or any factor in future filings or communications regarding our business or results, and we are not undertaking to address how any of these factors may have caused changes to information contained in previous filings or communications. Although we have attempted to list comprehensively these important cautionary risk factors, we must caution investors and others that other factors may in the future prove to be important and affect our business or results of operations.

ITEM 4. CONTROLS AND PROCEDURES

Disclosure Controls and Procedures

Under Rules 13a-15(e) and 15d-15(e) of the Securities Exchange Act of 1934 (the "Exchange Act"), "disclosure controls and procedures" means controls and other procedures that are designed to ensure that information required to be disclosed by the Company in the reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms. Disclosure controls and procedures include, without limitation, controls and procedures designed to ensure that information required to be disclosed by our Company in the reports that it files or submits under the Exchange Act is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

As of June 27, 2021, the Company's management, with the participation of the Company's Chief Executive Officer and Chief Financial Officer, evaluated the effectiveness of the design and operation of the Company's disclosure controls and procedures. Based on that evaluation, the Company's Chief Executive Officer and Chief Financial Officer concluded that, as of June 27, 2021, the Company's disclosure controls and procedures were effective to ensure that information required to be disclosed by the Company in reports that it files or submits under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms, and that information we are required to disclose in our reports filed with the SEC is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control over Financial Reporting

The Company's management, including the Chief Executive Officer and Chief Financial Officer, identified no change in the Company's internal control over financial reporting that occurred during the three months ended June 27, 2021 that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

The information required with respect to this item can be found in Part I, Item 1, Notes to Consolidated Financial Statements, "Note 18. Commitments and Contingencies" in this quarterly report and is incorporated by reference into this Item 1.

ITEM 1A. RISK FACTORS

For a discussion of the Company's potential risks and uncertainties, please see "Part I—Item 1A—Risk Factors" and "Part II—Item 7—Management's Discussion and Analysis of Financial Condition and Results of Operations" in the Company's annual report on Form 10-K for the year ended December 27, 2020 and "Part I—Item 2—Management's Discussion and Analysis of Financial Condition and Results of Operations" herein, in each case as updated by the Company's periodic filings with the SEC. Material changes to the risk factors disclosed in "Part I—Item 1A—Risk Factors" in the annual report on Form 10-K for the year ended December 27, 2020 are included below:

We are increasingly dependent on information technology, and our business and reputation could suffer if we are unable to protect our information technology systems against, or effectively respond to, cyber-attacks, other cyber incidents or security breaches or if our information technology systems are otherwise disrupted.

The proper functioning of our information systems is critical to the successful operation of our business. We rely on information technology networks and systems, including the Internet, to process, transmit, and store electronic and financial information, to manage a variety of business processes and activities, and to comply with regulatory, legal, and tax requirements. We also depend on our information technology infrastructure for digital marketing activities and for electronic communications among our locations, personnel, customers, and suppliers. Although our information systems are protected with robust backup systems, including physical and software safeguards and remote processing capabilities, information systems are still vulnerable to cyber-attacks, natural disasters, power losses, unauthorized access, telecommunication failures, and other problems. In addition, certain software used by us is licensed from, and certain services related to our information systems are provided by, third parties who could choose to discontinue their relationship with us. If critical information systems fail or these systems or related software or services are otherwise unavailable, our ability to process orders, maintain proper levels of inventories, collect accounts receivable, pay expenses, and maintain the security of Company and customer data could be adversely affected. Cyber-attacks and other cyber incidents are occurring more frequently and are constantly evolving in nature and sophistication. We have experienced and expect to continue to experience actual or attempted cyber-attacks of our information technology systems or networks. To date, none of these actual or attempted cyber-attacks has had a material effect on our operations or financial condition.

For example, we determined on May 30, 2021 that we were the target of an organized cybersecurity attack (the "Cyberattack") affecting some of the servers supporting our global IT systems. Upon learning of the intrusion, we contacted federal officials and activated our cybersecurity protocols, including voluntarily shutting down all affected systems to isolate the intrusion, limit the potential infection and preserve core systems. Restoring systems critical to production was prioritized. In addition, encrypted backup servers, which were not affected by the Cyberattack, allowed for a return to full operations within two days. We incurred a loss of approximately \$10.0 million related to the Cyberattack, which included an allocation of \$2.4 million of the total \$11.0 million ransom paid by our parent company.

Our response, IT systems and encrypted backup servers allowed for a rapid recovery from the Cyberattack. As a result, the loss of food produced was limited to less than one day of production. We continue to cooperate with government officials regarding this incident. We are not aware of any evidence that any customer, supplier, employee or financial data has been compromised or misused as a result of the Cyberattack.

Our failure to maintain our cyber-security measures and keep abreast of new and evolving threats may make our systems vulnerable. The potential consequences of a material cyber-security incident include reputational damage, litigation with third parties, regulatory actions, disruption of plant operations, and increased cyber-security protection and remediation costs. There can be no assurance that we will be able to prevent all of the rapidly evolving forms of increasingly sophisticated and frequent cyber-attacks. Moreover, our efforts to address network security vulnerabilities may not be successful, resulting potentially in the theft, loss, destruction or corruption of information we store electronically, as well as unexpected interruptions, delays or cessation of service, any of which would cause harm to our business operations. The vulnerability of our systems and our failure to identify or respond timely to cyber incidents could have an adverse effect on our operations and reputation and expose us to liability or regulatory enforcement actions.

Our operations are subject to general risks of litigation.

We are involved on an ongoing basis in litigation relating to alleged antitrust violations or arising in the ordinary course of business or otherwise. Trends in litigation may include class actions involving consumers, shareholders, employees or injured persons, and claims relating to commercial, labor, employment, antitrust, securities or environmental matters. Litigation trends and the outcome of litigation cannot be predicted with certainty, and adverse litigation trends and outcomes could result in material damages, which could adversely affect our financial condition and results of operations.

For example, between September 2, 2016 and October 13, 2016, a series of purported class action lawsuits were brought against PPC and 19 other defendants by and on behalf of direct and indirect purchasers of broiler chickens alleging violations of federal and state antitrust and unfair competition laws. The complaints sought, among other relief, treble damages for an alleged conspiracy among defendants to reduce output and increase prices of broiler chickens from the period of January 2008 to the present. The class plaintiffs filed three consolidated amended complaints: one on behalf of direct purchasers and two on behalf of distinct groups of indirect purchasers. In March 2021, PPC paid the direct purchasers \$75.0 million, which PPC recognized as an expense during the fourth quarter of fiscal 2020. In June 2021, PPC announced that it had entered into agreements to settle all claims made by the two distinct groups of indirect purchasers. Pursuant to these agreements, PPC agreed to pay the two distinct groups of indirect purchases an aggregate \$120.5 million, which PPC recognized as an expense during the second quarter of fiscal 2021.

Between August 30, 2019 and October 16, 2019, four purported class action lawsuits were brought against PPC, a number of other chicken producers and various other defendants on behalf of a nationwide class of processing plant production and maintenance workers alleging that the defendants conspired to fix and depress the compensation paid to these workers in violation of the Sherman Act and sought damages from January 1, 2009 to the present. In June 2021, PPC announced that it had entered into an agreement to settle all claims made by the workers. Pursuant to this agreement, PPC agreed to pay the workers an aggregate \$29.0 million, which PPC recognized as an expense during the second quarter of fiscal 2021.

In addition, on October 13, 2020, PPC announced that it had entered into a plea agreement with the U.S. Department of Justice pursuant to which the it agreed to plead guilty to one count of conspiracy in restraint of competition involving sales of broiler chicken products in the U.S. in violation of the Sherman Antitrust Act. In March 2021, PPC paid a fine of \$107.9 million.

For additional information, see Part I, Item 1, Notes to Condensed Consolidated Financial Statements, "Note 18, Commitments and Contingencies" in this quarterly report,

The consequences of the litigation matters PPC faces are inherently uncertain, and adverse actions, judgments or settlements in some or all of these matters has resulted and may in the future result in materially adverse monetary damages, fines, penalties, or injunctive relief against PPC. Any claims or litigation, even if fully indemnified or insured, could damage PPC's reputation and make it more difficult to compete effectively or to obtain adequate insurance in the future.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

On October 31, 2018, the Company's Board of Directors approved a \$200.0 million share repurchase authorization. The Company plans to repurchase shares through various means, which may include but are not limited to open market purchases, privately negotiated transactions, the use of derivative instruments and/or accelerated share repurchase programs. The extent to which the Company repurchases its shares and the timing of such repurchases will vary and depend upon market conditions and other corporate considerations, as determined by the Company's management team. The Company reserves the right to limit or terminate the repurchase program at any time without notice. As of June 27, 2021, the Company had repurchased 6,105,444 shares under this program for an aggregate cost of \$110.9 million and an average price of \$18.1707 per share. This program expired on February 16, 2021. Set forth below information regarding our stock repurchases for the three months ended June 27, 2021.

Issuer Purchases of Equity Securities

Period	Total Number of Average Price Shares Purchased Paid per Share				Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	That May Yet Be Pu	nte Dollar Value of the Shares irchased Under the Plans or ograms (a)	
March 29, 2021 through April 25, 2021		\$		6,105,444	\$	89,060,082		
April 26, 2021 through May 30, 2021	_		_	6,105,444		89,060,082		
May 31, 2021 through June 27, 2021	_		_	6,105,444		89,060,082		
Total		\$		6,105,444	\$	89,060,082		

(a) Reflects the remaining dollar value of shares that may yet be repurchased under our share repurchase authorization.

ITEM 6. EXHIBITS

- 3.1 Amended and Restated Certificate of Incorporation of the Company (incorporated by reference from Exhibit 3.1 of the Company's Form 8-A (No. 001-09273) filed on December 27, 2012),
- 3.2 Amended and Restated Corporate Bylaws of Pilgrim's Pride Corporation, as amended (incorporated by reference from Exhibit 3.2 of the Company's Quarterly Report on Form 10-Q (No. 001-09273) filed on November 8, 2017).
- 4.1 Indenture dated as of April 8, 2021 among the Company, Pilgrim's Pride Corporation of West Virginia, Inc., Gold'n Plump Poultry, LLC, Gold'n Plump Farms, LLC, JFC LLC and Regions Bank, as Trustee (incorporated by reference from Exhibit 4.1 of the Company's Current Report on Form 8-K (No. 001-09273) filed on April 9, 2021).
- 10.1 Share Purchase Agreement, dated as of June 17, 2021, by and among Kerry Group PLC, Pilgtim's Pride Corporation, Onix Investments UK Limited and Arkose Investments ULC.*
- 31.1 Certification of Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.*
- 31.2 Certification of Principal Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.*
- 32.1 Certification of Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.**
- 32.2 Certification of Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.**

 101.INS
 Inline XBRL Instance Document

 101.SCH
 Inline XBRL Taxonomy Extension Schema

 101.CAL
 Inline XBRL Taxonomy Extension Calculation

 101.DEF
 Inline XBRL Taxonomy Extension Definition

 101.LAB
 Inline XBRL Taxonomy Extension Label

 101.PRE
 Inline XBRL Taxonomy Extension Presentation

104 Cover Page Interactive Data File (formatted as Inline XBRL and contained in Exhibit 101)

Filed herewith.
Furnished herewith.

† Represents a management contract or compensation plan arrangement.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized. PILGRIM'S PRIDE CORPORATION

Date: July 28, 2021

/s/ Matthew Galvanoni
Matthew Galvanoni
Chief Financial Officer and Chief Accounting Officer
(Principal Financial Officer, Chief Accounting Officer and Duly Authorized Officer)

CERTAIN INFORMATION IDENTIFIED WITH [***] HAS BEEN EXCLUDED FROM THIS EXHIBIT BECAUSE IT IS BOTH (I) NOT MATERIAL AND (II) IS THE TYPE THAT THE REGISTRANT TREATS AS PRIVATE OR CONFIDENTIAL.

Dated June 2021

KERRY GROUP PLC

and

PILGRIM'S PRIDE CORPORATION

and

ONIX INVESTMENTS UK LIMITED

and

ARKOSE INVESTMENTS ULC

AGREEMENT FOR THE SALE AND PURCHASE OF THE ENTIRE ISSUED SHARE CAPITAL OF THE TARGET COMPANIES

NORTON ROSE FULBRIGHT

Legal-63238732 172878-0002

Contents

Clau	se Page	
1	Definitions and interpretation	
2	Agreement to sell the Sale Shares	
3	Conditions precedent4	
4	Consideration	
5	Hive Down	
6	Authorisations	
7	Business Receivables, Business Payables and invoicing	
8	Apportionments	
9	Inter-Company Balances	
10	Contracts	
11	Related Party Agreements	
12	Pensions 17	
13	Employment	
14	Completion	
15	The Warranties	
16	Limitations of liability	
17	Restrictive covenants	
18	Insurance	
19	Post-Completion matters	
20	Environmental indemnity	
21	Property Undertaking	
22	Entire Agreement	
23	Effect of Completion	
24	Remedies and waivers	
25	Release	
26	Severance	
27	Set off	
28	Withholdings and gross-up	
29	Payments and interest	
30	Alterations	
31	Counterparts44	
32	Costs, Transfer Taxes and sales taxes	
33	Agreement binding	
34	Agency structure	
35	Assignment	

36	Rights of third parties	46
37	Announcements and confidentiality	46
38	Notices	48
39	Governing law and jurisdiction	50
40	Service of process	51
	edule 1 Designated Sellers, Designated Buyers and Apportionment of Headline F kmark not defined.	Price Error!
Sche	edule 2 Target Companies	53
Sche	edule 3 Properties	58
Sche	edule 4 Seller Warranties	60
Sche	edule 5 Buyer Warranties	83
Sche	edule 6 Limitations on the liability of the Seller	85
Sche	edule 7 Pre-Completion Conduct and Undertakings	94
Sche	edule 8 Completion	99
Sche	edule 9 Completion Statements	102
Sche	edule 10 Agency provisions	114
Sche	edule 11 Definitions and Interpretation	119

Agreed form documents

Tax Covenant

Transitional Services Agreement

Van Sales Distribution Agreement

Data Room Index

Letter of resignation for directors and secretaries

Supplemental Transaction Agency Agreement

Meat Free Substitute Technology Licence

Galtee Licence

Everwell Licence

Raw Material Supply Agreement

- (1) KERRY GROUP PLC, a public limited company incorporated in the Republic of Ireland with company registration number 111471, whose registered office is at Prince's Street, Tralee, Co. Kerry, V92 EH11, Ireland, acting as principal and acting as agent for each Designated Seller (the Seller);
- (2) PILGIRM'S PRIDE CORPORATION, a company incorporated in Delaware, United States of America, whose registered office is at 1770 Promontory Circle, Greeley, Colorado 80634-9038, United States of America, acting as principal and acting as agent for each Designated Buyer (the Buyer);
- (3) ONIX INVESTMENTS UK LIMITED, a company incorporated in England and Wales with company registration number 10934285, whose registered office is at 2nd Floor Building, 1 Imperial Place, Maxwell Road, Borehamwood, Hertfordshire, WD6 1JN, United Kingdom (the UK Designated Buyer); and
- (4) ARKOSE INVESTMENTS ULC, a company incorporated in the Republic of Ireland with company registration number 612127, whose registered address is at 29 Earlsfort Terrace, Dublin 2, D02 AY28, Ireland (the Irish Designated Buyer),

each a Party and together the Parties.

NOW IT IS HEREBY AGREED as follows:

1 Definitions and interpretation

In addition to terms defined elsewhere in this Agreement, the definitions and other provisions in Schedule 11 apply, unless the context requires otherwise.

2 Agreement to sell the Sale Shares

- 2.1 Subject to the Conditions being satisfied (or, in respect of the Condition in clause 3.1(a), satisfied or waived in accordance with the terms of this Agreement), each Designated Seller shall sell, and each Designated Buyer shall buy, the particular Set of Sale Shares for which it is identified as the respective Designated Seller or Designated Buyer in Schedule 1 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price), in each case with full title guarantee and free from all Encumbrances.
- 2.2 Title to, beneficial ownership of, and any risk attaching to, the Sale Shares shall pass on Completion, together with all associated rights and benefits attaching or accruing to them on or

after Completion except as otherwise provided in this Agreement and the other Transaction Documents.

- 2.3 Each of the Seller and each Designated Seller irrevocably and unconditionally waives, and the Seller shall procure that any other relevant member of the Seller's Group shall irrevocably and unconditionally waive, any rights of pre-emption conferred on it by the articles of association of the Target Companies or otherwise over the particular Set of Sale Shares for which it is identified as the respective Designated Seller in Schedule 1 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price).
- 2.4 The Designated Buyers shall not be obliged to complete the purchase of any of the Sale Shares unless the purchase of all of the Sale Shares is completed simultaneously.

3 Conditions precedent

- 3.1 Completion of the sale and purchase of the Sale Shares is conditional on the satisfaction (or, in respect of the Condition in clause 3.1(a), satisfaction or waiver) of the following (the Conditions):
 - (a) confirmation having been received in writing by the Buyer from the United Kingdom Competition and Markets Authority (the CMA) that either:
 - (i) the CMA does not intend to request further information or open an investigation in relation to the transactions contemplated by this Agreement (together, the **Proposed Transaction**) or any matters arising therefrom, after submission by the Buyer of a briefing paper to the CMA's merger intelligence committee setting out the reasons why the CMA should not commence an investigation; or
 - (ii) the CMA does not intend to refer the Proposed Transaction or any matters arising therefrom for a Phase 2 CMA Reference or to seek undertakings in lieu of a Phase 2 CMA Reference under section 73 of the Enterprise Act 2002 except on terms satisfactory to the Buyer and Seller (both acting reasonably) and all conditions contained in such decision necessary to allow Completion having been satisfied or complied with; or
 - (iii) following a Phase 2 CMA Reference of the Proposed Transaction or any matters arising therefrom, the Proposed Transaction may proceed without any undertakings, conditions or orders, or may proceed subject to the giving of such undertakings by, or the imposition of such conditions or orders on, the Buyer or the Seller's Group as shall be satisfactory to the Buyer and the Seller (both acting reasonably) and all

conditions contained in such decision necessary to allow Completion of the transaction having been satisfied or complied with.

(b) ROI Merger Control

- (i) the CCPC having informed the Buyer and Seller in writing in accordance with Section 21(2)(a) of the Competition Act, that the sale and purchase of the Sale Shares may be put into effect and where such determination is subject to any commitments under Section 20(3) of the Competition Act, on terms satisfactory to the Buyer and Seller (both acting reasonably); or
- (ii) the period specified in Section 21(2) of the Competition Act having elapsed without the CCPC having informed the Buyer and Seller in writing of the determination it has made under Section 21(2) of the Competition Acts in relation to the sale and purchase of the Sale Shares; or
- (iii) the CCPC having furnished to the Buyer and Seller a copy of its determination in writing, in accordance with Section 22(3)(a) of the Competition Act, that the sale and purchase of the Sale Shares may be put into effect and where such determination is subject to any commitments under Section 20(3) of the Competition Act, on terms satisfactory to the Buyer and Seller (both acting reasonably); or
- (iv) the CCPC having furnished to the Buyer and Seller a copy of its determination in writing, in accordance with Section 22(3)(c) of the Competition Act, that the sale and purchase of the Sale Shares may be put into effect subject to any conditions specified by the CCPC in relation to the sale and purchase of the Sale Shares being complied with (such conditions being satisfactory to the Buyer and Seller acting reasonably); or
- (v) the period specified in Section 22(4)(a) (including where any such period is suspended by virtue of the application of Section 22(4)(A)) of the Competition Act having elapsed without the CCPC having made a determination in writing under Section 22(3) of the Competition Act in relation to the sale and purchase of the Sale Shares.
- 3.2 The Condition in clause 3.1(a) may be waived by notice in writing from the Buyer to the Seller at the Buyer's sole discretion.
- 3.3 The Buyer agrees to use all reasonable endeavours to ensure that the Conditions are satisfied as soon as reasonably practicable after the date of this Agreement and in any event by the Longstop Date.

- 3.4 Without prejudice to the generality of clause 3.3, the Buyer shall:
 - make promptly all notifications, submissions, responses and communications with the CMA and the CCPC (each, a Competition Authority) as are necessary or desirable to expedite the satisfaction of the Conditions;
 - (b) provide to the Seller a draft of any notification, submission, response or communication before it is submitted to a Competition Authority and to take into account any reasonable comments of the Seller prior to making that notification, submission, response or communication:
 - keep the Seller informed as to the progress of any notification, submission, response or communication made to a Competition Authority, and inform the Seller of all material communications from a Competition Authority;
 - (d) permit the Seller's attendance at meetings with a Competition Authority unless prohibited by the Competition Authority in question;
 - (e) offer, or if proposed by either Competition Authority (as the case may be), accept, any conditions, commitments, undertakings and the like (together, Commitments) that are necessary to obtain satisfaction of the conditions precedent as set out in clauses 3.1(a)(ii) or 3.1(a)(iii) above (in respect of the CMA) or satisfaction of the conditions precedent set out in clauses 3.1(b)(i), 3.1(b)(iii) or 3.1(b)(iv) (in respect of the CCPC), provided those Commitments are not material in the view of the Parties, acting reasonably;
 - (f) when considering offering or accepting any Commitments, allow the Seller to review the draft Commitments and take into account any reasonable comments of the Seller in the Commitments before they are submitted to, or agreed with, a Competition Authority; and
 - (g) provide the Seller with such information about the Buyer (or in respect of commercially sensitive information, to Seller's counsel on a strictly confidential basis) as is reasonably necessary and to which it is legally entitled to enable the Seller to make appropriate submission(s) to a Competition Authority, if any such submission is requested by that Competition Authority, for the satisfaction of the Conditions.
- 3.5 The Seller shall promptly provide to the Buyer (or in respect of commercially sensitive information, to Buyer's counsel on a strictly confidential basis) such information about the Target Companies and the Business as is reasonably necessary to enable the Buyer to make appropriate submissions, notifications and responses to a Competition Authority.

- 3.6 Each Party shall promptly notify the other if at any time it becomes aware of a fact or circumstance that might prevent or materially delay the Conditions being satisfied.
- 3.7 The Buyer shall promptly notify the Seller that the Conditions have been satisfied.
- 3.8 If the Conditions are not satisfied (or, in respect of the Condition in clause 3.1(a), satisfied or waived in accordance with the terms of this Agreement) on or before the Longstop Date, this Agreement shall terminate (except for the Surviving Provisions which shall remain in full force and effect) and no Party shall have any claim against any other except for any prior breach of this Agreement.
- 3.9 The Buyer may, by notice to the Seller, terminate this Agreement at any time before Completion with immediate effect if any Material Adverse Change arises or occurs at any time before Completion, and in such circumstances this Agreement shall terminate (except for the Surviving Provisions which shall remain in full force and effect) and no Party shall have any claim against any other except for any prior breach of this Agreement.
- 3.10 Pending Completion, the Seller shall comply and procure compliance with the Pre-Completion Undertakings.

4 Consideration

- 4.1 The consideration for the sale of the Sale Shares shall be the payment by the relevant Designated Buyer (or, failing which, the Buyer) to the Seller of the Purchase Price.
- 4.2 The Purchase Price shall be payable as follows:
 - (a) at Completion, each Designated Buyer (or, failing which, the Buyer) shall pay the amount of the Initial Purchase Price apportioned (pursuant to clause 4.3 below) to the Sale Shares in respect of which it is the Designated Buyer in accordance with paragraph 1 of Part B of Schedule 8 (Completion); and
 - (b) after Completion, such further payments as are provided for under clauses 4.9 to 4.11 (inclusive) shall be made by the relevant Designated Buyer (or the Buyer) to the relevant Designated Seller or by the relevant Designated Seller to the relevant Designated Buyer (as the case may be) in accordance with those clauses.
- 4.3 The Purchase Price shall be apportioned as between the Sale Shares in the amounts set out in column (4) of the table contained in Schedule 1 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price) as to the Headline Price, adjusted by reference to the Estimated Working Capital and Estimated Net Indebtedness to the extent that the relevant

adjustment is attributable to the value of the Sale Shares of a particular Target Company, and further adjusted (pursuant to clause 4.12 below) as required to reflect any payments made pursuant to clauses 4.9 to 4.11 (inclusive).

The Stock Take

- 4.4 The Seller shall, not less than five Business Days prior to Completion, provide the Buyer with its reasonable estimate calculated in good faith of the value of the Stock as at Completion (as a component of the Estimated Working Capital notified in accordance with clause 4.7), which shall be calculated in accordance with Schedule 9 (Completion Statements).
- 4.5 On the Completion Date, the Seller shall procure that, for the purposes of the Working Capital Statement, a stock take (Stock Take) is carried out at each of the Properties in accordance with the customary practices of the Seller's Group and otherwise as reasonably agreed between the Buyer and Seller. The description and quantity of each item of Stock shall be determined by the Seller acting reasonably and in accordance with the past practices of the Seller's Group in respect of the Business for the purposes of the preparation of its annual accounts, and the value of each item of Stock shall be valued in accordance with Schedule 9 (Completion Statements). Any Stock that is held at a third party premises (Third Party Stock) shall be verified by third party statements in accordance with ordinary business practices.
- 4.6 Without prejudice to the Buyer's right to dispute any items in the Draft Completion Statements pursuant to Schedule 9 (Completion Statements), the Buyer and its Representatives shall be invited, and the Seller shall procure that the Buyer is provided with sufficient opportunity, to be present at the Properties during the stocktaking activities relating to the Stock Take and observe the Stock Take being conducted. The Seller shall, and shall procure that other members of the Seller's Group shall, provide the Buyer and its Representatives with all documents, books, records and other information within the power, possession or control of the Seller's Group that they may reasonably request in connection with the Stock Take and the verification of the Third Party Stock.

Determination of the Purchase Price

- 4.7 The Seller shall, not less than five Business Days prior to Completion, provide the Buyer with its reasonable estimate calculated in good faith of the Estimated Net Indebtedness and the Estimated Working Capital, which shall be calculated in accordance with Schedule 9 (Completion Statements).
- 4.8 The provisions of Schedule 9 (Completion Statements) shall apply in respect of the preparation and finalisation of the Completion Statements and the consequent determination of the Purchase Price.

- 4.9 If the Actual Working Capital, as shown by the Working Capital Statement, is:
 - (a) less than the Estimated Working Capital, the relevant Designated Seller shall pay to the relevant Designated Buyer in accordance with clause 4.11 an amount equal to the difference; or
 - (b) greater than the Estimated Working Capital, the relevant Designated Buyer shall pay to the relevant Designated Seller in accordance with clause 4.11 an amount equal to the difference; or
 - (c) equal to the Estimated Working Capital, no payment shall be due from the Seller, the Buyer, any Designated Seller(s) or any Designated Buyer(s) pursuant to this clause 4.9.
- 4.10 If the Actual Net Indebtedness, as shown by the Net Debt Statement, is:
 - (a) greater than the Estimated Net Indebtedness, the relevant Designated Seller shall pay to the relevant Designated Buyer in accordance with clause 4.11 an amount equal to the difference; or
 - (b) less than the Estimated Net Indebtedness, the relevant Designated Buyer shall pay to the relevant Designated Seller in accordance with clause 4.11 an amount equal to the difference; or
 - (c) equal to the Estimated Net Indebtedness, no payment shall be due from the Seller, the Buyer, any Designated Seller or the any Designated Buyer pursuant to this clause 4.10.
- 4.11 Any payments which the Buyer, the Seller, each Designated Buyer and each Designated Seller are respectively obliged to pay to another Party under clauses 4.9 and 4.10 shall be aggregated or set off against each other, as the case may be. Any resulting sums shall be paid by the relevant Designated Buyer or the relevant Designated Seller (as the case may be, or failing which the Buyer or Seller, as the case may be) within five Business Days following the date of finalisation of the Completion Statements. Any such payment shall be made in accordance with clause 29 (Payments and interest).
- 4.12 For the purposes of clause 4.2(b) and clauses 4.9 to 4.11, relevant Designated Buyer or relevant Designated Seller shall mean:
 - (a) to the extent that the relevant adjustment is attributable to the value of the Sale Shares of a particular Target Company, the Designated Buyer that has purchased those Sale Shares or the Designated Seller that has sold those Sale Shares (as appropriate);

- (b) to the extent that the relevant adjustment cannot be attributed to the value of any particular Sale Shares, all Designated Buyers or all Designated Sellers (as appropriate) pro rata to the proportion of the Initial Purchase Price paid (in the case of a Designated Buyer) or received (in the case of a Designated Seller) by each of them.
- 4.13 Any payment made by the Seller, the Buyer, any Designated Seller or by any Designated Buyer pursuant to this Agreement shall, so far as possible, be treated, for the purposes of Tax, as an adjustment to the consideration paid or payable for the Sale Shares to which the payment relates.
- 4.14 The parties agree that all professional or other fees, liabilities, expenses or other costs paid or agreed to be paid or incurred or owing by any member of the Seller's Group since the Accounts Date (including any VAT in respect of such costs) in connection with the implementation of the transactions contemplated by the Transaction Documents and the Hive Down Agreements (including, for the avoidance of doubt, any transaction or retention bonuses payable to any Employee in connection with the implementation of such transactions (including Tax payable as a consequence)) (Transaction Costs), shall be and remain the responsibility of the Seller and the Seller shall be solely obliged to settle all sums comprised therein except if, and to the extent, included in the Completion Statements.

5 Hive Down

- 5.1 Notwithstanding any other provision of this Agreement, the Seller shall indemnify and hold harmless the Buyer against, and shall pay promptly on demand to the Buyer an amount equal to all Losses suffered by, and all Tax paid, payable or liable to be paid by, the Buyer and/or any member of the Buyer's Group as a result of:
 - (a) the Hive Down (or any part of it, including the transfer or termination of any individual's employment in connection with the Hive Down) subsequently being unwound or declared void in whole or in part by a court of competent jurisdiction (whether pursuant to section 238 of the Insolvency Act 1986 (or equivalent provisions in Northern Ireland or the ROI, where applicable) or otherwise); or
 - (b) any failure by any member of the Seller's Group to implement or record, or to procure the implementation or recording of, any part of the Hive Down in accordance with applicable laws; or
 - (c) any failure by any member of the Seller's Group to implement or record, or to procure the implementation or recording of, any part of the Hive Down in accordance with the terms of the Hive Down Agreements; or

(d) any failure by the Seller or any member of the Seller's Group to comply with its obligations to inform and consult with employee representatives of any Employee or group of Employees or with any Employee or group of Employees directly in connection with the Hive Down,

in each case together with any Losses arising to the extent they are due and payable by the Buyer or any member of the Buyer's Group in investigating or preparing any *bona fide* claim brought against the Seller under this clause 5.1 (provided for this purpose, Losses shall not include the internal costs of Buyer's Group employees, and any third party expenses must be properly incurred).

- 5.2 Notwithstanding any other provision of this Agreement, where any Authorisation the subject of the Hive-Down Agreements is not held by or registered or in the name of, and capable of lawful implementation or reliance by, a Newly-Trading Company at Completion:
 - (a) the Seller shall use its commercially reasonable efforts to procure that, as soon as reasonably practicable after Completion, such Authorisation (or the relevant part thereof) is transferred to and/or registered in the name of the relevant Newly-Trading Company where possible; and
 - (b) where such Authorisation cannot be assigned, transferred or re-registered in whole or in part (as relevant):
 - (i) the Seller shall procure that, to the extent possible, the benefit of the Authorisation is held on trust for the relevant Newly-Trading Company until the Newly-Trading Company obtains a replacement, and shall indemnify and hold harmless the Buyer under this Agreement in respect of any Losses incurred by the Buyer, any member of the Buyer's Group and any Newly-Trading Company arising from any failure by the Newly-Trading Company to hold such Authorisation at Completion; and
 - the Buyer agrees it shall use its commercially reasonable efforts to procure such replacement as soon as practical following Completion; and
 - (iii) the Buyer agrees to procure that each Newly-Trading Company complies with the obligations contained in such Authorisation and the requirements of any Environmental Authority made pursuant to such Authorisation pending replacement (to the extent such obligations have been notified in writing to the Buyer sufficiently in advance), and shall indemnify and hold harmless the Seller under this Agreement in respect of any Losses incurred by any member of the Seller's Group which arise in respect of any failure of any Newly-Trading Company or other member of the

Buyer's Group to comply with any such obligations or requirements until such time as a replacement Authorisation is procured, but only to the extent that such failure to comply is caused by circumstances first arising after Completion.

5.3 The Parties shall, and shall procure that their respective Related Undertakings shall, enforce their rights and carry out their obligations under the Hive Down Agreements in a manner which is consistent with the terms of this Agreement. If there is any conflict between the terms of this Agreement and the Hive Down Agreements, the terms of this Agreement shall prevail (as between the Parties to this Agreement and as between any Related Undertakings of either the Buyer or the Seller) to the extent of any inconsistency.

6 Authorisations

- 6.1 Notwithstanding any other provision of this Agreement, where any Authorisation (other than those which are subject to the Hive Down Agreements and to which clause 5.2 shall apply) is not held by or registered or in the name of, and capable of lawful implementation or reliance by, an Existing Company at Completion:
 - (a) the Seller shall use its commercially reasonable endeavours to procure that, as soon as reasonably practicable after Completion, such Authorisation (or the relevant part thereof) is transferred to and/or registered in the name of the relevant Existing Company where possible; and
 - (b) where such Authorisation cannot be assigned, transferred or re-registered in whole or in part (as relevant):
 - (i) the Seller shall procure that, to the extent possible, the benefit of the Authorisation is held on trust for the relevant Existing Company until the Existing Company obtains a replacement:
 - (ii) the Buyer agrees it shall use its commercially reasonable efforts to procure such replacement as soon as practical following Completion; and
 - (iii) the Buyer agrees to procure that each Existing Company complies with the obligations contained in such Authorisation and the requirements of any Environmental Authority made pursuant to such Authorisation pending replacement (to the extent such obligations have been notified in writing to the Buyer sufficiently in advance) and shall indemnify and hold harmless the Seller under this Agreement in respect of any Losses incurred by any member of the Seller's Group which arise in respect of any failure of any Existing Company or other member of the Buyer's

Group to comply with any such obligations or requirements until such time as a replacement Authorisation is procured, but only to the extent that such failure to comply is caused by circumstances first arising after Completion.

7 Business Receivables, Business Payables and invoicing

- 7.1 The Business Receivables shall be and remain the property of the Seller's Group and, subject to clause 8 below, the Seller's Group shall be solely entitled to all sums which are comprised within the Business Receivables in accordance with their respective terms. The Seller's Group shall be solely entitled to collect for its own account the Business Receivables, and, acting reasonably so as not to prejudice the Buyer and the Target Companies' relationships with the counterparties to such Business Receivables, to enforce for its own benefit all securities for the Business Receivables.
- 7.2 The Business Payables shall be and remain the responsibility of the Seller's Group and, subject to clause 8 below, the Seller (or such member of the Seller's Group as the Seller may determine) shall be solely obliged to settle all sums which are comprised within the Business Payables in accordance with their respective terms.
- 7.3 The Parties acknowledge and agree that, following Completion, invoice arrangements for all amounts receivable by or owing to the Buyer (or any member of the Buyer's Group) for goods and/or services supplied by the Business following Completion shall be conducted by the Buyer or otherwise by the Seller solely in accordance with the terms of the Transitional Services Agreement and the Van Sales Distribution Agreement.

8 Apportionments

- 8.1 Where any charges or outgoings are paid or incurred, or any payments are received, by either the Seller's Group or the Target Companies, in respect of Apportionable Items in relation to the Business, and such items are of a periodic nature and relate or are otherwise attributable to a period of time commencing before but ending after the Completion Date, such amount shall be apportioned pro rata on a time basis between the Parties so that such part of the relevant charges, outgoings or payments as is attributable to the period ended at (but excluding) the Completion Date shall be borne by or belong to the Seller and such part of the relevant charges, outgoings or payments as is attributable to the period commencing at (and including) the Completion Date shall be borne by or belong to the Buyer.
- 8.2 The Seller may from time to time send to the Buyer a statement setting out its good faith estimate of any apportionments required to be paid by either the Buyer or the Seller, together with such supporting evidence as the Buyer may require, and the Parties shall promptly verify and pay such

apportionments as are due. Where the Buyer considers an apportionment is due, the Buyer may request the Seller to promptly pay such apportionments. The Buyer and Seller shall confirm all such apportionments by no later than the date falling 18 months after Completion, following which neither the Buyer nor the Seller shall be liable for any apportionments requested after that date.

8.3 No apportionments shall be required in respect of Apportionable Items to the extent they have already been taken into account in the Completion Statements.

9 Inter-Company Balances

- 9.1 In relation to Inter-Company Balances:
 - the Seller shall at least five Business Days prior to Completion send to the Buyer an estimate of each of the Inter-Company Payables, the Inter-Company Receivables and the resulting Inter-Company Balances;
 - (b) the Seller shall procure that, at Completion, each relevant Target Company repays in accordance with applicable law to the relevant member of the Seller's Group (for itself or, as the case may be, as agent for the members of the Seller's Group to which the relevant Inter-Company Payables are owed) the amount in the applicable currency equal to each of the Inter-Company Payables; and
 - (c) the Seller shall procure that, at Completion, each relevant member of the Seller's Group (for itself or, as the case may be, as agent for each relevant member of the Seller's Group) repays in accordance with applicable law to the relevant Target Company to which the relevant Inter-Company Receivables are owed the amount in the applicable currency equal to each of the Inter-Company Receivables,

and the Inter-Company Balances shall be discharged to the extent of such payments.

9.2 The Seller (for itself and on behalf of the Seller's Group) agrees with the Buyer (for itself and as trustee for the Target Companies) to, with effect from Completion, waive any right or claim which it or any member of the Seller's Group may have against any Target Company in respect of any fact, matter or circumstance arising prior to Completion, provided at all times that nothing in this clause 9.2 shall in any way limit or restrict any right or claim under the terms of or in connection with the Transaction Documents or any Continuing Related Party Agreements.

10 Contracts

10.1 Without prejudice to clause 7, if:

- (a) a Customer Agreement, Supply Agreement or Shared Agreement has been documented in a legal contract and is capable of assignment; and
- (b) the Customer Agreement, Supply Agreement or Shared Agreement has not been legally assigned or assigned in equity to a Target Company pursuant to the Hive Down Agreements,
- (a **Transferring Contract**) then, if and to the extent that any obligations owing by any counterparty are outstanding at Completion and remain to be performed in whole or in part, the Seller shall, subject to clause 10.2 below, procure an equitable assignment to the Target Company to which the Transferring Contract relates of the benefit (other than any Business Receivables), subject to the burden (other than any Business Payables), of the relevant Transferring Contract or, in the case of a Shared Agreement, the part of the Shared Agreement that relates exclusively to the Business (as applicable).
- 10.2 Without prejudice to clause 7, if the benefit of any Transferring Contract cannot effectively be assigned to a Target Company pursuant to clause 10.1 (or such an assignment would constitute an event of default under the relevant Transferring Contract) without obtaining any consent, approval, waiver or agreement from any counterparty:
 - (a) this Agreement shall not constitute an equitable assignment or attempted equitable assignment of the benefit of such Transferring Contract; and
 - for a period of six months following Completion, the relevant member of the Seller's Group (b) shall from Completion to the extent that any obligations owing by any counterparty are outstanding at Completion and remain to be performed in whole or in part, hold the benefit, subject to the burden (other than any Business Payables or Business Receivables), of such Transferring Contract or, in the case of a Shared Agreement, the part of the Shared Agreement that relates exclusively to the Business (as applicable), on trust for the relevant Target Company absolutely and the Buyer shall from Completion procure that the Target Company performs all the obligations of the Seller's Group under such Transferring Contract (or in the case of a Shared Agreement, that part that relates exclusively to the Business), and the Seller's Group shall (so far as it lawfully may) give such assistance (but without being required to incur unreasonable costs and without such assistance being to the detriment of the business of the Retained Kerry Group) to the Buyer as the Buyer may reasonably require to enable the Target Company to enforce its rights under such Transferring Contract and shall provide access to all relevant books, documents and other information in relation to such Transferring Contract as the Buyer may reasonably require from time to time.

- 10.3 If the terms of the relevant Transferring Contract do not permit that Transferring Contract to be held on trust for the relevant Target Company or for the Target Company to perform obligations as a sub-contractor or agent of the Seller (or member of the Seller's Group) then the Seller and the Buyer shall use commercially reasonable endeavours to implement an alternative solution which enables the relevant Target Company to receive the full benefit of the Transferring Contract to the extent that any obligations owing by any counterparty are outstanding at Completion and remain to be performed in whole or in part.
- 10.4 Without prejudice to clauses 10.1 and 10.2 above, the Seller shall, and shall procure that other members of the Seller's Group shall, use commercially reasonable endeavours to obtain prior to the Completion Date:
 - a consent to assign each Relevant Named Contract (to the extent required under such contract) signed by the relevant counterparty; and
 - (b) a waiver or consent signed by the relevant counterparty in relation to the change of control provisions in the following agreements as triggered by the Hive Down and the transactions contemplated by this Agreement:
 - the trade mark licence agreement between Rye Valley Foods Limited and Premier Foods Group Limited dated 9 December 2020; and
 - the licensing and manufacturing agreement between, amongst others, Yo! Sushi UK
 Limited and Kerry Foods Limited dated 5 November 2020,

and the Buyer shall provide the Seller with reasonable assistance in such connection.

- 10.5 The Seller shall, and shall procure that other members of the Seller's Group shall, use commercially reasonable endeavours to transfer the relationships with customers and suppliers to the Target Companies following Completion including by:
 - (a) introducing the Buyer and its Representatives to the authorised representatives of the counterparties to the Customer Agreements and the Supply Agreements and taking all other actions reasonably required in order to transition sales or supplies (as applicable) in connection with the Business for the benefit of the Target Companies following Completion;
 - (b) to the extent permitted by applicable law and confidentiality obligations owed to third parties, providing such information and assistance as the Buyer (or any person acting on the Buyer's behalf) may reasonably request in order to facilitate the continuity after Completion of such sale or supplies in connection with the Business;

- (c) to the extent permitted by applicable law and confidentiality obligations owed to third parties, giving to the Buyer such information as it may reasonably require relating to the Business, and passing on any trade enquiry relating to the Business which the Seller (or any member of the Seller's Group) receives as well as any other written notice, correspondence, information or enquiry received in relation to the Business; and
- (d) seeking such consents or waivers (as applicable) from third parties in respect of confidentiality obligations owed by the Seller's Group to such third parties as the Buyer (or any person acting on the Buyer's behalf) may reasonably request in connection with subclauses (b) and (c) above.
- Save as provided above, the Seller shall not be required to give notice, or seek the consent, approval, waiver or agreement from, any counterparty in connection with any assignment (or attempted assignment) of a Transferring Contract.
- 10.7 Without prejudice to clause 5, the terms of this clause 10 shall apply to the transfer of Contracts to the relevant Target Companies under the Hive Down Agreements in addition to the provisions set out therein (for the purposes of this clause Contracts has the meaning defined in the respective Hive Down Agreements).

11 Related Party Agreements

- 11.1 The Seller shall procure that each of the Related Party Agreements in force immediately prior to Completion shall terminate in its entirety with immediate effect upon Completion and the Seller shall procure that each counterparty thereto shall settle all outstanding financial obligations arising out of such Related Party Agreements in accordance with clause 9 above.
- 11.2 Effective on and from Completion, the Buyer (for itself and as trustee for the other members of the Buyer's Group) and the Seller (for itself and as trustee for the other members of the Seller's Group) irrevocably and unconditionally release and discharge each other party to the Related Party Agreement (for itself and as trustee for the other members of the Buyer's Group or the Seller's Group (as applicable)) from all obligations and liabilities under or in connection with the Related Party Agreements and waive any right or claim which it may have under or in connection with the Related Party Agreements.

12 Pensions

12.1 The Seller shall indemnify and hold harmless the Buyer against and from any and all Losses incurred by the Buyer and/or any Target Company that may arise out of or relate to:

- (a) the participation of Noon Products Limited, Spurway Foods Limited, Northfield Foods Limited, Attleborough Foods Limited, Consumer Foods Van Sales Limited, Rollover Limited and Oakhouse Foods Limited in the UK DC Section;
- (b) the participation of Duffy Meats Limited, Consumer Foods Van Sales (Ireland) Limited and Rye Valley Foods Limited in the ROI DC Plan;
- (c) the participation of Consumer Foods Van Sales (Ireland) Limited, Duffy Meats Limited and Rye Valley Foods Limited in the ROI DB Scheme and any liability of Consumer Foods Van Sales (Ireland) Limited to the ROI DB Scheme; and
- (d) any:
 - (i) financial support direction (under section 43 of the Pensions Act 2004); or
 - (ii) contribution notice (under section 38 of the Pensions Act 2004),

issued by the UK Pensions Regulator against the Buyer or a member of the Buyer's Group in relation to the UK DB Section.

- 12.2 Subject to clause 12.3, the provisions of Schedule 6 apply to clause 12.1, other than the words "by no later than the date falling 18 months after the Completion Date" in paragraph 2.2(c) of Schedule 6, which shall instead read:
 - (a) in relation to clause 12.1(a), "by no later than the date falling six years after the Completion
 - (b) in relation to clause 12.1(b), "by no later than the date falling six years after the Completion Date";
 - (c) in relation to clause 12.1(d)(i), "by no later than the date falling two years after the Completion Date"; and
 - (d) In relation to clause 12.1(d)(ii), "by no later than the date falling six years after the Completion Date".
- 12.3 Paragraph 2.2 of Schedule 6 shall not apply to clause 12.1(c).
- 12.4 The Seller shall procure that on or prior to Completion:
 - (a) each of Noon Products Limited, Spurway Foods Limited, Northfield Foods Limited, Attleborough Foods Limited, Consumer Foods Van Sales Limited, Rollover Limited and

Oakhouse Foods Limited shall (if it has not previously done so) enter into a deed by which it agrees with the principal employer and the trustees to participate in and be bound by the provisions of the UK DC Section with effect from the date such Target Company first employed an active member of the UK DC Section and the Seller shall provide a copy of such deed or deeds to the Buyer; and

(b) each of Noon Products Limited, Spurway Foods Limited, Northfield Foods Limited, Attleborough Foods Limited, Consumer Foods Van Sales Limited, Rollover Limited and Oakhouse Foods Limited shall enter into a deed, documenting with effect from Completion its cessation of participation in the UK DC Section and its discharge from all further liabilities to the UK DC Section.

13 Employment

- 13.1 The Seller shall indemnify and hold harmless the Buyer against and from any and all Losses incurred by the Buyer and/or any member of the Buyer's Group relating to (i) the employee aspects of the Hive Down and (ii) (save insofar as the liability arises as a result of an act or omission of the Buyer or a member of the Buyer's Group) the transfer under the Transfer Regulations of the Central Services Employees in both cases to the extent that those Losses relate to:
 - (a) any claims by or on behalf of any individual who is not an Employee that their employment should have transferred to the Target Companies as part of the Hive Down or as part of the transfer of the Central Services Employees prior to Completion (save where that selection has been agreed by the Buyer under the process set out in clauses 13.2 to 13.4 below); and;
 - (b) any claims by or on behalf of any Central Services Employee or group of Central Service Employees or by any employee representative on their behalf relating to a claim that [***] (save where that selection has been agreed by the Buyer under the process set out at clauses 13.2 to 13.4 below) or that the Seller or any member of the Seller's Group failed to comply with its obligations to inform and consult with employee representatives in respect of the transfer of the Central Services Employees under the Transfer Regulations prior to Completion.
- 13.2 As soon as reasonably practicable after the date of this Agreement the Seller will provide the Buyer with the proposed finalised list of Central Services Employees who it is envisaged will transfer to the Target Companies prior to Completion under the Transfer Regulations or otherwise so as to enable the Seller and the Buyer to discuss such proposals [***].

- 13.3 The Seller shall provide the Buyer with any information reasonably requested to understand the basis for the categorisation of any individual as a Central Services Employee.
- 13.4 Any final decision on which [***] will rest with the Seller (acting reasonably and having taken into account any representations made by the Buyer before reaching a final decision). A finalised list of Central Services Employees will be provided by the Seller to the Buyer as soon as practicable following its finalisation.
- 13.5 The Buyer shall indemnify and hold harmless the Seller against and from any and all Losses incurred by the Seller and/or any member of the Seller's Group relating to any Central Services Employee (whether made or brought by any Central Services Employee, group of Central Services Employees or any trade union or employee representatives on their behalf) as a result of:
 - (a) any claim or demand by or on behalf of any Central Services Employee arising from or in connection with their employment or its termination arising from any act or alleged act, or omission of the Buyer or any member of the Buyer's Group; and
 - (b) any claims by or on behalf of any Central Services Employee or group of Central Services Employees or by any trade union or employee representative on their behalf that (including an entitlement of an employee under or consequent to such claim) arising from the failure of the Buyer or a member of the Buyer's Group to provide sufficient information to the Seller or any member of the Seller's Group to allow it or them to comply with its or their obligations to inform and/or consult or arising from or connected with any failure by the Buyer or any member of the Buyer's Group to provide information about the measures or equivalent thereto which the Buyer or a member of the Buyer's Group intends to take or envisages taking or which a transferee employer intends or envisages taking at the direction of the Buyer or a member of the Buyer's Group in relation to any Central Services Employee;
 - (c) any claim in connection with a change in the terms of any Central Services Employee's employment or working conditions proposed, initiated or carried out by the Buyer or a member of the Buyer's Group.

14 Completion

14.1 Completion shall take place on the Business Day immediately following the closing of the Accounting Month during which the Conditions are satisfied (or, in respect of the Condition in clause 3.1(a), satisfied or waived in accordance with the terms of this Agreement) at the offices of Arthur Cox LLP at Ten Earlsfort Terrace, Dublin 2, D02 T380, Ireland (or at such other place

as the Parties may agree on or prior to the Completion Date), provided that if the Conditions become satisfied or waived on a date which is later than the 14th day of the calendar month, Completion shall take place on the Business Day immediately following the closing of the following Accounting Month. For this purpose, "Accounting Month" means the normal monthly accounting period of the Seller's Group (whereby each accounting period (other than year-end) ends on a Saturday), provided always that under no circumstances for the purposes of this Agreement can the relevant Accounting Month close later than the last day of the calendar month following the calendar month during which the Conditions are satisfied or waived.

14.2 The Buyer shall, not less than seven Business Days prior to Completion, finalise the identities of the Designated Buyers. These shall be the companies identified in column (3) of the table contained in Schedule 2 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price) subject to (i) any addition or change established by a new company becoming a Designated Buyer as a result of a Supplemental Transaction Agency Agreement entered into prior to Completion pursuant to Part B Schedule 10 (Agency provisions), (ii) the Buyer itself electing to be a Designated Buyer for one or more or all Sets of Sale Shares or (iii) any other alterations agreed in writing between the Buyer and the Seller prior to Completion.

14.3 At Completion:

- (a) the Seller shall do those things listed in Part A of Schedule 8; and
- (b) the Buyer shall do those things listed in Part B of Schedule 8.
- 14.4 If the Seller or the Buyer (the Affected Party) fails or is unable to comply with any of its obligations under clause 14.2 or Schedule 8 on the date specified in clause 14.1 then the other (the Unaffected Party) may:
 - (a) defer Completion (by notice to the Affected Party) to a date (being a Business Day) not less than ten nor more than 20 Business Days after that date (in which case the provisions of clauses 14.2, 14.3 and 14.5 shall apply to Completion as so deferred); or
 - (b) proceed to Completion so far as practicable but without prejudice to the Unaffected Party's rights where the Affected Party has not complied with its obligations under this Agreement; or
 - (c) terminate this Agreement on such date by notice to the Affected Party.
- 14.5 If the Affected Party fails or is unable to comply with any of its obligations under clause 14.2 or Schedule 8 on any date to which Completion is deferred in accordance with clause 14.4(a), the

Unaffected Party shall have the right, in addition to its rights in clauses 14.4(a) and 14.4(b), to terminate this Agreement on such date by notice to the Affected Party.

14.6 If this Agreement is terminated in accordance with clause 14.4(c) or 14.5, all rights and obligations of the Parties under this Agreement shall end (except for rights and obligations under the Surviving Provisions which shall remain in full force and effect), provided that nothing in this clause 14.6 shall limit any rights or obligations of any Party under this Agreement which have accrued before termination.

15 The Warranties

- 15.1 The Seller warrants to the Buyer that each of the Seller Warranties is true and accurate as at the date of this Agreement. Each of the Fundamental Warranties shall be deemed to be repeated on Completion by reference to the facts and circumstances subsisting at that time on the basis that any reference in the Fundamental Warranties, whether express or implied, to the date of this Agreement is substituted by a reference to that time.
- 15.2 The Seller undertakes to notify the Buyer in writing promptly if it or any other member of the Seller's Group becomes aware of any Event which would constitute a breach of the warranty set out in paragraph 6.3(b) of Schedule 4 if it was repeated at any time before Completion by reference to the facts and circumstances then existing.
- 15.3 The Buyer warrants to the Seller in respect of itself and each Designated Buyer that each of the Buyer Warranties is true and accurate at the date of this Agreement. Each of the Buyer Warranties shall be deemed to be repeated on Completion by reference to the facts and circumstances subsisting at that time on the basis that any reference in the Buyer Warranties, whether express or implied, to the date of this Agreement is substituted by a reference to that time.
- 15.4 The Seller Warranties (other than the Fundamental Warranties) are subject to and qualified by, and the Buyer shall be deemed to have actual knowledge of, those matters Disclosed in the Disclosure Letter.
- 15.5 The Seller Warranties (other than the Fundamental Warranties) are further subject to and qualified by, and the Buyer shall be deemed to have actual knowledge of:
 - (a) all documents and information Disclosed in the Data Room;
 - (b) all information Disclosed in the Due Diligence Reports and the Information Memorandum;

- (c) all information Disclosed, specifically noted or specifically referred to in the Accounts (and the accompanying reports of the directors and the auditors) and the Carve-Out Accounts; and
- (d) all information Disclosed in the Q&A Responses.
- 15.6 In each Seller Warranty, where any statement is qualified as being made "so far as the Seller is aware" or any similar expression, such statement shall be deemed to refer to the actual knowledge of Gavin Caplis, Nick Robinson, William Cadbury, Ronan Deasy, Fergal Ryan, Declan Crowley, Tina Mulvihill and Martin O'Donoghue after such persons have made due and careful enquiry.
- 15.7 Save for the Seller Warranties contained in:
 - (a) paragraphs 7 and 10 of Schedule 4, none of the Seller Warranties shall extend to, or be construed as a warranty relating to the Properties (such matters being the subject of the Seller Warranties contained only in paragraphs 7 and 10 of Schedule 4);
 - (b) paragraph 11 of Schedule 4, none of the Seller Warranties shall extend to, or be construed as a warranty relating to Environmental matters (such matters being the subject of the Seller Warranties contained only in paragraph 11 of Schedule 4); and
 - (c) paragraphs 3, 5.3 and 5.4 (in each case, insofar as the matter relates to or is in connection with Tax) and paragraph 20 of Schedule 4, none of the Seller Warranties shall extend to, or be construed as a warranty relating to Tax (such matters being the subject of the Seller Warranties contained only in paragraphs 3, 5.3 and 5.4 (in each case, insofar as the matter relates to or is in connection with Tax) and paragraph 20 of Schedule 4).
- 15.8 None of the Seller Warranties shall extend to, or be construed as a warranty relating to Excluded Assets, Excluded Contracts and Excluded Liabilities (except in respect of their exclusion).
- 15.9 Without prejudice to the generality of clauses 15.6 and 22 (Entire agreement), the Buyer acknowledges that the only warranties or other assurances of any kind given by or on behalf of the Seller are the Seller Warranties and that accordingly the Seller makes no warranty or representation as to:
 - the accuracy or completeness of any statements (including any statements of opinion) contained in the Disclosure Letter;

- (b) the information or documents made available to the Buyer or its advisers or representatives prior to the date of this Agreement including such as may be contained in the Data Room; and
- (c) any projections, estimates, budgets, statements of intent or statements of opinion delivered to or made available to the Buyer or its advisers or representatives of future revenues, future results of operations (or any component thereof), future cash flows or future financial condition (or any component thereof) of the Target Companies.
- The Seller agrees with the Buyer (for itself and as trustee for the Target Companies and their respective directors, employees, agents and professional advisers) to waive any right or claim which the Seller's Group may have in respect of any misrepresentation or error in, or omission from any information or opinion supplied or given by the Target Companies and/or any of their respective directors, employees, agents or professional advisers in the course of negotiating this Agreement or any other Transaction Document, and that any such right or claim shall not constitute a defence to any claim by the Buyer's Group under or in relation to this Agreement or any other Transaction Document.
- 15.11 The Seller shall not be liable in respect of any Seller Warranty Claim where the Buyer and/or any of its agents and/or any of its professional advisers had actual knowledge of the matter giving rise to that Seller Warranty Claim on or before the date of this Agreement.
- 15.12 The Buyer confirms to the Seller that, save to the extent that it is Disclosed in the Disclosure Letter, there is no material fact, matter, event or circumstance which is actually known to the Buyer which constitutes a breach of a Seller Warranty, and it is not aware of any reason why the Seller's Warranties would be incorrect or untrue in any respect, as at the date of this Agreement.
- 15.13 The Warranties shall not in any respect be extinguished or affected by Completion.
- 15.14 Each of the Warranties shall be construed as a separate and independent warranty and the relevant Parties shall have a separate claim and right of action in respect of every breach of a relevant Warranty.
- 15.15 The satisfaction by the Seller of any claim under this Agreement (including the Seller Warranties) shall (so far as is legally possible) be deemed to constitute a reduction in the consideration payable by the Buyer for the sale of the Sale Shares.

16 Limitations of liability

All Relevant Claims shall be subject to the limitations and other provisions set out in Schedule 6 (*Limitations on the liability of the Seller*).

17 Restrictive covenants

- 17.1 The Seller undertakes to the Buyer that it shall not, and shall procure that no member of the Seller's Group shall, directly or indirectly, alone or with, through or as any manager, adviser, consultant, partner, employee or agent for any person:
 - (a) for a period of two years from Completion carry on any business in the United Kingdom or the Republic of Ireland which is engaged in the manufacture or sale of any Restricted Products in competition with the Business, save that nothing in this clause 17.1(a) shall operate to prohibit the Seller or any member of the Seller's Group (alone or with any associated person) from (i) holding up to five per cent of the shares of any company or group, the shares of which are listed or dealt in on a recognised stock exchange nor (ii) from carrying on ingredients solutions under the Seller's Group's Taste and Nutrition division in support of Restricted Products or otherwise;
 - (b) for a period of two years from Completion, solicit from the Buyer any Senior Employee, whether or not that person would commit any breach of their contract of employment by reason of leaving the service of the Buyer, save that nothing in this clause 17.1(b) shall prevent the Seller or any member of the Seller's Group from:
 - publishing any bona fide recruitment advertisement which is not targeted at any person who was employed by the Seller in the Business at any time during the 12 months immediately prior to Completion;
 - (ii) recruiting any person in response to any such advertisement; or
 - (iii) negotiating with or offering to employ any person who initiates any contact with the Seller or any member of the Seller's Group with regard to such employment; or
 - (c) for a period of two years from Completion, without the prior written consent of the Buyer, use, whether on its own behalf or on behalf of any third party, any Confidential Information or divulge to any third party any such information, save that nothing in this clause 17.1(c) shall prevent the Seller or any member of the Seller's Group from:
 - (i) using or divulging any information which is in or enters the public domain, other than through a breach of the obligations of confidentiality set out in this Agreement; or
 - (ii) disclosing any information to the extent it is required to do so by any applicable law, governmental order, decree, regulation, licence or rule or pursuant to the regulations of any securities exchange or regulatory or governmental body to which it is subject,

in each case provided that any disclosure shall, so far as is practicable, be made only after consultation with the Buyer.

- 17.2 The Buyer undertakes to the Seller that it shall not, and shall procure that no member of the Buyer's Group shall, directly or indirectly, alone or with, through or as any manager, adviser, consultant, partner, employee or agent for any person for a period of two years from Completion, solicit from the Seller's Group any employee of the Seller's Group (other than an Employee) who earns a basic annual salary of £80,000 (or local currency equivalent) or more, whether or not that person would commit any breach of their contract of employment by reason of leaving the service of the Seller's Group, save that nothing in this clause 17.2 shall prevent the Buyer or any member of the Buyer's Group from:
 - (a) publishing any bona fide recruitment advertisement which is not targeted at any person who was employed by the Seller's Group in the Retained Business at any time during the 12 months immediately prior to Completion;
 - (b) recruiting any person in response to any such advertisement; or
 - (c) negotiating with or offering to employ any person who initiates any contact with the Buyer or any member of the Buyer's Group with regard to such employment.
- 17.3 The Seller and the Buyer agree that the restrictive covenants in clauses 17.1 and 17.2 are reasonable and necessary for the protection of the value of the Business or Retained Business (as applicable).
- 17.4 The Seller and the Buyer agree that if the restrictive covenants in clauses 17.1 and 17.2 are adjudged to go beyond what is reasonable in all the circumstances for the protection of the legitimate interests of the Buyer or the Seller (as applicable) but would be adjudged reasonable if part or parts of their wording were deleted or amended or qualified or the periods referred to were reduced or the range of products or area dealt with were reduced in scope, then the relevant restriction or restrictions shall apply with such modification or modifications as may be necessary to make it or them valid and effective.
- 17.5 Save to the extent necessary to protect its rights in a bona fide manner in any legal proceedings, the Buyer undertakes to the Seller that it shall not, and shall procure that no member of the Buyer's Group shall, directly or indirectly, alone or with, through or as any manager, adviser, consultant, partner, employee or agent for any person at any time make, publish or communicate to any person or in any public forum any defamatory or disparaging remarks, comments or statements concerning the Seller's Group, the Retained Business, or the Business in respect of the period prior to Completion.

17.6 Save to the extent necessary to protect its rights in a bona fide manner in any legal proceedings, the Seller undertakes to the Buyer that it shall not, and shall procure that no member of the Seller's Group shall, directly or indirectly, alone or with, through or as any manager, adviser, consultant, partner, employee or agent for any person at any time make, publish or communicate to any person or in any public forum any defamatory or disparaging remarks, comments or statements concerning the Buyer's Group or the Business in respect of the period prior to or following Completion.

18 Insurance

- 18.1 Seller's Group Insurance Policies means the material insurance policies of the Seller's Group (whether such policies are maintained with third party insurers or with other members of the Seller's Group) which relate to the Business or the Target Companies (and any renewals or replacement of such policies).
- 18.2 From the date of this Agreement until (and including) the Completion Date, the Seller shall (and shall procure that the relevant members of the Seller's Group shall) maintain in force and comply with the Seller's Group Insurance Policies in all material respects (including, where required, notifying the insurer of the Proposed Transaction) on the same terms and with a similar level of cover to that prevailing at the date of this Agreement for the benefit of the Target Companies.
- 18.3 If any insured event occurs before Completion of which the Seller becomes aware before Completion in relation to any Target Company or the Business, the Seller shall, provided that to do so would be consistent with the Seller's past practice, give notice to the relevant insurer in accordance with the terms of the relevant Seller's Group Insurance Policy and shall use all reasonable efforts to make recovery under the relevant policy prior to Completion. To the extent that such recovery is made prior to the Completion Date, the Seller shall ensure that the proceeds are dealt with in accordance with procedures normally observed by the Business.
- 18.4 From Completion, the Seller shall use its commercially reasonable endeavours (in each case without incurring further cost, premium or liability on the part of the Seller's Group (but excluding any subsequent increase in premia and/or costs under the Seller's Group Insurance Policies arising as a result of making and/or pursuing any Permitted Claim in accordance with this Agreement)) to ensure that all Seller's Group Insurance Policies which are on a "claims occurring" basis and which are in force at Completion continue in force on the same terms to the extent that (i) they provide cover in relation to any matter or event occurring in relation to any Target Company and/or the Business before Completion; and (ii) under their respective terms, claims can still be made or pursued after Completion for matters or events occurring or existing prior to Completion. The only claims that will be made or pursued by or on behalf of the Buyer's Group

under those policies (together the **Permitted Claims**) will be: (x) those that have already been notified to the relevant insurer(s) before Completion and are pending or outstanding at Completion; and (y) any additional claims that are notified to the Seller on or before the third anniversary of Completion in relation to any event or act that occurred or existing prior to Completion.

18.5 The Seller shall procure that each member of the Seller's Group shall take such commercially reasonable steps as the Buyer reasonably requires (in each case without incurring further cost or liability on the part of the Seller's Group (but excluding any subsequent increase in premia and/or costs under the Seller's Group Insurance Policies arising as a result of making and/or pursuing any Permitted Claim in accordance with this Agreement)) to make and/or pursue any Permitted Claim (including giving notice of the claim to the insurer at the request of the Buyer) or to assist any Target Company or any member of the Buyer's Group in making the claim, and shall pay to the Buyer (on behalf of the relevant Target Companies) any proceeds actually received within ten Business Days of their receipt (less any Tax in respect of such proceeds). To the extent the Buyer considers it has a Permitted Claim, all such Permitted Claims shall be made through the Seller, and the Buyer shall not engage directly with the insurer. The Buyer shall ensure that any information provided to the Seller for the purposes of making a Permitted Claim is accurate and shall indemnify and hold harmless the Buyer in the event of any losses arising as a result of the inaccuracy of such information or otherwise arising from making an improper claim on the Buyer's behalf in accordance with the Buyer's requests.

18.6 The Seller undertakes (in the event that a claim is made against it in connection with the Proposed Transaction) not to make a claim against any Target Company or any person who was at any time prior to Completion an employee, consultant, officer or director of any Target Company (a Covered Person) on whom the Seller may have relied in negotiating this Agreement, except in the case of fraud, bad faith or deliberate concealment by such Target Company or Covered Person.

19 Post-Completion matters

Further assurance

19.1 Each Party shall at its own cost (and shall use all reasonable endeavours to procure that any third parties shall) promptly execute and deliver such other documents in a form reasonably satisfactory to the Party concerned, and take such other actions, as required by law or as may be necessary or reasonably required by the other Party to give the other Party the full benefit of this Agreement, the Hive Down Agreements and the other Transaction Documents.

"Wrong pockets"

- 19.2 If, following Completion, in error the Seller or any member of the Seller's Group (a) owns, possesses, receives or has the benefit of any asset; (b) is party to or has the benefit of any contract or part of a contract; or (c) has a liability, in each case which relates exclusively to the Business, the Seller shall, at its sole expense, procure that such asset is transferred or the relevant contract is assigned, novated or otherwise transferred to, or such liability is assumed by, the Buyer (or such member of the Buyer's Group as the Buyer shall nominate), in each case together with any benefit or sum, net of any liability to Tax (or any liability that would arise but for the availability of any Relief), accruing to any member of the Seller's Group as a result of holding such asset, contract or liability since Completion that would have accrued for the benefit of the Buyer but for such error for no additional consideration as soon as practicable after becoming aware of such error.
- 19.3 If, following Completion, in error the Buyer or any member of the Buyer's Group (a) owns, possesses, receives or has the benefit of any asset; (b) is party to or has the benefit of any contract; or (c) has a liability, in each case which relates exclusively to the Retained Business, the Buyer shall, at the Seller's sole expense, procure that such asset is transferred or the relevant contract is assigned, novated or otherwise transferred to, or such liability is assumed by, the Seller (or such member of the Seller's Group as the Seller shall nominate), in each case together with any benefit or sum, net of any liability to Tax (or any liability that would arise but for the availability of any Relief), accruing to any member of the Seller's Group as a result of holding such asset, contract or liability since Completion, that would have accrued for the benefit of the Seller but for such error for no consideration as soon as practical after becoming aware of such error.

19.4 Pending:

- (a) any transfer, novation or assignment of an asset or contract pursuant to clause 19.2 or 19.3, the relevant asset or contract shall be held on trust for the person entitled thereto until the completion of such transfer or assignment, and such person shall pay on demand to the other Party an amount equal to all Losses suffered or incurred by such other Party and/or its Related Undertakings in respect of holding such asset or contract until the completion of such transfer, novation or assignment; and
- (b) any assumption of a liability pursuant to clause 19.2 or 19.3, the Party properly liable therefor shall pay on demand to the other Party an amount equal to all Losses suffered or incurred by such other Party and/or its Related Undertakings in respect of such liability until the completion of such assumption.
- 19.5 Subject to the terms of the Transitional Services Agreement, if, at any time after Completion:

- (a) the Buyer (or any member of the Buyer's Group) receives any monies in respect of Business Receivables, the Buyer shall, as soon as practicable, pay to the Seller (or as the Seller may direct) the amount of the monies received in respect of such Business Receivable; and
- (b) the Seller (or any member of the Seller's Group) receives any monies in respect of goods and/or services supplied by the Business following Completion, the Seller shall, as soon as practicable, pay to the Buyer (or as the Buyer may direct) the amount of the monies received.
- 19.6 Subject to the terms of the Transitional Services Agreement, if, at any time after Completion:
 - (a) the Seller receives any invoice or other demand or request for payment in respect of goods and/or services supplied to the Business following Completion, the Seller shall promptly notify the Buyer and the Parties shall use their respective reasonable endeavours to procure that such invoice, demand or request is cancelled by way of an appropriate credit note and re-issued by the relevant counterparty in the name of the Buyer (or as the Buyer may direct) provided that if such invoice, demand or request is not so reissued by the time it falls due, the Buyer shall, as soon as practicable, pay to the Seller (or to the relevant counterparty or otherwise as the Seller may direct) the amount of such invoice, demand or request (including any VAT payable in respect thereof); and
 - (b) the Buyer receives any invoice or other demand or request for payment of any Business Payable, the Buyer shall promptly notify the Seller and the Parties shall use their respective reasonable endeavours to procure that such invoice, demand or request is cancelled by way of an appropriate credit note and re-issued by the relevant counterparty in the name of the Seller (or as the Seller may direct) provided that if such invoice, demand or request is not so reissued, the Seller shall, as soon as practicable, pay to the Buyer (or to the relevant counterparty or otherwise as the Buyer may direct) the amount of such Business Payable (including any VAT payable in respect thereof).

Books and records

19.7 The Buyer shall, and shall procure that the Target Companies shall, for a period of seven years after Completion, retain and permit the Seller and other members of the Seller's Group (and their duly authorised Representatives) to have reasonable access during normal business hours to (with the right to take copies of, including electronic copies of, at the Seller's expense), the books, records and documents of the Target Companies (including any financial and/or tax records, assessments and returns, and related correspondence with any auditor of any Target Company and/or any Tax Authority) to the extent they relate to periods prior to or which include Completion.

19.8 The Seller shall, and shall procure that the relevant members of the Seller's Group shall, for a period of seven years after Completion, retain and permit the Buyer and other members of the Buyer's Group (and their duly authorised Representatives) to have reasonable access during normal business hours to (with the right to take copies of, including electronic copies of, at the Buyer's expense), the books, records and documents of the relevant members of the Seller's Group (including any financial and/or tax records, assessments and returns, and related correspondence with any auditor of any such member of the Seller's Group and/or any Tax Authority) to the extent they relate to a Target Company and/or the Business and to periods prior to or which include Completion.

Debranding

- 19.9 Subject to clause 19.10 below, the Buyer shall, and shall procure that each member of the Buyer's Group shall, within three months following Completion, remove from all business stationary and from all premises occupied by the Buyer's Group any trademarks or business name that consist of or includes the word "Kerry" (the Retained Name).
- 19.10 The Buyer and each member of the Buyer's Group shall:
 - (a) not be required to remove any trademarks or business name that consists of or includes the Retained Name from Stock held at Completion but shall use commercially reasonable endeavours to dispose of such Stock in the ordinary course of trading; and
 - (b) be permitted to use, on an non-exclusive, royalty-free, non-transferable basis, any trade marks of business names that consist of or include the Retained Name on the Target Companies' product labels and packaging for a maximum period of 12 calendar months from Completion in order to give the Target Companies a reasonable grace period to rebrand their product labels and packaging provided that no new orders of packaging including the Retained Name are placed after a period of 3 months from Completion and the Retained Name is used only: (i) on labels and packaging that display the Retained Name at Completion; and (ii) in the same manner and form as the Retained Name is used as at Completion, and provided further that in the event of any actual or suspected material breach of Food Safety Laws in respect of products bearing the Retained Name, the Buyer shall promptly notify the Seller and consult with the Seller about any potential product recall or similar remediation, and the Seller shall be entitled at its option acting reasonably to terminate this licence.

Change of Companies' Names

- 19.11 The Seller shall procure that Henry Denny & Sons (Ireland) Limited shall change its company name to a name which shall not contain the word "Denny" within one month of Completion.
- 19.12 In connection with the change of names of companies in the Seller's Group to be implemented pursuant to clause 19.11, the Seller shall, and shall procure that each member of the Seller's Group shall, within three months following Completion, remove from all business stationary and from all premises occupied by the Seller's Group any trademarks or business names that consist of or include the word "Denny".

Release of outstanding Guarantees

19.13 The Buyer shall:

- use its reasonable endeavours to secure as soon as reasonably practicable after Completion the release of the Seller and any other member of the Seller's Group (without cost to the Seller) from the Guarantees and other contingent liabilities listed in Schedule 2 of the Disclosure Letter, including (if required) offering its own Guarantee or liability on the same terms as and in substitution for the existing Guarantee or other liability of the relevant member of the Seller's Group, provided that the obligation to ensure release of all such Guarantees and other contingent liabilities shall not require the Buyer to provide or offer any guarantees, surety or other security (including a security deposit or bank guarantee) or agree to any amendment or variation to the agreement to which any such Guarantee or other contingent liability relates that would, in the reasonable opinion of the Buyer, result in less favourable terms under such agreement for the Buyer or any Target Company. In addition, the Buyer shall use its reasonable endeavours to ensure that, as soon as reasonably practicable after becoming aware of any other Guarantee or other contingent liabilities in respect of any obligations of any Target Company (but at the earliest at Completion), the Seller and any other member of the Seller's Group is released (without cost to the Seller) from such Guarantee or other contingent liabilities (which for the avoidance of doubt shall not include the Section 357 Guarantees); and
- (b) pay to the Seller on demand an amount equal to all Losses which the Seller's Group may suffer or incur in respect of any valid claim made under or in respect of any such Guarantees, the Section 357 Guarantees or other contingent liabilities which relates to any matter or event occurring after Completion.

19.14 The Seller shall:

 ensure that at Completion each Target Company (without cost to any member of the Buyer's Group) is released from all Guarantees and other contingent liabilities given by such Target Company in respect of obligations of the Seller or any other member of the Seller's Group, including (if required) offering its own Guarantee or liability on the same terms as and in substitution for the existing Guarantee or other liability of the relevant Target Company; and

(b) pay to the Buyer on demand an amount equal to all Losses which the Buyer's Group may suffer or incur in respect of any valid claim made under or in respect of any such Guarantees or other contingent liabilities which relates to any matter or event occurring after Completion.

20 Environmental indemnity

- 20.1 The Parties acknowledge and agree that notwithstanding any other provision of this Agreement, the Buyer's Group shall be entitled to procure the Groundwater Survey required in respect of the Hyde Site on or before the second anniversary of Completion, provided that the Buyer shall:
 - (a) submit to the Seller the proposed scope and methodology for the Groundwater Survey and the name of the proposed environmental consultant to be appointed;
 - take account of all reasonable and prompt requests of the Seller in finalising with the Seller such proposed scope and methodology and the appointment of the environmental consultant;
 - (c) submit the proposed scope and methodology agreed between the Buyer and Seller for review by the Environment Agency (and shall use reasonable endeavours to obtain approval for such scope and methodology from the Environment Agency, it being understood that the Buyer may proceed with the Groundwater Survey without such approval if the Environment Agency's response is unreasonably delayed); and
 - (d) provide to the Seller copies of any interim and final reports from the relevant environmental consultant as soon as reasonably practicable.
- 20.2 The Seller shall indemnify and hold harmless the Buyer, and shall pay promptly on demand to the Buyer an amount equal to:
 - (a) the costs incurred by the Buyer or any member of the Buyer's Group of undertaking any Remedial Works to remove any asbestos containing material identified as "high risk" and for which the recommendation is "remove" in the Asbestos Reports to the extent that it (if any) remains present at the Hyde Property on the date of Completion, and any Losses suffered by the Buyer and/or any member of the Buyer's Group as a result of such presence (including without limitation any reasonable costs or Losses arising from any interruption to

the operation of the Business at the Hyde Site necessary to procure such removal) incurred within 18 months of the Completion Date; and

- (b) any Losses suffered by the Buyer and/or any member of the Buyer's Group as a result of any claim, action, cause of action, suit, proceeding or written notice alleging liability arising out of, based on or resulting from the personal injury, death, or other impairment or risk to the health of any person as a result of their exposure to asbestos or asbestos-containing materials, to the extent that such exposure occurred at the Hyde Property prior to Completion as a result of the failure of the Seller's Group to remove the asbestos containing material identified as "high risk" as described in sub-paragraph (a); and
- (c) the costs incurred by the Buyer or any member of the Buyer's Group of any Remedial Works:
 - (i) in respect of any contamination or pollution:
 - (A) arising out of or in connection with the activities of any member of the Seller's
 Group at the Hyde Property; or
 - (B) any other contamination or pollution of which the Seller is aware as at the Completion Date (provided that "aware" for the purpose of this sub-paragraph shall have the same meaning as described in clause 15.6); and
 - (ii) which are required to bring the Hyde Property (or any part of it) to a standard suitable for its current use as at the Completion Date (unless otherwise expressly required by an Environmental Authority); and
 - (iii) as required or requested by an Environment Authority; and
 - (iv) the requirement for which has arisen or been identified as a result of undertaking the Groundwater Survey,

provided that the Buyer and Seller shall consult with respect to planning any Remedial Works and such Remedial Works are undertaken by and on the recommendation of an appropriately qualified asbestos or environmental consultant (as the case may be) approved by the Seller (such approval not to be unreasonably delayed or withheld), in accordance with clauses 20.13 to 20.19 (inclusive).

20.3 The Parties acknowledge and agree clause 20.2(c) constitutes an "agreement on liabilities" under the Department for Environment, Food and Rural Affairs' statutory guidance on the contaminated land regime set out in Part IIA of the Environment Protection Act 1990, and the Seller shall

- promptly (and the Buyer may) produce a copy of it to any Environmental Authority for the purposes of procuring that any liability is allocated between the Parties in accordance with its provisions.
- 20.4 The provisions of Schedule 6 apply, save that paragraphs 2.2, 2.3 and 3 of Schedule 6 shall not apply to the indemnities given by the clause 20.2 above, and the Parties acknowledge and agree that no provision of Schedule 6 shall operate to limit the Buyer's entitlement to indemnification pursuant to clause 20.2(c) as a result of the Groundwater Survey conducted pursuant to clause 20.1, provided that the Groundwater Survey is conducted in accordance with this clause 20.

Limitations on Asbestos Indemnity Claims

- 20.5 The Seller in respect of any Asbestos Indemnity Claim shall only be liable up to a maximum aggregate of [***] and such amount will be reduced in an equal and proportionate amount to any costs incurred by or on behalf of the Seller in relation to undertaking the removal prior to the Completion Date of any asbestos containing material identified as "high risk" and for which the recommendation is "remove" in the Asbestos Reports.
- 20.6 The Seller shall not be liable in respect of any Asbestos Indemnity Claim unless the Buyer has given written notice to the Seller as soon as reasonably practicable after the Buyer becomes aware of any Asbestos Indemnity Claim and in any event on or before the date falling 18 months after the Completion Date.

Limitations on Groundwater Indemnity Claims

- 20.7 The Seller shall not be liable in respect of any Groundwater Indemnity Claim unless the aggregate amount of all Groundwater Indemnity Claims for which the Seller would be liable but for this clause 20.7 exceeds in aggregate [***] in which case the Seller shall be liable for the excess.
- 20.8 The Seller shall not be liable in respect of any Groundwater Indemnity Claim unless the Buyer has given written notice to the Seller as soon as reasonably practicable after the Buyer becomes aware of any Groundwater Indemnity Claim and in any event on or before the date falling 5 years after the Completion Date.
- 20.9 The Seller shall not be liable in respect of any Groundwater Indemnity Claim which arises due to any change in Environmental Laws or any new Environmental Laws which comes into force after the Completion Date;
- 20.10 Without limiting the generality of clause 20.2(c), the Seller shall not be liable for any Groundwater Indemnity Claim to the extent that such claim arises from or is increased by:

- (a) any development of the Hyde Property other than in the ordinary course of business and in the context of the current use of the Hyde Property;
- (b) any change of use, closure or temporary or permanent cessation of operations at the Hyde Property after the Completion Date;
- (c) any voluntary investigation carried out or Remedial Works carried out by or on behalf of the Buyer after the Completion Date (and for the avoidance of doubt the Parties acknowledge and agree that the Groundwater Survey is not a voluntary investigation);
- (d) any Remedial Works commenced without the prior approval of such Remedial Works by the Seller (not to be unreasonably withheld or delayed).
- 20.11 Without limiting the generality of clause 20.2(c), the Seller shall not be liable for any Groundwater Indemnity Claim under clause 20.2(c) to the extent that such claim arises from or is increased by:
 - (a) the voluntary provision of information by the Buyer or the Buyer's Group, or any of their officers, employees, agents or contractors to an Environmental Authority or other third party unless such disclosure is:
 - (i) required by Environmental Laws, requested by any Environmental Authority;
 - (ii) necessary to respond to any emergency where there is a risk of significant damage to the Environment (including human health); or
 - (iii) made with the prior written consent of the Seller;
 - (b) any contamination or pollution in, on at or under the Hyde Property which is knowingly caused, contributed to, exacerbated or permitted due to any actions or negligent omissions by the Buyer or any member of the Buyer's Group, its contractors or agents.

Conduct

- 20.12 The Buyer shall have conduct of any Remedial Works relating to any Groundwater Indemnity Claim in accordance with the terms of this Agreement.
- 20.13 In conducting any Remedial Works the subject of a Groundwater Indemnity Claim, the Buyer shall ensure (unless otherwise agreed by the Parties acting reasonably) that:
 - (a) a detailed description of the proposed Remedial Works is provided to the Seller for approval (not to be unreasonably withheld or delayed) prior to undertaking any such Remedial Works, or any significant change to the scope of such Remedial Works;

- (b) monthly detailed reports (including invoices and other supporting financial information) are provided to the Seller regarding the progress of any Remedial Works;
- (c) copies of any relevant notices, directions, reports or material correspondence and documents in relation to such Remedial Works are provided to the Seller, provided that the Buyer shall not be required to provide to the Seller any document the disclosure of which could prejudice any legal privilege;
- the Buyer shall have due regard to any reasonable and timely requirements and requests by the Seller;
- the Seller is allowed to attend any material site visits, meetings with the Environmental Authority and/or meetings with the environmental consultants or contractors undertaking the Remedial Works;
- (f) the Buyer shall take reasonable steps to mitigate any costs at all relevant times.
- 20.14 The Seller shall only be liable for a Groundwater Indemnity Claim to the extent that any Remedial Works the subject of such Groundwater Indemnity Claim are carried out:
 - (a) by suitably qualified and experienced environmental consultants and other contractors holding suitable insurance policies and who will, in each case, provide appropriate warranties to the Seller;
 - (b) in compliance with all Environmental Laws;
 - competently, diligently and in accordance with standard industry practice and techniques;
 and
 - (d) with the prior approval (including the costs for such Remedial Works) of the Seller, not to be unreasonably withheld or delayed, and provided that such prior approval shall not be required if (and only to the extent that) such Remedial Works are the minimum necessary to respond to an emergency.
- 20.15 If the Seller and the Buyer are unable to agree any dispute under clause 20.2 of this Agreement (Unresolved Issue) within 20 Business Days after receipt by the Buyer or the Seller of notice from the other that the relevant issue is in dispute and that the party giving such notice requires expert determination (unless it is resolved to the satisfaction of that party during that period or the Environmental Authority provides specific directions or guidance to resolve the issue), then the Buyer or the Seller shall have the right thereafter to require that the Unresolved Issue shall be referred for binding determination to an independent environmental consultant (Environmental

Expert). If Buyer and the Seller fail to appoint an Environmental Expert together within 10 Business Days of specifying a requirement for a referral, then the Environmental Expert will be appointed on the application of either party by the President for the time being of the Institute of Environmental Management and Assessment in the United Kingdom.

- 20.16 The costs and expenses of the Environmental Expert shall be borne by the parties as determined by the Environmental Expert (or, in the absence of any such determination, by the Buyer and the Seller equally).
- 20.17 The Environmental Expert shall be an independent consultant and shall be required to have at least 10 years' experience in relation to matters of the same general description as the relevant Unresolved Issue.
- 20.18 The Environmental Expert shall be instructed to reach his determination and publish it in a private and confidential report addressed to the Buyer and the Seller within 20 Business Days of his appointment on the basis of the information provided by the Buyer and the Seller, unless the Environmental Expert bona fide believes that such information is so incomplete, unrepresentative, unreliable or inconclusive that it would be unreasonable for a suitably qualified expert to reach such a determination of the Unresolved Issue on that basis, in which case the Environmental Expert shall require such further information from the parties as is appropriate to make its determination.
- 20.19 The determination of the Environmental Expert shall be final and binding on the parties and shall be conclusive in any proceedings between the parties, save in the case of manifest error.

21 Property Undertaking

21.1 The Seller undertakes to take such actions as the Buyer may reasonably request in the period between the date of this Agreement and Completion and for a period of up to 12 months following Completion, the Seller undertakes to assist the Buyer as it may reasonably request, to help to preserve the current land usage by the relevant Target Company of the lands at Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan currently occupied by a Target Company shown hatched green and the lands shaded pink on Map 2 annexed to the Declaration of Identity of Peader Stack in respect of the Carrickmacross property (the **Occupied Lands**) and where possible to remediate or resolve the title to the Occupied Lands in order to obtain good and marketable freehold title to the Occupied Lands, to the Buyer's reasonable satisfaction, including by (i) assisting in discussions with the registered owners of the Occupied Lands to acquire the Occupied Lands if requested by the Buyer; (ii) making enquiries with title insurers in relation to the availability of title insurance to protect the Buyer and the relevant Target Company against challenge by the registered owners of the Occupied Lands and effecting title

insurance if available (iii) obtaining valuations in respect of the Occupied Lands; (iv) defending and challenging any claim by any third party to title to the Occupied Lands and providing all assistance reasonably required by the Purchaser to endeavour to register the relevant Target Company occupying the Occupied Lands with possessory title free from encumbrances in the Land Registry, provided always that such actions shall be at the Buyer's exclusive cost.

- 21.2 The Seller undertakes to not knowingly do or permit to be done anything which would result in the registered owners of the Occupied Lands or any third party becoming aware that there is any question as to the relevant Target Companies title to the Occupied Lands unless requested to make representations to the registered owners of the Occupied Lands by notice in writing from the Buyer.
- 21.3 The Seller, and any relevant member of the Seller's Group, undertakes to engage in good faith discussions during the period between the date of this Agreement and Completion in respect of the entering into of licences, effective on and from Completion, in a customary form for the occupation by the Buyer or relevant member of the Buyer's Group of the Burton Property and the Portadown Property (together, the Licensed Properties). Such licences shall include the following terms:
 - (a) in relation to the Burton Property:
 - (i) an initial term of six (6) months from Completion;
 - (ii) a right for the Buyer or relevant member of the Buyer's Group to extend the initial six
 (6) month term for a further three (3) month period by providing at least two (2) months' notice. The Buyer or relevant member of the Buyer's Group may extend the licence twice;
 - (iii) the licence shall reserve a fee payable by the Buyer or relevant member of the Buyer's Group of [***] (exclusive of VAT) per calendar month in respect of the initial six (6) month term and an additional [***] (exclusive of VAT) per calendar month in respect of the additional two three (3) month extensions (if exercised by the Buyer or relevant member of the Buyer's Group); and
 - (iv) a right for the Buyer or relevant member of the Buyer's Group to terminate the licence by providing at least one (1) months' written notice (at any time) to the Seller or relevant member of the Seller's Group;
 - (b) in relation to the Portadown Property:
 - (i) an term of one (1) month from Completion;

- (ii) the licence shall reserve a fee payable by the Buyer or relevant member of the Buyer's Group of [***] (exclusive of VAT) per calendar month in respect of the one (1) month term; and
- (c) in relation to all Licensed Properties:
 - an obligation on the Buyer or relevant member of the Buyer's Group to vacate the premises if the landlord has notified the Seller or relevant member of the Seller's Group of its intention to take forfeiture action in respect of the licence; and
 - (ii) an obligation on the Seller or relevant member of the Seller's Group to use reasonable endeavors to assist the Buyer or relevant member of the Buyer's Group in re-locating to an alternative suitable premises if the lease is forfeited, which shall include the Seller and the Buyer entering into good faith discussions about a suitable alternative, which could include other premises of the Buyer or other members of the Buyer's Group or the premises of any third party. To the extent there are any incremental costs as a result of the relocation, the Seller and the Buyer shall share such costs on a 50/50 basis; and
 - (iii) the Buyer or relevant member of the Buyer's Group shall not:
 - (A) make any structural alterations or material additions whatsoever to the Licensed Properties, and shall ensure that on the expiry date of the relevant licence the Licensed Properties are left in a clean and tidy condition, with all of the Buyer or relevant member of the Buyer's Group's furniture, equipment and goods removed from the Licensed Properties, making good any damage thereby caused, but, for the avoidance of doubt, the Buyer or relevant member of the Buyer's Group shall not be obliged to put the Licensed Properties in a condition that is better than at the date of commencement of the licence;
 - (B) use the Licensed Properties other than for the permitted use; and
 - (C) use the Licensed Properties in a manner which would cause the Seller or relevant member of the Seller's Group to breach any term of the relevant lease.

22 Entire Agreement

22.1 In this clause 22, the Relevant Parties shall mean the Seller, the Buyer, each Designated Seller and each Designated Buyer, and each of them shall be a Relevant Party.

22.2 It is agreed that:

- (a) the Transaction Documents and Hive Down Agreements constitute the entire agreement between the Relevant Parties and supersede any prior agreement, understanding, undertaking or arrangement between the Relevant Parties relating to the subject matter of the Transaction Documents and Hive Down Agreements;
- (b) in entering into the Transaction Documents, the Relevant Parties do not rely on any statement, representation, assurance or warranty of any person (whether a party to the Transaction Documents, the Hive Down Agreements or not and whether made in writing or not) other than as expressly set out in the Transaction Documents or the Hive Down Agreements:
- (c) except as otherwise provided in this Agreement, no Relevant Party may rescind or terminate this Agreement for breach of contract or for negligent or innocent misrepresentation or otherwise; and
- (d) nothing in this clause, and no other limitation in this Agreement, shall exclude or limit any liability for fraud.

23 Effect of Completion

All provisions of this Agreement shall, so far as they are capable of being performed or observed, continue in full force and effect notwithstanding Completion except in respect of those matters then already performed and Completion shall not constitute a waiver of any of the Buyer's rights in relation to this Agreement.

24 Remedies and waivers

The rights and remedies of each Party are, except where expressly stated to the contrary, without prejudice to any other rights and remedies available to it. No neglect, delay or indulgence by any Party in enforcing any provision of this Agreement shall be construed as a waiver and no single or partial exercise of any rights or remedy of any Party under this Agreement will affect or restrict the further exercise or enforcement of any such right or remedy.

25 Release

The liability of any Party to this Agreement may, in whole or in part, be released, compounded or compromised and if any Party shall give time or indulgence to the person under such liability, this will in no way prejudice or affect that Party's rights against any other person under the same or similar liability.

26 Severance

Each provision of this Agreement is severable and distinct from the others and, if any provision is, or at any time becomes, to any extent or in any circumstances invalid, illegal or unenforceable for any reason, that provision shall to that extent be deemed not to form part of this Agreement but the validity, legality and enforceability of the remaining provisions of this Agreement shall not be affected or impaired, it being the Parties' intention that every provision of this Agreement shall be and remain valid and enforceable to the fullest extent permitted by law.

27 Set off

Except as otherwise provided in the Transaction Documents, no Party shall be entitled to set off, deduct or retain from any sum due by it to another Party any amount in respect of a claim under this Agreement or any of the other Transaction Documents.

28 Withholdings and gross-up

- 28.1 All sums payable under this Agreement shall be paid free and clear of all deductions or withholdings whatsoever, save only as may be required by law, regulation or regulatory requirement.
- 28.2 If, at any time, any applicable law, regulation or regulatory requirement requires the Buyer or the Seller (as applicable) make any deduction or withholding from any sums payable in respect of a Buyer Obligation or a Seller Obligation, the amount so due shall be increased by such amount as is necessary to ensure that, after the making of such deduction or withholding, the recipient of that payment receives, on the due date for such payment, a net sum equal to the sum which it would have received had no such deduction or withholding been required to be made.
- 28.3 If any Party is required by law to make any deduction or withholding as referred to in clause 28.2, that Party shall:
 - (a) make such deduction or withholding; and
 - (b) pay the full amount deducted or withheld to the relevant Tax Authority in accordance with applicable law, regulation or regulatory requirement.
- 28.4 If any amount paid or due to the Seller in respect of a Buyer Obligation or to the Buyer in respect of a Seller Obligation (in this clause 28.4, the recipient of any such amount being the **recipient**) is subject to Tax, or would (but for the availability of any Relief) be subject to Tax, in the hands of the recipient, then the amount so paid or due (in this clause 28.4, the **net amount**) shall be increased to an amount (in this clause 28.4, the **grossed-up payment**) which (after subtraction

of the amount of any Tax which the grossed-up payment is subject to, or would, but for the availability of any Relief, be subject to) shall equal the net amount, provided that if any amount is initially paid on the basis that the amount paid or due to the recipient is not subject to Tax in the hands of the recipient or *vice versa* and it is subsequently determined that it is or *vice versa*, an additional amount shall be paid to or by the recipient (as the case may be) as places the recipient in the same after-tax position as it would have been in if the amount paid or due to the recipient had not been taxable in the hands of the recipient.

28.5

- (a) either the Buyer or a Designated Buyer is resident for tax purposes in any jurisdictions other than the UK or ROI;
- (b) either the Seller or a Designated Seller is resident for tax purposes in any jurisdiction other than the UK or ROI.

then the liability of the Seller (where sub-clause (a) applies) or the Buyer (where sub-clause (b) applies) under clauses 28.2 and 28.4 shall be no greater than it would have been if the relevant sub-clause had not applied.

28.6 If there is an assignment or transfer of all or any of the Buyer's rights (or those of a Designated Buyer) under this Agreement in accordance with clause 35.2, then the Buyer shall be entitled to receive a payment under clauses 28.2 and 28.4 only to the same extent as the Buyer would have been entitled if the assignment or transfer had not occurred.

29 Payments and interest

- 29.1 Unless otherwise expressly stated (or otherwise agreed in writing in the case of a given payment), each payment under this Agreement shall be made on or before the date the payment is due in immediately available funds by electronic funds transfer for value on that date to:
 - (a) in the case of a payment to the Seller (or a Designated Seller or any other member of the Seller's Group), the Seller's Bank Account, and (where applicable) the Seller agrees to pay each Designated Seller or other member of the Seller's Group that part of each payment to which it is entitled; and
 - (b) in the case of a payment to the Buyer (or a Designated Buyer or any other member of the Buyer's Group), the Buyer's Bank Account, and (where applicable) the Buyer agrees to pay each Designated Buyer or other member of the Buyer's Group that part of each payment to which it is entitled.

and receipt of the amount due into the Seller's Bank Account or the Buyer's Bank Account (as the case may be) shall constitute an effective discharge of the relevant payment obligation.

29.2 If any sum due for payment in accordance with this Agreement is not paid on the due date for payment, the person in default shall pay interest thereon at the Interest Rate (being such rate per annum which is equal to 2 per cent over the base rate from time to time of Barclays Bank plc) as determined on the due date for payment which shall accrue from day to day and shall be calculated on the basis of a year of 365 days from but excluding the due date to and including the date of payment.

30 Alterations

No amendment to this Agreement will be effective unless it is made in writing and signed by or on behalf of the Parties.

31 Counterparts

- 31.1 This Agreement may be executed in any number of counterparts. Each counterpart, when duly exchanged or delivered, is an original, but the counterparts together are one and the same agreement.
- 31.2 Any counterpart may take the form of an electronic copy of this Agreement and that counterpart:
 - (a) will be treated as an original counterpart;
 - (b) is sufficient evidence of the execution of the original; and
 - (c) may be produced in evidence for all purposes in place of the original.

32 Costs, Transfer Taxes and sales taxes

- 32.1 Each of the Parties shall be responsible for its respective legal and other costs incurred in relation to the negotiation, preparation and completion of this Agreement, the Hive Down Agreements and all other Transaction Documents.
- 32.2 All Transfer Taxes in respect of the sale and purchase of the Sale Shares shall be for the account of, and be paid for by, the Buyer.
- 32.3 All sums payable under or pursuant to this Agreement shall be exclusive of any VAT and, if any party (for the purposes of this clause 32.3 only, the **Supplier**) makes a supply to any other party (for the purposes of this clause 32.3 only, the **Recipient**) for VAT purposes and the Supplier or

an affiliate of the Supplier is required to account for VAT in respect of that supply, the Recipient shall, subject to the receipt of a Valid VAT Invoice, pay to the Supplier an amount equal to such VAT in addition to any other consideration for that supply. Such payment shall be made at the same time as any such consideration is payable.

33 Agreement binding

This Agreement shall be binding on and shall enure for the benefit of the successors in title of each Party.

34 Agency structure

Each of the Seller and the Buyer agrees that it is entering into this Agreement on its own behalf as principal and as agent on behalf of the Designated Sellers or (as the case may be) the Designated Buyers on the basis set out in Schedule 10 (*Agency provisions*).

35 Assignment

- 35.1 Save as provided in clause 35.2, neither the Seller nor the Buyer shall be entitled to assign the benefit of any rights under this Agreement. Any purported assignment in contravention of this clause 34 shall be void.
- 35.2 Subject to any assignment not increasing any liability or creating a liability which would not otherwise have existed for the non-assigning Party, the benefit of rights under this Agreement may be assigned:
 - (a) by the Seller (for itself and/or as agent for each Designated Seller) to another member of the Seller's Group (the Seller's Assignee), provided that, if the Seller's Assignee ceases to be a member of the Seller's Group, the Seller shall procure that the Seller's Assignee shall assign its rights to a member of the Seller's Group and, until such assignment becomes effective, such rights shall cease to be enforceable; and
 - (b) by the Buyer (for itself and/or as agent for each Designated Buyer) to another member of the Buyer's Group (the Buyer's Assignee), provided that, if the Buyer's Assignee ceases to be a member of the Buyer's Group, the Buyer shall procure that the Buyer's Assignee shall assign its rights to a member of the Buyer's Group and, until such assignment becomes effective, such rights shall cease to be enforceable.
- 35.3 The parties acknowledge that this Agreement is a contract entered into for the purposes of, or in connection with, the acquisition, disposal or transfer of an ownership interest in a firm (as defined

in s 1173(1) of the Companies Act 2006). Regulation 2 of The Business Contract Terms (Assignment of Receivables) Regulations 2018 does not apply to any term of this Agreement.

36 Rights of third parties

- 36.1 Save as provided in clause 36.2, a person who is not a Party to this Agreement has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce any of its terms.
- 36.2 The Parties agree that certain provisions of this Agreement confer a benefit on their respective Related Undertakings, and that such provisions are intended to benefit, and be enforceable by, such Related Undertakings or Covered Persons in their own right under the Contracts (Rights of Third Parties) Act 1999, provided that the rights of any Covered Persons under this 36.2 shall only be used as a defence to any proceedings and not to initiate proceedings. Notwithstanding the foregoing, under no circumstances shall any consent be required from any such Related Undertaking or Covered Persons for the termination, rescission, amendment or variation of this Agreement, whether or not such termination, rescission, amendment or variation affects or extinguishes any such benefit or right.

37 Announcements and confidentiality

- 37.1 Subject to clause 37.2, no announcement, public filing, circular or communication (each an Announcement) concerning the existence or content of this Agreement, the Hive Down Agreements or any other Transaction Document shall be made by a Party (or any of its respective Related Undertakings) without the prior written approval of the Seller (in the case of the Buyer or a Designated Buyer) and the Buyer (in the case of the Seller or a Designated Seller) (such approval not to be unreasonably withheld or delayed).
- 37.2 Clause 37.1 does not apply to:
 - (a) any Announcement to the extent that it is required to be made by the rules of any stock exchange or any governmental, regulatory or supervisory body or court of competent jurisdiction (Relevant Authority) to which the Party making the Announcement is subject, whether or not any of the same has the force of law, provided that any Announcement shall, so far as is practicable, be made after consultation with the other Party and after taking into account its reasonable requirements regarding the content, timing and manner of despatch of the Announcement in question; or
 - (b) any Announcement made to the European Employee Forum, any recognised trade union or other appropriate representatives of the Employees in accordance with regulatory or legal requirements or the terms of the relevant agreements and/or understandings with

those trade union bodies or to any employees of the Seller's Group pursuant to the Seller's employee communication strategy.

- 37.3 Subject to clause 37.4, each Party shall treat as strictly confidential, and not disclose in any way to anyone whomsoever, all information received or obtained as a result of entering into or performing any Hive Down Agreement or any Transaction Document (and, in the case of the Seller, as a result of its ownership of the Sale Shares prior to Completion) which relates to:
 - the existence, subject matter and provisions of this Agreement, the Hive Down Agreements or any other Transaction Document;
 - (b) the negotiations relating to this Agreement, the Hive Down Agreements and the other Transaction Documents;
 - (c) the other Party and/or its Related Undertakings; and/or
 - (d) in the case of the Seller with effect only on and after Completion (and subject always to clause 17.1(c)), the Sale Shares.
- 37.4 A Party may disclose information which would otherwise be confidential to the extent:
 - (a) required by the law of any relevant jurisdiction;
 - (b) required by any Relevant Authority to which the Party making the disclosure is subject, whether or not such requirement has the force of law;
 - (c) disclosed to a Tax Authority where such disclosure is required for the reasonable management of that Party's Tax affairs;
 - (d) required to vest the full benefit of this Agreement in any Party;
 - (e) disclosure is made to its Related Undertakings and/or its Representatives, provided that any such Related Undertaking or Representative is first informed of the confidential nature of the information and such Related Undertaking or Representative acts in accordance with the provisions of clause 37.3 as if it were a party hereto and that Party shall be responsible for any breach by such Related Undertakings and/or Representatives of the provisions of clause 37.3;
 - in the case of the Seller only, disclosure is made as part of an information and consultation process with appropriate representatives of the Seller's employees;

- (g) the information has come into the public domain through no fault of that Party or through no breach by that Party of this Agreement; or
- (h) the other Party has given its prior written approval to the disclosure,

provided that, in the case of sub-clauses (a) and (b), any disclosure shall, so far as is practicable, be made only after consultation with the other Party.

38 Notices

- A notice or other communication given under or in connection with this Agreement must be in writing, in the English language, and sent by a Permitted Method to the Notified Address.
- 38.2 The **Permitted Method** means any of the methods set out in column (1) below. A notice given by the Permitted Method will be deemed to be given and received on the date set out in column (2) below.

(1) Permitted Method	(2) Date on which notice deemed given and received
Personal delivery	When left at the Notified Address if left before 5pm on a Business Day, and otherwise on the next Business Day
E-mail	When sent if sent before 5pm on a Business Day, and otherwise on the next Business Day (unless the sender receives an automated notification of non-delivery or rejection by the recipient's e-mail server, in which case the notice shall be deemed not to have been given or received)

38.3 The Notified Address of each of the Parties is as set out below:

Name of Party	Address	E-mail address	Marked for the attention of:
Buyer	1770 Promontory Circle, Greeley, Colorado 80634-9038, United States of America		Matthew Galvanoni (CFO)

Name of Party	Address	E-mail address	Marked for the attention of:
	Copy (which shall not	Copy (which shall	Copy (which shall not
	constitute notice) to:	not constitute	constitute notice) to:
	Freshfields Bruckhaus	notice) to:	Graham Watson
	Deringer LLP		1000 to 0.000 to 0.00
	100 Bishopsgate		
	London, EC2P 2SR		
	United Kingdom		
UK Designated	2nd Floor Building, 1		Matthew Galvanoni
Buyer	Imperial Place, Maxwell		(Director)
100000000000000000000000000000000000000	Road, Borehamwood,		- 1 1 1 1 1 1 1 1.
	Hertfordshire WD6 1JN,		
	United Kingdom	Copy (which shall	
		not constitute	
	Conv (which shall not	notice) to:	Copy (which shall not
	Copy (which shall not constitute notice) to:		constitute notice) to:
	100 NOSCOSO - 100 NOS		Graham Watson
	Freshfields Bruckhaus		Granam Watson
	Deringer LLP		
	100 Bishopsgate		
	London, EC2P 2SR		
	United Kingdom		
Irish Designated	29 Earlsfort Terrace,		Fabio Sandri
Buyer	Dublin 2, D02 AY28,		(Director)
	Ireland) y
		Copy (which shall	_
	Copy (which shall not	not constitute	Copy (which shall not
	constitute notice) to:	notice) to:	constitute notice) to:
	Freshfields Bruckhaus		Graham Watson
	Deringer LLP		
	100 Bishopsgate		

Name of Party	Address	E-mail address	Marked for the attention of:
	London, EC2P 2SR United Kingdom		
Seller	Prince's Street, Tralee, Co. Kerry, V92 EH11, Ireland	Copy (which shall not constitute notice) to:	Ronan Deasy Allan O'Connor (Legal)
	Copy (which shall not constitute notice) to: Norton Rose Fulbright LLP 3 More London Riverside, London, SE1 2AQ, United Kingdom		Copy (which shall not constitute notice) to: Jon Perry

or such other Notified Address as any Party may, by notice to the others, substitute for its Notified Address set out above.

39 Governing law and jurisdiction

- 39.1 This Agreement and any non-contractual obligations connected with it shall be governed by English law.
- 39.2 The Parties irrevocably agree that all disputes arising under or in connection with this Agreement, or in connection with the negotiation, existence, legal validity, enforceability or termination of this Agreement, regardless of whether the same shall be regarded as contractual claims or not, shall be exclusively governed by and determined only in accordance with English law.
- 39.3 The Parties irrevocably agree that the courts of England and Wales are to have exclusive jurisdiction, and that no other court is to have jurisdiction to:
 - (a) determine any claim, dispute or difference arising under or in connection with this Agreement, any non-contractual obligations connected with it, or in connection with the

negotiation, existence, legal validity, enforceability or termination of this Agreement, whether the alleged liability shall arise under the law of England and Wales or under the law of some other country and regardless of whether a particular cause of action may successfully be brought in the English courts (**Proceedings**); and

- (b) grant interim remedies, or other provisional or protective relief.
- 39.4 The Parties submit to the exclusive jurisdiction of the courts of England and Wales and accordingly any Proceedings may be brought against the Parties or any of their respective assets in such courts.

40 Service of process

- 40.1 The Seller hereby irrevocably authorises and appoints Kerry Foods Limited (for the attention of Kevin Pring) of Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, TW20 8HY (the Seller's Agent) to accept on its and each Designated Seller's behalf service of all legal process arising out of or in connection with any proceedings before the courts of England and Wales in connection with this Agreement. The Seller agrees that (for itself and each Designated Seller):
 - failure by the Seller's Agent to notify it of the process will not invalidate the proceedings concerned; and
 - (b) if this appointment is terminated for any reason whatsoever, it will appoint a replacement agent having an office or place of business in England or Wales and will notify the Buyer of this appointment.
- 40.2 The Buyer hereby irrevocably authorises and appoints the UK Designated Buyer (the **Buyer's Agent**) to accept on its and each Designated Buyer's behalf service of all legal process arising out of or in connection with any proceedings before the courts of England and Wales in connection with this Agreement. The Buyer agrees that (for itself and each Designated Buyer):
 - failure by the Buyer's Agent to notify it of the process will not invalidate the proceedings concerned; and
 - (b) if this appointment is terminated for any reason whatsoever, it will appoint a replacement agent having an office or place of business in England or Wales and will notify the Seller of this appointment.

IN WITNESS whereof, this Agreement has been executed the day and year first above written.

Schedule 1

Schedule 2 Target Companies

Part A – Existing Companies

Company Name	Rollover Limited
Date and place of incorporation	1 May 1991 (England and Wales)
Registration number	02606878
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY
Issued share capital	10,548,999 ordinary shares of £0.01 each
Shareholder	Rollover Group Limited (100%)
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon
Auditors	PricewaterhouseCoopers
Accounting reference date	31 December

Company Name	Spurway Foods Limited	
Date and place of incorporation	12 August 1992 (England and Wales)	
Registration number	02739017	
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY	
Issued share capital	150,000 ordinary shares of £1.00 each	
Shareholder	Kerry Holdings (U.K.) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Marguerite Larkin	
Auditors	PricewaterhouseCoopers	
Accounting reference date	31 December	

Company Name	Noon Products Limited
Date and place of incorporation	17 September 1987 (England and Wales)
Registration number	02166664
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY
Issued share capital	1,000,000 ordinary shares of £0.10 each
Shareholder	Noon Group Limited (100%)
Directors	Ronan Deasy, Trevor James Horan, Marguerite Larkin
Auditors	PricewaterhouseCoopers
Accounting reference date	31 December

Company Name	Oakhouse Foods Limited	
Date and place of incorporation	31 May 2001 (England and Wales)	
Registration number	04226390	
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY	
Issued share capital	34 Ordinary A shares of £1.00 each 34 Ordinary B shares of £1.00 each 34 Ordinary C shares of £1.00 each	
Shareholder	Kerry Holdings (U.K.) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors	PricewaterhouseCoopers	
Accounting reference date	31 December	

Company Name	Duffy Meats Limited
Date and place of incorporation	21 January 1982 (Republic of Ireland)
Registration number	87159
Registered office	Prince's Street, Tralee, Co. Kerry, Ireland, V92 EH11
Issued share capital	250,000 ordinary shares of €1.27 each
Shareholder	Kerry Holdings (Ireland) Limited (100%)
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon
Auditors	PricewaterhouseCoopers
Accounting reference date	31 December

Company Name	Rye Valley Foods Limited	
Date and place of incorporation	23 July 1986 (Republic of Ireland)	
Registration number	115669	
Registered office	Prince's Street, Tralee, Co. Kerry, Ireland, V92 EH11	
Issued share capital	146,536 A ordinary shares of €1.27 each 107,520 B ordinary shares of €1.27 each 73,561 C ordinary shares of €1.27 each 100 voting non-participating shares of €1.27 each	
Shareholder	Kerry Holdings (Ireland) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors	PricewaterhouseCoopers	
Accounting reference date	31 December	

Part B - Newly-Trading Companies

Company Name	Northfield Foods Limited	
Date and place of incorporation	22 April 2021 (England and Wales)	
Registration number	13352340	
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY	
Issued share capital	2 ordinary shares of £1.00 each	
Shareholders	Kerry Holdings (U.K.) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors	141	
Accounting reference date	-	

Company Name	Attleborough Foods Limited	
Date and place of incorporation	22 April 2021 (England and Wales)	
Registration number	13352332	
Registered office	Thorpe Lea Manor, Thorpe Lea Road, Egham, Surrey, United Kingdom, TW20 8HY	
Issued share capital	2 ordinary shares of £1.00 each	
Shareholders	Kerry Holdings (U.K.) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors	-	
Accounting reference date	-	

Company Name	Consumer Foods Van Sales Limited	
Date and place of incorporation	ration 21 April 2021 (Northern Ireland)	
Registration number	NI678671	

Registered office	Millburn Road, Coleraine, Co Londonderry, Northern Ireland BT52 1QZ	
Issued share capital	2 ordinary shares of £1.00 each	
Shareholders	Kerry Holdings (U.K.) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors		
Accounting reference date		

Company Name	Consumer Foods Van Sales (Ireland) Limited	
Date and place of incorporation	5 December 2019 (Republic of Ireland)	
Registration number	661948	
Registered office	Prince's Street, Tralee, Co. Kerry, Ireland, V92 EH11	
Issued share capital	54,631 ordinary shares of €1.00 each	
Shareholder	Kerry Holdings (Ireland) Limited (100%)	
Directors	Ronan Deasy, Trevor James Horan, Claire Salmon	
Auditors	PricewaterhouseCoopers	
Accounting reference date	e date 31 December	

Schedule 3 Properties

Part A - Owned Real Estate

Property Address	Title number (if relevant)
Coolattin, Shillelagh, Co. Wicklow, Y14 PY73, Ireland	Folio WW13126F Folio WW16550F
Land to the north east of Station Road, Godley Hill, Hyde, Cheshire, SK14 3BR	GM682146 GM865732
Rookery Farm, Little Ellingham, Attleborough, Norfolk NR17 1JH UK	NK131085
Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan, A81 E027	Folio MN11786F

Part B - Leased Real Estate

Property Address	Title number (if relevant)
Carran Business Park, Enniskillen, Co. Fermanagh, Northern Ireland	Folio FE94980L Folio FE61223L
801 Oxford Avenue Trading Estate, Slough SL1 4LN	BK490112
802 Oxford Avenue Trading Estate, Slough SL1 4LN	BK490114
Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan, A81 E027	Folio MN735L Folio MN752L Folio MN789L
The Noon Unit, Great Western Industrial Park, Windmill Lane, Southall, Middlesex UB2 4NA	AGL113738
Land and buildings on the east side of Dean Way, Great Western Industrial Park, Southall UB2 4SB	AGL65178
Unit 2 Collett Way, Great Western Industrial Estate, Southall UB2 4SE	AGL332635
Unit 3 Auriol Drive, Off Oldfield Lane, Greenford, Middlesex, UB6 0TP	AGL338988
Unit 5 Stoke Mill, Stoke Mill Road, Sharnbrook, Bedfordshire	Not registered at the Land Registry.
Unit 8 Valley Point Industrial Estate, Beddington Farm Road, Croydon, CR0 4WP	SGL710495
Unit 7 Orchard Court, Heron Road, Sowton Industrial Estate, Exeter, EX2 7LL	DN674745
Unit 21 Collingwood Court, Riverside Park, Middlesbrough, TS2 1RP	Not registered at the Land Registry.
Unit 22 Collingwood Court, Riverside Park, Middlesbrough, TS2 1RP	Not registered at the Land Registry.
Unit 23 Burnhouse Industrial Estate, Whitburn, EH4 0LQ	Not registered at the Land Registry.
Unit 102 Coed Aben Road, Wrexham Industrial Estate, Wrexham	Not registered at the Land Registry.
4A and 4B, Phase 1 Cornbrash Park, Bumpers Lane, Chippenham, Wiltshire	Not registered at the Land Registry.

Schedule 4

Seller Warranties

1 Powers and obligations of the Seller

- 1.1 The Seller and each Designated Seller is validly incorporated, in existence and duly registered under the laws of its jurisdiction of incorporation and not liable to tax elsewhere.
- 1.2 The Seller and each Designated Seller has the right, power and authority and has taken all action necessary to execute and deliver, and to exercise its rights and perform its obligations under, this Agreement, each of the Hive Down Agreements and each of the other Transaction Documents executed or to be executed by it at or before Completion, and each such document will, when executed, constitute legal, valid and binding obligations of the Seller and each Designated Seller.
- 1.3 Other than to the extent relevant for the Conditions, no consent, authorisation, licence or approval of any governmental, administrative, judicial or regulatory body, authority or organisation is required to authorise the execution, delivery, validity, enforceability or admissibility in evidence of this Agreement, any of the Hive Down Agreements or any of the other Transaction Documents or the performance by the Seller and each Designated Seller of its obligations under this Agreement, any of the Hive Down Agreements or any of the other Transaction Documents to which the Seller and each Designated Seller is or will be a party.
- 1.4 Each Designated Seller has entered into an agency agreement with the Seller substantially in the form of the Standard Transaction Agency Agreement.

2 Sale Shares

- 2.1 The Sale Shares, details of which are set out opposite "Issued share capital" in respect of each Target Company in Schedule 2 (*Target Companies*), have been validly issued and allotted and constitute the entire issued and allotted share capital of that Target Company and are fully paid up.
- 2.2 Each Designated Seller is the sole legal and beneficial owner of the Sale Shares in the Target Company set out opposite its name in column (2) of Schedule 1 (*Designated Sellers, Designated Buyers and Apportionment of Purchase Price*), and such Designated Seller is entitled to sell and transfer the full legal and beneficial ownership in such Sale Shares to the Designated Buyers on the terms set out in this Agreement.
- 2.3 There is no Encumbrance on, over or affecting the Sale Shares, there is no agreement or commitment to give or create any such Encumbrance and no person has made any claim to be entitled to any right over or affecting the Sale Shares.

3 Hive down

3.1 The Seller's Group has completed the Hive Down in accordance with the Hive Down Agreements.

4 The Target Companies

- 4.1 The information set out in Schedule 2 (Target Companies) is complete and accurate.
- 4.2 Each Target Company is validly incorporated, in existence and is a private company limited by shares formed and registered under the laws of its place of incorporation and not liable to tax elsewhere.
- 4.3 The Seller and each Target Company has the full power to conduct its business as conducted at the date of this Agreement.
- 4.4 No Target Company owns any shares or debentures in the capital of, nor does any Target Company have any beneficial interest in, any other company, nor has any Target Company entered into any legally binding obligations to acquire any shares, debentures or beneficial interest in any other company.
- 4.5 All registers required by law to be kept by each Target Company are in the possession or under the control of the Seller's Group, have been properly kept during the period of ownership by the Seller's Group, and are up to date in all material respects and contain records of the matters which are required by applicable laws to be recorded in them.

5 Compliance with legal requirements

- 5.1 The Seller's Group is conducting, and has at all times during the Relevant Period conducted, the Business, and the business of the Existing Companies, in all material respects in accordance with all applicable laws and regulations of the jurisdictions in which the Business and/or the Existing Companies are engaged in business.
- 5.2 The Seller's Group has obtained all material licences, permissions, consents and other approvals (including the Authorisations) (Approvals) and made all fillings required for the carrying on of the Business, and the business of the Existing Companies in the places and in the manner in which such activities are now carried on. So far as the Seller is aware, the Approvals are in full force and effect, have been complied with during the Relevant Period in all material respects and there are no circumstances existing which are likely to cause any such Approvals to be revoked or not renewed, whether in whole or in part.
- 5.3 No member of the Seller's Group has been notified in writing in the Relevant Period that any investigation, review, enquiry, proceeding or claim in respect of the Business or other activity of

the Existing Companies is being or has been initiated, conducted or is threatened or pending by any governmental, administrative, judicial or regulatory or other body, authority or organisation in relation to any Financial Crime Law, the Modern Slavery Act 2015, Food Laws or any applicable antitrust, merger control or state aid laws, and so far as the Seller is aware there are no circumstances likely to give rise to any such investigation, enquiry, proceeding or claim.

- 5.4 No member of the Seller's Group (or, in the case of Financial Crime Law, and so far as the Seller is aware, any person who is or has been a director, officer or employee of any member of the Seller's Group or who is an Associated Person of the Seller's Group) is or has at any time in the Relevant Period engaged in any activity, practice or conduct or has taken any action or failed to act, in each case in connection with the Business and whether directly or indirectly, which would constitute a material offence under any Food Law or an offence under any Financial Crime Law or the UK Modern Slavery Act 2015, or any applicable antitrust, merger control or state aid laws.
- 5.5 The Seller's Group has in place, and at all times during the last three years has had in place, in relation to the Business policies, systems, controls and procedures designed to prevent it from violating any Financial Crime Laws and has kept accurate and fair records of its activities, including records of training, financial records, books and accounts, in a form and manner appropriate for a business of its size.

6 Accounting information

6.1 The Accounts:

- (a) comply with the requirements of CA 2006 or CA 2014 (as applicable) and the Accounting Standards; and
- (b) give a true and fair view of the state of affairs of the Existing Companies as at the AccountsDate and their profit or loss for the financial year ended on that date.

6.2 The Carve-Out Accounts:

- (a) were properly prepared in a manner described in the sections of the Deloitte VFDD Report headed "Executive Summary – Basis of preparation", "Meat Platform Historical Trading", "Meat Platform Balance Sheet", "Ready Meals Platform Historical Trading", "Ready Meals Platform Balance Sheet" and Appendices A1 and A2; and
- (b) fairly present with reasonable accuracy, and do not materially misstate, the financial and/or trading position of the Business on a pro forma basis as at the date to which they have been prepared having regard to the purpose for which they were prepared and the fact that they are not audited.

- 6.3 Since the Accounts Date and in each case other than in connection with, or as a consequence of, the Hive Down (including the incorporation and set-up of the Newly-Trading Companies) or this Agreement:
 - the Seller's Group has entered into transactions and incurred liabilities in connection with the Business in the ordinary course of day-to-day trading operations and not otherwise;
 - (b) there has been no Material Adverse Change;
 - (c) the Business has not been materially and adversely affected by the loss of any important customer or source of supply;
 - (d) no Encumbrance has been created over the Business or any of the material assets of the Target Companies, other than in the ordinary course of business;
 - (e) no unlawful act of any person has depleted in any material respect the assets of the Target Companies;
 - (f) no Target Company has acquired or disposed of, or agreed to acquire or dispose of, any one or more assets (excluding Stock) in a single transaction or series of connected transactions, where the value of such assets, exceeds £1,000,000;
 - (g) no Target Company has made any changes in terms of employment (excluding for the avoidance of doubt the annual pay review), including pension funding commitments, which taken together could increase the total staff costs of the Target Companies in aggregate by more than £1,000,000 per annum;
 - (h) no Target Company has repaid any borrowing or indebtedness in advance of its stated maturity;
 - (i) no Target Company has repaid, redeemed or otherwise reduced any share or loan capital (or agreed to do so) or issued (or agreed to issue) any share or loan capital or other similar interest (other than inter-company borrowing or indebtedness);
 - (j) no Target Company has declared, paid or made a dividend or other distribution;
 - (k) no Target Company has changed its name or altered its articles of association; and
 - no resolution of any Target Company's shareholders has been passed (except for those representing ordinary business at an annual general meeting).

7 The Assets

- 7.1 The property, rights, contracts and assets owned, leased or otherwise used by the Target Companies, when taken together with the rights and services to be provided under the Transaction Documents, comprise all the property, rights, contracts and assets which, are required to carry on the Business after Completion substantially in the manner as it was carried on at the date of this Agreement and since 1 January 2021.
- 7.2 The property, rights and assets owned, leased or otherwise used by the Target Companies at Completion which are required to carry on the Business after Completion substantially in the manner as it was carried on at the date of this Agreement and since 1 January 2021:
 - (a) are the property of a Target Company free from any hire or hire-purchase agreement, agreement for payment on deferred terms, bill of sale, lien, Encumbrance or other adverse claim and are in the possession of or under the control of or otherwise capable of access by a Target Company; or
 - (b) where they are held by a Target Company under any hire purchase, conditional sale, leasing or rental agreement, particulars of such agreements are contained in the Data Room.

8 Trading

- 8.1 The Business has not during the Relevant Period manufactured, sold or supplied any goods or products which:
 - (a) have been subject to a product recall or other field safety corrective action; or
 - (b) were removed from product lines due to non-compliance with applicable laws and regulations in respect thereof; or
 - (c) have resulted in material third party claims or complaints.
- 8.2 No part of the Business is carried on under the agreement or consent of a third party, and there are no sales agreements, market sharing agreements or other arrangements relating to the Business which restrict the Seller's Group's freedom to operate the whole or any part of the Business in any jurisdiction as it may decide.
- 8.3 The Disclosure Letter contains details of each grant or subsidy or other financial assistance received during the Relevant Period or receivable by the Seller's Group in respect of the Business from any governmental, quasi-governmental or other authority or body (including in the context of Covid-19).

8.4 There are no guarantees or similar financial support obligations binding upon any member of the Seller's Group (other than the Target Companies) in relation to the IED Licences held by any Target Companies in respect of the ROI Sites.

9 Contracts

- 9.1 For the purposes of this paragraph 9, a Material Contract means any contract or agreement entered into by the Seller's Group which relates exclusively or predominantly to the Business (including distributorship, manufacturing, licensing or management agreements) but excluding the Excluded Contracts, and which:
 - is of an unusual or abnormal nature, or outside the ordinary course of trading or involving or which may involve obligations on the Seller's Group calling for special mention; or
 - (b) is incapable of termination by the Seller's Group in accordance with its terms on no more than six months' notice; or
 - gives any party an option to acquire or dispose of any material asset or requires another person to do so; or
 - (d) is likely to result in a material loss on completion or performance; or
 - (e) cannot readily be fulfilled or performed by the Seller's Group on time without undue or unusual expenditure of money, effort or personnel; or
 - establishes any joint venture, consortium, partnership or profit (or loss) sharing agreement or arrangement; or
 - (g) involves the sale or disposal by a Target Company of any company or business where it remains subject to any liability (whether contingent or otherwise) which is not fully provided for in the Accounts or the Carve-Out Accounts; or
 - is dependent on the guarantee or covenant of or security provided by any person other than the Seller's Group and the relevant counterparty,

and if there is any contract or agreement entered into by the Seller's Group which is material to the Business and any part of such contract or agreement satisfies any of limbs (a) to (h) of this paragraph, then such contract shall also be a Material Contract.

9.2 The Data Room or Disclosure Letter contains complete, up-to-date and accurate copies of all Material Contracts

- 9.3 With regard to each of the Material Contracts:
 - each such Material Contract is legally binding on the parties to it and is in full force and effect; and
 - (b) there is no dispute in relation to any such Material Contract nor do circumstances exist which are likely to give rise to such a dispute.
- 9.4 The Data Room or Disclosure Letter contains:
 - (a) lists, of all supplier, third party manufacturing and other trading arrangements with third parties which involve annual income or expenditure in respect of the Business of more than £1,000,000, whether or not such arrangements are exclusive to the Business and whether or not formally documented (the Material Trading Arrangements);
 - (b) a summary of the trading arrangements with the top 5 customers of the Meats Business by revenue during the financial year ended on 31 December 2020, representing at least 65% of revenue in such financial year; and
 - (c) a summary of the trading arrangements with the top 5 customers of the Ready Meals Business by revenue during the financial year ended on 31 December 2020, representing at least 73% of revenue in such financial year.
- 9.5 There are no outstanding bids, tenders, proposals or offers which, if accepted, would constitute a Material Contract or a Material Trading Arrangement between a Target Company and a third party.
- 9.6 There is no material dispute in relation to any such Material Trading Arrangements nor, so far as the Seller is aware, do circumstances exist which are likely to give rise to such a dispute. During the Relevant Period, no customer of or supplier to the Business with annual income or expenditure (as applicable) in respect of the Business of more than £1,000,000 has ceased to deal with the Seller's Group or has indicated in writing an intention to do so, either in whole or in part.
- 9.7 The Data Room contains a list of all material Related Party Agreements.

10 The Properties and other interests in land

- 10.1 The information set out in Schedule 3 (*Properties*) is complete and accurate.
- 10.2 A Target Company is the legal and beneficial owner of each of the Properties, which are the only properties owned or occupied for the purpose of the Business or by a Target Company or in which it has any interest or has any liability.

- 10.3 Save as disclosed in Schedule 2 (*Properties*), each of the Properties is free from any mortgage, charge, rent-charge, lien, Encumbrance or other third party right in the nature of security and, so far as the Seller is aware, no such matter exists which is capable of registration against any of the Properties.
- 10.4 The relevant Target Company has vacant possession of each of the Properties vested in it and there are no circumstances known to the Seller which would entitle or require any landlord or any other person to exercise any powers of entry or right to forfeiture or right to take possession or which would otherwise restrict or terminate the continued sole and exclusive possession or occupation of each of the Properties by the relevant Target Company.
- 10.5 There are no insurance policies relating to any issue of title affecting any of the Properties.
- 10.6 In relation to the Leased Real Estate, the Seller's Group has paid all principal rent or licence fees which have become due and demanded.
- 10.7 No member of the Seller's Group has received written notice of any disputes or claims relating to any of the Properties.
- 10.8 So far as the Seller is aware, there is no planning or building regulations enforcement action or prosecution in progress in respect of the Properties or the Carrickmacross Site nor has any warning letter, enforcement notice or written complaint been served or threatened in writing by any regulatory authority or third party in respect of compliance with the Planning Acts.
- 10.9 All material outgoings in respect of the Properties have been paid when due and demanded.
- 10.10 All material covenants, restrictions, stipulations and other encumbrances affecting the Properties have been observed and performed in all material aspects.
- 10.11 In relation to the following properties occupied by the relevant Target Company (the **Occupied**Real Estate):

(i) Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan, A81 E027	a) part of Folio MN15520F shown crosshatched green; b) part of Folio MN705L shown shaded pink; and c) part of Folio MN1413F shown shaded yellow on Map 2 annexed to the Declaration of Identity of Peadar Stack dated 31 May 2021 located at 1.8.2.4.1 of the Data Room.
(ii) Coolattin, Shillelagh, Co. Wicklow, Y14 PY73, Ireland	a) part of Folio WW3057F and WW18025F shown shaded blue on Map 2 annexed to the Declaration of Identity of Peadar Stack dated 31 May 2021 located at 1.8.5.5.1 of the Data Room

the Seller warrants:

- (a) there are no insurance policies relating to any issue of title affecting the Occupied Real Estate.
- (b) no member of the Seller's Group has received written notice of any disputes or claims relating to the Occupied Real Estate.
- (c) all material outgoings in respect of the Occupied Real Estate insofar as they relate to the relevant Target Company have been paid when due and demanded
- (d) to the best of the Seller's knowledge, information and belief the relevant Target Companies have been in occupation and sole and exclusive possession of the Occupied Real Estate to the exclusion of all others for in excess of 15 years and no notice, dispute or claim adverse to the relevant Target Company's rights, title and interest in the Occupied Real Estate has been made against the relevant Target Company.

11 Environmental Matters

- 11.1 The Seller's Group is conducting, and has within the Relevant Period conducted, the Business in all material respects in accordance with all Environmental Laws affecting the Business.
- All material Environmental Permits and Environmental Instruments required for any activities of the Business have been obtained and, so far as the Seller is aware, and are being complied with in all material respects and are in full force and effect.
- 11.3 So far as the Seller is aware, there are no current circumstances which might give rise to any material non-compliance with, material contravention of, or material liability under any

Environmental Laws or Environmental Permits, or which might give rise to any Environmental Claim.

- 11.4 No member of the Seller's Group is currently and has not within the Relevant Period been involved in any Environmental Claim (whether as claimant, defendant or otherwise) in connection with the Business and, so far as the Seller is aware, no Environmental Claim has been threatened in writing against any member of the Seller's Group in connection with the Business.
- 11.5 So far as the Seller is aware, there is no pollution or contamination of the Environment on, in, at, under or migrating to or from the Properties, which is reasonably likely to give rise to any material liability, or obligation or duty requiring material expenditure, under Environmental Laws.
- 11.6 So far as the Seller is aware, no Target Company bears any liability with respect to the groundwater contamination present at, in or under the Windmill Lane Site (as detailed in the "EHS assessment" report prepared by ERM and included in the Data Room with reference number 1.3.1.10), and all Remedial Works which is or will be required in connection with such contamination is the responsibility of third parties.
- 11.7 No claims against the Seller's Group have been made by any person during the previous ten years with respect to personal injury, death, or other impairment or risk to the health of any person as a result of their exposure to asbestos or asbestos-containing materials, to the extent that such exposure occurred at any of the Properties or in connection with the operations of the Business or of the Target Companies, in each case prior to Completion.

12 Employees

- 12.1 For the purposes of this paragraph 12, Employment Legislation means legislation applying in the United Kingdom and/or the Republic of Ireland (as applicable) affecting contractual or other relations between employers and their employees or workers or any employee representative which is enforceable by any Target Company, any Employee or worker or employee representative.
- 12.2 So far as the Seller is aware, the relevant member of the Seller's Group has in relation to each of its Employees (and, insofar as relevant, in relation to each of its former, prospective and future employees) complied in all material respects with all material obligations under Employment Legislation, agreements (written, verbal and/or by custom and practice) with employee representatives and trade unions and terms and conditions of employment.
- 12.3 True, up-to-date and complete copies of all agreements relating to the provision of agency workers whose services are required to support the normal operation of the Business as at the

date of this Agreement where the number of agency workers supplied under the relevant agreement is over 50 at any one time, are contained in the Data Room.

- True, up-to-date and complete copies of the anonymised Employment Contracts of the Senior Employees and a representative sample of template Employment Contracts of other Employees and any material handbooks, policies and other documents currently in force governing the terms of employment of the Employees, the terms of all share incentive, share option, profit sharing, bonus, Transaction-related bonus, or other incentive schemes applicable to any of the Employees (in addition to the details (including award and vesting schedule in each case) of any specific outstanding awards made under any such incentive schemes to Employees) have been Disclosed.
- 12.5 Save as Disclosed as at the date of this Agreement no Employee is in receipt of any income protection payments under a relevant policy provided by the Seller.
- 12.6 The individuals whom the Seller acting reasonably has determined to be Central Services Employees as at the date of this Agreement have been identified (by role and reference number) on the employee census data which has been Disclosed.
- 12.7 Since the Accounts Date, no material change (other than changes made in the ordinary course of business, to effect the Hive Down, to effect the movement of the Central Services Employees to the Target Companies or as a result of a change in role or career advancement) has been made or has been proposed in the terms of employment or engagement of any material category of Employees or any Senior Employee.
- 12.8 There are no investigations existing, pending or threatened in writing affecting any Target Company which involve any material category or number of Employees or any Senior Employee by any governmental or statutory body in respect of employment matters.
- 12.9 There are no amounts due to or in respect of any Employee, worker or former employee or former worker which are in arrears and unpaid other than salary, wages, commission or pension for the month current at the date of this Agreement or in respect of annual bonus payments or the reimbursement of ordinary course business expenses.
- 12.10 Since the Accounts Date, no Senior Employee:
 - (a) has given notice or is under notice of dismissal;
 - (b) has been off sick for any period which has resulted in such Senior Employee having been off sick for a period of 40 days or more in any six-month period within the Relevant Period

- (whether or not consecutive) or is receiving or is due to receive payment under any sickness or disability or permanent health insurance scheme: or
- (c) is the subject of any current disciplinary warning or procedure or has raised a grievance under any grievance procedure or a protected disclosure under any whistleblowing policy which has not been resolved in accordance with any relevant grievance procedure or whistleblowing policy.
- 12.11 Neither this Agreement nor Completion will cause any Target Company or any Designated Buyer to be liable to a Senior Employee for any payment or any other benefit including any transaction bonus, or material enhancement or improvement to his/her remuneration and/or terms and conditions of employment.
- 12.12 Other than in connection with, or as a result of, the Hive Down, in the 24 months prior to the date of this Agreement in relation to the Business or to any of the Employees:
 - (d) no Target Company has given notice of any redundancies to any government department pursuant to section 193 Trade Union and Labour Relations (Consolidation) Act 1992 (TULRCA 1992), Article 221 of the Employment Rights (Northern Ireland) Order 1996 or the Protection of Employment Acts, 1977 to 2014; and
 - (e) there has not been any relevant transfer as defined in the Transfer Regulations.
- 12.13 The Data Room contains details of any enhanced redundancy payment scheme and any current custom and practice of making enhanced redundancy payments on redundancy operated in respect of the Central Services Employees or by the Target Companies during the Relevant Period.
- 12.14 There is no existing, pending or threatened in writing material industrial or trade dispute (official or unofficial) or negotiation between any member of the Seller's Group and any material number or category of the Employees or any trade union or other organisation formed for a similar purpose or any other employee representative(s).
- 12.15 The Data Room contains full and up-to-date particulars of all existing works councils and collective agreements, arrangements and understandings, in each case whether binding or not with any trade union, staff association or other body representing any Employee or which may affect any Employee.
- 12.16 Each Employee is legally permitted, or has the required authorisation, to work in the jurisdiction and location he / she is working in.

- 12.17 Insofar as any member of the Seller's Group has availed itself of any government funding, tax relief, grant or other governmental support relating to COVID-19, in relation to the Employees, the terms of such schemes have been adhered to in all material respects and no repayment or other material liability is owed by any of the Target Companies, and the Target Companies have not been notified in writing that any such liability will be owned in the future, in respect of any such scheme.
- 12.18 The Data Room contains details of all material claims pending or threatened in writing against any Target Company:
 - by any current Employee or former director or employee of any Target Company in respect of an accident or injury which is not covered by insurance; or
 - (b) by any current Employee or former director or employee of any Target Company in relation to their terms and conditions of employment, appointment or termination thereof.

13 Pensions

- 13.1 Save for the UK DC Section, the ROI Pension Schemes and the UK Life Assurance Arrangement and the PHI Arrangements neither the Seller's Group in relation to the Employees, nor the Target Companies in relation to the Employees, are under any legal obligation to provide or contribute towards any pension, lump sum, death, or ill health benefits to any of the present or former Employees engaged in the Business or their dependants, and no proposal or announcement has been made as to the introduction of any such arrangement.
- 13.2 The only benefits payable under the UK DC Section are "money purchase" benefits (as defined in section 181(1) of the Pension Schemes Act 1993) and the only benefits payable under the ROI DC Plan are those which would be capable of being paid under a "defined contribution scheme" (as defined in section 2 of the Irish Pensions Act 1990).
- 13.3 Details of the contribution rates payable to the UK DC Section and the ROI DC Plan have been disclosed to the Buyer.
- 13.4 All employer and employee contributions and premiums due to be deducted and remitted or paid by a Target Company as at the date of this Agreement to the UK DC Section, the ROI DC Plan and the UK Life Assurance Arrangement in respect of an Employee have been deducted and remitted or paid when due.
- 13.5 Any benefits payable on the death of an Employee whilst in employment (other than a return of an employee's own contributions to the UK DC Section or the ROI DC Plan as applicable, and

contributions paid in respect of him) are fully insured and all premiums due to the insurance company have been paid.

- 13.6 In relation to Employees located in the United Kingdom, the Seller's Group and the Target Companies have complied in all material respects with their duties to employees under the Pensions Act 2008 and utilises the UK DC Section to meet its employer duties under that Act.
- 13.7 In relation to Employees, the Seller's Group and the Target Companies have complied in all material respects with all applicable legal obligations in respect of (i) the PHI Arrangements in relation to Employees who are entitled to PHI benefits; and (ii) so far as the Seller is aware, the provision of retirement, death and disability benefits more generally.
- 13.8 In relation to the Employees, so far as the Seller is aware, there is no material proceeding or claim (other than routine claims for benefits) outstanding, pending or threatened against:
 - (a) the trustees or administrators of the UK DC Section;
 - (b) the trustees or administrators of the ROI DC Plan; or
 - (c) any Target Company,

in connection with the UK DC Section, the ROI DC Plan or the provision of any pension, lump sum, death, or ill health benefits.

- 13.9 There are no "excluded employees" (within the meaning given to that term in Section 121 of the Pensions Act 1990 (as amended) of Ireland (Section 121)) in respect of whom a Target Company is obliged to provide access to a standard personal retirement savings account in accordance with Section 121.
- None of the Employees or former employees of any Target Company has any right to benefits under or in relation to an occupational pension scheme which are not "old-age, invalidity or survivors' benefits" (as referred to in Article 4(a) of the EU Acquired Rights Directive 2001/23/EC and any local regulations implementing that Directive).
- 13.11 On Completion, the Target Companies will immediately cease to participate in and will cease to have any liability (whether relating to contributions (except for contributions deducted or falling due prior to Completion but not remitted to the UK DC Section and the ROI Pension Schemes), expenses or under any trustee indemnity) to the UK DC Section and the ROI Pension Schemes, and any Employees who at Completion have any entitlements to retirement benefits under the UK DC Section and the ROI Pension Schemes shall as from Completion become entitled to fully preserved benefits under the UK DC Section and the ROI Pension Schemes.

13.12 A copy of:

- (a) the governing provisions of the UK DC Section, UK DB Section and the ROI DC Plan;
- (b) an extract of the most recent "scheme return" (as that term is defined in the Pensions Act 2004) which was submitted to the UK Pensions Regulator prior to this Agreement in respect of the UK DB Section on behalf of the trustee of the Kerry Retirement Benefits Scheme and which states the only statutory employers in relation to the UK DB Section are Kerry Foods Limited and Kerry Ingredients (UK) Limited; and
- (c) a deed dated 2 April 2017 between Kerry Foods Limited, Kerry Ingredients (UK) Limited, the trustees of the Kerry Retirement Benefits Scheme, Noon Products Limited, Spurway Foods Limited and others in relation to a flexible apportionment arrangement relating to the Kerry Retirement Benefits Scheme,

are contained in the Data Room.

- 13.13 No Target Company is, or has ever been, a participating or statutory employer in the UK DB Section.
- 13.14 The UK DC Section is fully segregated from the UK DB Section and is treated as a separate scheme for statutory purposes.
- 13.15 No contribution notice or financial support direction under the Pensions Act 2004 has been issued to any Target Company in respect of the UK DB Section and so far as the Seller is aware there is no fact or circumstance likely to give rise to any such notice or direction.

14 Insurance

- 14.1 A list of the key insurance policies maintained by the Seller's Group in respect of the Business and currently in force (the **Policies**), together with details of all claims under the Policies in excess of £1,000,000 made under any policy of insurance in respect of the Business in the Relevant Period, are Disclosed in the Data Room. Except as Disclosed in the Data Room in accordance with this paragraph 14.1, there are no outstanding claims in excess of £1,000,000 under any Policy maintained by the Seller's Group, or circumstances which may give rise to any such claim(s).
- 14.2 All premiums due on the Policies have been paid and no member of the Seller's Group has received written notification that such Policies are invalid or unenforceable for any reason insofar as they relate to the Business.

14.3 The Seller's Group has at all times during the Relevant Period fully complied with all applicable statutory requirements to maintain sufficient insurance coverage in respect of the Business.

15 Intellectual Property Rights

- 15.1 Details of the following are Disclosed in the Data Room:
 - (a) all Intellectual Property Rights:
 - (i) for which any Target Company is or has applied to be registered as the proprietor;
 - that are owned by a member of the Retained Kerry Group and used or held for use in the Business;
 - (b) all material unregistered Intellectual Property Rights owned by any Target Company or by a member of the Retained Kerry Group and used or held for use in the Business; and
 - (c) all material licences of Intellectual Property Rights used in the Business to or from any third
 party to which any Target Company, or any member of the Retained Kerry Group, is a party
 (IP Licences),

(together, the Business IP).

- 15.2 All rights and interests in the Business IP:
 - (a) are legally and beneficially owned solely by a Target Company or a member of the Retained Kerry Group or are validly licensed to a Target Company or member of the Retained Kerry Group; and
 - (b) are not subject to any claim or opposition from any person as to title, validity or enforceability that would have a material adverse effect on the Business.
- 15.3 The Intellectual Property Rights owned by the Target Companies, or by a member of the Retained Kerry Group and used in the Business, are not subject to any Encumbrance.
- 15.4 All application and renewal fees payable in respect of the Intellectual Property Rights owned by, or applied for by, any Target Company or by any member of the Retained Kerry Group and used in the Business have been paid.
- 15.5 So far as the Seller is aware:

- (a) none of the operations of the Target Companies or the Business is infringing, nor has in the Relevant Period infringed, the Intellectual Property Rights of any other person and no third party has disputed the right of any Target Company or member of the Retained Kerry Group to use the Business IP; and
- (b) no third party is infringing, nor has in the Relevant Period infringed, the Intellectual Property Rights of any Target Company or any member of the Retained Kerry Group in relation to the Business and no Target Company or member of the Retained Kerry Group in relation to the Business has disputed the right of a third party to use the Intellectual Property Rights owned or used by the third party,

where, in either case, such infringement would be material to the Business.

- 15.6 So far as the Seller is aware, the IP Licences are valid and binding and each party thereto has complied with their terms in all material respects and they are not the subject of any claim, dispute or proceeding, pending or threatened in writing where such claim or dispute would have a material adverse effect on the Business.
- 15.7 With the exception of the Retained Name and the Intellectual Property Rights that shall be provided to the Target Companies under the Transitional Services Agreement (and any Intellectual Property Rights which relate to Excluded Services as defined in the Transitional Services Agreement), Meat Free Substitute Technology Licence and Brand Licences, the Retained Kerry Group does not make available to the Business, nor has made available in the Relevant Period, any other Intellectual Property Rights that are owned by the Retained Kerry Group and that are material to the Business.
- 15.8 So far as the Seller is aware, the Intellectual Property Rights that are (a) owned by the Target Companies; (b) expressly licensed in writing to the Target Companies; (c) acquired by the Target Companies under the IP Assignment Agreements and Hive Down IP Assignment Agreements; and (d) made available to the Target Companies under the Transitional Services Agreement, together comprise all of the Intellectual Property Rights that are required to carry on the Business after Completion substantially in the manner as it was carried on at the date of this Agreement and since 1 January 2021.
- 15.9 All Intellectual Property Rights and Know-how that, pre-Hive-Down, were owned by the Seller Group (other than the Existing Companies) and used exclusively in connection with the Business transferred to the Newly-Trading Companies as part of the Hive Down (other than the Excluded Intellectual Property Rights (as such term is defined in the Hive Down Agreements)).

16 Information Technology

- 16.1 So far as the Seller is aware, all of the Information Technology used in the Business is either owned by, or validly licensed, leased or supplied under contracts to a Target Company or a member of the Seller's Group (other than a Target Company). Details of all material Information Technology used by the Business are Disclosed in the Data Room.
- 16.2 So far as the Seller is aware, all material contracts in respect of the material Information Technology used by the Business are valid and binding and each party thereto has complied with their terms in all material respects and they are not the subject of any claim, dispute or proceeding, pending or threatened in writing where such claim or dispute would have a material adverse effect on the Business.
- 16.3 All material Information Technology used by the Business is functioning, and has throughout the Relevant Period functioned, without any problems or defects that would have or have had (as applicable) a material adverse effect on the Business.
- 16.4 So far as the Seller is aware, there have not in the Relevant Period been any logical or physical intrusions to the Information Technology used by the Business resulting in the loss or unauthorised access to any material data or material Know-how relating to the Business.
- 16.5 The Information Technology used by the Business includes up-to-date anti-virus software and security and back-up systems designed to preserve the availability, security and integrity of the material Information Technology used by the Business and any material data or Know-how relating to the Business stored on such Information Technology.
- 16.6 So far as the Seller is aware, the Information Technology that is (i) owned by the Target Companies; (ii) licensed, leased or supplied to the Target Companies under contracts with third parties; and (iii) made available to the Target Companies under the Transitional Services Agreement, together comprise all the Information Technology that is required to carry on the Business after Completion substantially in the manner as it was carried on at the date of this Agreement and since 1 January 2021.

17 Data protection

17.1 The Seller's Group and Target Companies are conducting the Business in all material respects in accordance with the Data Protection Legislation.

- 17.2 No Personal Data Breaches in respect of the Business have been reported to the relevant Supervisory Authority in the Relevant Period.
- 17.3 In connection with the Business, no member of the Seller's Group has in the Relevant Period received any formal written notice of any complaint or claim made by a Data Subject or any sanctions or enforcement action imposed by a Supervisory Authority.
- 17.4 The Target Companies, and the Seller's Group in connection with the Business, have taken appropriate and proportionate measures which are intended to manage the risks posed to the security of the IT and Personal Data (including the ability to resist action that compromises the availability, authenticity, integrity or confidentiality of data or outputs, or a material systems failure, data loss or theft, unauthorised access, malware attack or other security breach or failure (each a Security Incident)).
- 17.5 In connection with the Business, no member of the Seller's Group has in the Relevant Period been the subject of a Security Incident which has had a material impact on the continuity of the operations of the Business.

18 Litigation

- Apart from the collection of debts in the ordinary course of the Business (involving debts of not more than £1 million in any individual case or £5 million in aggregate) or ordinary course litigation involving employees or former employees, no Target Company, nor member of the Seller's Group in connection with the Business, is engaged in any capacity in any material litigation, arbitration, conciliation, mediation, prosecution or other legal proceedings or in any proceedings or hearings before any statutory or governmental body (including regulators), department, board or agency; and so far as the Seller is aware, no such matters have been threatened in writing.
- There is no outstanding judgment, order, decree, arbitral award or decision of any court, tribunal, arbitrator or governmental agency against a Target Company, or against a member of the Seller's Group in connection with the Business, which is material in the context of the Business.

19 Insolvency

- 19.1 No Target Company:
 - (a) is insolvent or unable to pay its debts;
 - (b) has ceased to carry on business, stopped payment of its debts or any class of them or entered into any compromise or arrangement in respect of its debts or any class of them, nor has any step been taken to do any of those things; or

(c) has been dissolved or entered into liquidation, administration, moratorium, administrative receivership, receivership, a voluntary arrangement, a scheme of arrangement with creditors, any analogous or similar procedure in any jurisdiction or any other form of procedure relating to insolvency, reorganisation or dissolution in any jurisdiction, nor has a petition been presented or other step been taken by any person with a view to any of those things.

19.2 The Seller has not:

- (a) become insolvent or unable to pay its debts;
- (b) ceased to carry on business, stopped payment of its debts or any class of them or entered into any compromise or arrangement in respect of its debts or any class of them, nor has any step been taken to do any of those things, where any such actions would affect its ability to enter into or perform its obligations under this Agreement, any of the Hive Down Agreements or any of the other Transaction Documents to which the Seller is or will be a party; or
- (c) been dissolved or entered into liquidation, administration, moratorium, administrative receivership, receivership, a voluntary arrangement, a scheme of arrangement with creditors, any analogous or similar procedure in any jurisdiction or any other form of procedure relating to insolvency, reorganisation or dissolution in any jurisdiction, nor has a petition been presented or other step been taken by any person with a view to any of those things, where, in any case, such action would affect its ability to enter into or perform its obligations under this Agreement, any of the Hive Down Agreements or any of the other Transaction Documents to which the Seller is or will be a party.

20 Tax

Accounts and Tax

20.1 All liabilities, whether actual, deferred, contingent or disputed, of each Target Company for Tax as at the Account Dates are provided for in accordance with relevant accounting standards or (as appropriate) disclosed in the Accounts. Since the Accounts Date, no Target Company has been involved in any transaction (other than the Hive Down) which has given or may give rise to a material liability to Tax on any Target Company (or would have given or might give rise to such a liability but for the availability of any Relief) other than Tax in respect of normal trading income or

receipts of the Target Company concerned arising from transactions entered into by it in the ordinary course of business.

20.2 All material Tax or amounts in respect of Tax due and payable by any Target Company prior to the date hereof has been paid in full.

Tax returns, corporation tax self assessment and records

20.3 All returns, computations, notices, accounts, statements, reports or information which ought to have been made by or in respect of each Target Company for any Tax purpose, have been properly and punctually submitted by each Target Company to the relevant Tax Authority; all such returns, computations, notices, accounts, statements, reports and information supplied to any Tax Authority are up to date, correct in all material respects and were made on a proper basis.

Withholdings

20.4 Each Target Company has made all deductions and retentions of or on account of Tax as it was or is obliged or is entitled to make and has accounted to the Tax Authority concerned for any such deductions and retentions for which it was obliged to account.

Disputes

20.5 No Target Company is involved in any current dispute with any Tax Authority. No Target Company is, or has in the six years prior to the date of this Agreement been, the subject of any review, audit, investigation or non-routine visit by any Tax Authority and, so far as the Seller is aware, there is no such review, audit, investigation or non-routine visit planned by any Tax Authority and, so far as the Seller is aware, there are no facts which might cause such an

investigation, enquiry, audit or non-routine visit to be instituted or which may be the subject of any dispute with any Tax Authority.

Penalties and interest

20.6 No Target Company is liable, nor has been liable within six years before the date of this Agreement, to pay any penalty, fine, surcharge or material amount of interest in connection with any Tax.

Employees

20.7 Each Target Company has properly operated the PAYE and National Insurance systems or equivalent social security contributions deducting and accounting for Tax and maintaining all such records, in each case as required by applicable law or regulation or any relevant Tax Authority.

Special arrangements with any Tax Authority

20.8 No Tax Authority has, in the last six years, operated any concession, agreement or other formal or informal arrangement with any Target Company which is not based on relevant legislation or published practice.

Residence

20.9 Each Target Company is, and has at all times, been resident in its jurisdiction of incorporation for tax purposes and is not and has not been treated as resident or liable to pay or register for Tax in any other jurisdiction for any Tax purposes (including for the purposes of any double taxation agreement).

VAT

20.10 Each Target Company is a registered and taxable person for the purposes of VATA or VATCA (as applicable), has complied in all material respects with VATA or VATCA (as applicable) and all

orders, regulations, provisions, directions or other conditions made or imposed thereunder or under any law relating to VAT.

20.11 No Target Company has applied to HM Revenue and Customs under section 43 VATA or section 43B VATA, or to the Revenue Commissioners of ROI (as applicable), to be treated as nor have two or more such companies been treated as a group for VAT purposes.

Groups

- 20.12 No Target Company is or could become liable to Tax by reason of the failure of any person other than another Target Company to pay or discharge when due any Tax properly assessed on that other person.
- 20.13 In the last six years, all material transactions between any Target Companies, or between (on the one hand) any Target Company and (on the other hand) the Seller or any other company that has been treated as a member of the same group as the Seller for any Tax purpose, have been on arm's length terms.

Stamp duty, stamp duty reserve tax and stamp duty land tax

20.14 All documents in the possession of any Target Company, to the production of which any Target Company is entitled or which establish or are necessary to establish the title of any Target Company to an asset and, in each case, which attract stamp or transfer duty in any jurisdiction have been duly stamped.

Implementation of this Agreement

20.15 The entry into and the implementation of the transactions contemplated by this Agreement will not result in: (i) any Target Company becoming liable to Tax by reason of that or any other Target Company leaving the Seller's Group for any Tax purpose; or (ii) any deemed disposal or realisation by any Target Company of any asset or liability for any Tax purposes; or (iii) the withdrawal, postponement, restriction or loss of any Buyer Relief (as defined in the Tax Covenant).

Section 980(8) of the TCA

20.16 No Sale Shares derive, directly or indirectly, their value or the greater part of their value from the assets specified in paragraphs (a) to (c) of Section 980(2) of the TCA and are not shares to which paragraph (e) of the said Section 980(2) of the TCA applies.

Schedule 5

Buyer Warranties

- 1 The Buyer and each Designated Buyer is validly incorporated, in existence and duly registered under the laws of its jurisdiction of incorporation and not liable to tax elsewhere.
- 2 The Buyer and each Designated Buyer has the right, power and authority and has taken all action necessary to execute and deliver, and to exercise its rights and perform its obligations under, this Agreement and each of the other Transaction Documents executed or to be executed by it at or before Completion, and each such document will, when executed, constitute legal, valid and binding obligations of the Buyer and each Designated Buyer.
- Other than to the extent relevant for the Conditions, no consent, authorisation, licence or approval of any governmental, administrative, judicial or regulatory body, authority or organisation is required to authorise the execution, delivery, validity, enforceability or admissibility in evidence of this Agreement or any of the other Transaction Documents or the performance by the Buyer or any Designated Buyer of its obligations under this Agreement or any of the other Transaction Documents to which the Buyer or any Designated Buyer is or will be a party.
- 4 Each Designated Buyer has appointed the Buyer as its agent under Part B of Schedule 10 (Agency provisions) or has otherwise entered into an agency agreement with the Buyer substantially in the form of the Standard Transaction Agency Agreement.
- 5 The Buyer and each Designated Buyer has not:
 - (a) become insolvent or unable to pay its debts.
 - (b) ceased to carry on business, stopped payment of its debts or any class of them or entered into any compromise or arrangement in respect of its debts or any class of them, nor has any step been taken to do any of those things, where any such actions would affect its ability to enter into or perform its obligations under this Agreement or any of the other Transaction Documents to which the Buyer or each Designated Buyer is or will be a party.
 - (c) been dissolved or entered into liquidation, administration, moratorium, administrative receivership, receivership, a voluntary arrangement, a scheme of arrangement with creditors, any analogous or similar procedure in any jurisdiction or any other form of procedure relating to insolvency, reorganisation or dissolution in any jurisdiction, nor has a petition been presented or other step been taken by any person with a view to any of those things, where, in any case, such action would affect its ability to enter into or perform its

- obligations under this Agreement or any of the other Transaction Documents to which the Buyer or each Designated Buyer is or will be a party.
- The Buyer and each Designated Buyer has not been notified in writing in the last three years that any investigation, review, enquiry, proceeding or claim in respect of the Buyer or a Designated Buyer is being or has been initiated, conducted or is pending by any governmental, administrative, judicial or regulatory or other body, authority or organisation in relation to any Financial Crime Law or the UK Modern Slavery Act 2015 and, so far as the Buyer is aware, there are no circumstances likely to give rise to any such investigation, enquiry, proceeding or claim.
- 7 To the extent relevant to the transaction contemplated by this Agreement, each of the Buyer and each Designated Buyer is conducting, and has at all times during the Relevant Period conducted, its business in all material respects in accordance with all applicable laws and regulations of the United Kingdom and the Republic of Ireland.

Schedule 6

Limitations on the liability of the Seller

1 Financial limits

Minimum claims

1.1 The Seller shall not be liable in respect of any Seller Warranty Claim or any Tax Covenant Claim unless the amount claimed by the Buyer exceeds [***].

Aggregate minimum claims

- 1.2 The Seller shall not be liable in respect of any Seller Warranty Claim unless the aggregate amount of all Seller Warranty Claims for which the Seller would be liable but for this paragraph 1.2 exceeds in aggregate [***], in which case the Seller shall (subject to the other provisions of this Agreement) be liable for the whole amount claimed and not only the excess.
- 1.3 For the avoidance of doubt, the Buyer may give notice of any single Seller Warranty Claim or Tax Covenant Claim in accordance with and for the purpose of paragraph 2.1, irrespective of whether, at the time the notice is given, the amount set out in paragraph 1.2 has been exceeded.

Maximum liability

- 1.4 The Seller's maximum aggregate liability:
 - in respect of all Seller Warranty Claims (other than Fundamental Warranty Claims and Tax Warranty Claims) shall not exceed [***];
 - (b) in respect of (i) all Tax Claims; (ii) all Indemnity Claims; and (iii) all Relevant Claims under paragraph 1.4(a) above, shall not exceed [***]; and
 - (c) in respect of all Relevant Claims (including those claims under paragraphs 1.4(a) and 1.4(b) above) shall not exceed an amount equal to the Purchase Price (ignoring any adjustment pursuant to clause 4.13).

Relevant Costs

1.5 For the purposes of paragraph 1.4 only, the amount of any costs, expenses and other amounts (including interest) claimed by the Buyer, or ordered or determined to be payable to the Buyer (together with any irrevocable VAT thereon) under, in connection with or arising out of any Relevant Claim (or any judgment, order or award) (Relevant Costs) shall be included in the sums

referred to therein as a liability in respect of the Relevant Claim. For the avoidance of doubt, Relevant Costs shall be excluded from the sums referred to in paragraphs 1.1 and 1.2.

Conversion Rate

1.6 If any monetary sum to be taken into account for the purposes of determining the amount of any Relevant Claim is expressed in a currency other than pounds sterling (GBP), the amount of that monetary sum shall be translated into sterling at the Conversion Rate at the close of business in London on the Business Day immediately preceding the date of the notice of that Relevant Claim.

2 Time limits

Notice requirements

- 2.1 If the Buyer becomes aware of a matter or circumstance which gives rise to or may give rise to a Relevant Claim (other than a Tax Claim which shall be dealt with in accordance with paragraph 4(a) of Schedule 2 to the Tax Covenant), the Seller shall not be liable in respect of it unless the Buyer provides notice to the Seller in accordance with the provisions of this Agreement specifying that matter or circumstance in reasonable detail (including, where practicable, the Buyer's reasonable estimate of the amount of such Relevant Claim) as soon as reasonably practicable after the Buyer became aware of that matter or circumstance and in any event within the time periods set out in paragraph 2.2.
- 2.2 The Seller shall not be liable in respect of a Relevant Claim unless the Buyer has given notice in accordance with paragraph 2.1:
 - (a) in the case of a Tax Claim, by no later than the earlier of:
 - (i) the expiry of the relevant statute of limitations for the Tax Claim in question; and
 - (ii) the date falling seven years after the Completion Date;
 - (b) in the case of:
 - a claim under the Seller Warranties relating to Environmental matters contained in paragraph 11 of Schedule 4; or
 - (ii) a claim under the Fundamental Warranties,

by no later than the date falling three years after the Completion Date;

- (c) in the case of a claim under the Seller Warranties relating to Pensions matters contained in paragraph 13.1 of Schedule 4, by no later than the date falling six years after the Completion Date; and
- (d) in the case of any other Relevant Claim, by no later than the date falling 18 months after the Completion Date.

Commencement of proceedings

- 2.3 The Seller's liability in respect of any Relevant Claim (other than a Tax Claim) shall terminate (if such claim has not been previously remedied, satisfied or withdrawn) on the date falling 12 months after the date it was first notified to the Seller in accordance with paragraph 2.1 of this Schedule unless court proceedings in respect of it have been commenced within such period except:
 - (a) where the Relevant Claim relates to a contingent liability, in which case it shall be deemed to have been withdrawn unless court proceedings in respect of it have been commenced by being both issued and served within 12 months of it having become an actual liability; or
 - (b) where the Relevant Claim is a Seller Warranty Claim of which notice is given for the purposes of paragraph 2.1 at a time when the amount set out in paragraph 1.2 has not been exceeded, in which case it shall be deemed to have been withdrawn unless court proceedings in respect of it have been commenced by being both issued and served within 12 months of the date of any subsequent notification to the Seller pursuant to paragraph 2.1 of one or more Seller Warranty Claims which result(s) in the total amount claimed in all Seller Warranty Claims notified to the Seller pursuant to paragraph 2.1 exceeding the amount set out in paragraph 1.2 for the first time.

For these purposes, **court proceedings** shall mean any proceedings, suit or action before the courts of England and Wales and court proceedings shall not be deemed to have been commenced unless they have been both issued and validly served on the Seller in accordance with relevant procedural rules.

3 Remediable breaches

Where the matter or default giving rise to a Relevant Claim (other than a Tax Covenant Claim) is capable of remedy, the Seller will not be liable for such Relevant Claim unless the matter or default is not remedied to the reasonable satisfaction of the Buyer within 20 Business Days after the date on which notice of such Relevant Claim is given to the Seller pursuant to paragraph 2.1 of this

Schedule. If the breach has not been remedied within that 20 Business Day period, then the date on which notice of a Relevant Claim in respect of that breach shall be deemed to have been given to the Seller for the purpose of paragraph 2.1 shall be the date on which notice was given under this paragraph 3, provided that the notice satisfied the other requirements of paragraph 2.1 when so given.

4 General limitations

- 4.1 The Seller shall not be liable in respect of a Relevant Claim (other than a Tax Claim, to which the limitations at clause 3 of the Tax Covenant shall apply) (or such liability will be reduced) to the extent that the matter giving rise to, or the loss arising from, or any increase in that claim:
 - (a) occurs as a result of or is otherwise attributable to any introduction, enactment, change, amendment or withdrawal of any enactment, regulation, rules of any regulator or administrative practice or guidance occurring after the date of this Agreement (whether or not that introduction, enactment, change, amendment or withdrawal purports to have retrospective effect in whole or in part), or any change in the interpretation of any of the foregoing by any court of law or tribunal after the date of this Agreement;
 - (b) occurs as a result of or is otherwise attributable to:
 - any change in the accounting reference date of any member of the Buyer's Group after Completion;
 - a change in the accounting or Tax policies, bases or practices of any member of the Buyer's Group (including the method of submitting Tax returns) introduced or having effect after Completion;
 - (iii) any increase after the date of this Agreement in any rate of Tax or any change in the tax residence of any member of the Buyer's Group;
 - (iv) any member of the Buyer's Group disclaiming any part of the benefit of capital or other allowances against Tax claimed or proposed to be claimed on or before Completion or which was otherwise taken into account in computing the provisions for Tax in the Accounts or the Completion Statements;
 - (v) any failure or omission to make any valid claim, election, surrender or disclaimer, to give any valid notice or consent or to do any other thing under the provisions of any law, regulation or guidance relating to Tax after Completion, which was claimed or proposed to be claimed on or before Completion or the making, giving or doing of

which was taken into account in computing the amount of any allowance, provision or reserve for Tax in the Accounts or the Completion Statements; or

- (vi) any claim, election, surrender or disclaimer made or notice or consent given after Completion under the provisions of any law, regulation or guidance relating to Tax, other than any claim, election, surrender or disclaimer, notice or consent assumed to have been made, given or done in computing the amount of any allowance, provision or reserve for Tax in the Accounts or the Completion Statements;
- (c) would not have arisen or occurred but for:
 - (i) an act, omission or transaction on the part of; or
 - (ii) any change to, or non-compliance with, the operating procedures, policies or practices of any Target Company as at Completion by,

any member of the Buyer's Group or any of their respective directors, employees or agents after Completion otherwise than as required by law or pursuant to a legally binding commitment of the Business created on or before Completion;

- (d) arises as a consequence of any act or omission carried out (i) in consequence of the execution and performance of and in accordance with the terms of this Agreement, or (ii) at the request or with the written consent or approval of a member of the Buyer's Group;
- (e) was provided or reserved for (and not released prior to Completion) in the Accounts or the Completion Statements (including in the calculation of any allowance, provision or reserve in the Accounts or the Completion Statements), or was referred to in or in the notes to the Accounts or in accordance with generally accepted accounting principles has not been so taken account of or referred to, provided that, for the avoidance of doubt, liability shall not be excluded by virtue of this paragraph to the extent that any such provision or reserve is inadequate); or
- (f) would not have arisen but for any change in ownership, reorganisation, amalgamation, restructuring (solvent or otherwise) or cessation of business of any member of the Buyer's Group after Completion.

5 No double recovery

The Buyer agrees with the Seller (for itself and as trustee for the other members of the Seller's Group) that, in respect of any matter which may give rise to a liability (including for the avoidance

of doubt any indemnities) under this Agreement and/or any of the other Transaction Documents and/or the Hive Down Agreements, no such liability shall be met more than once.

6 Insurance

If in respect of any matter which would otherwise give rise to a Relevant Claim, any member of the Buyer's Group makes a recovery under any policy of insurance (or would have been so entitled had the Business maintained in force insurance cover on the terms in place immediately before Completion), that Relevant Claim shall be reduced by the net amount (after taking account of any excess and impact on premia) which that member of the Buyer's Group recovers under such policy of insurance.

7 Claims by or against third parties

- 7.1 If any member of the Buyer's Group becomes aware of any matter:
 - (a) which has given, or might give rise, to a claim being made by a third party against a member of the Buyer's Group which will or may give rise to a Relevant Claim (other than a Tax Claim); or
 - (b) in respect of which any member of the Buyer's Group is or may become entitled to recover (whether by way of payment, discount, credit, set-off, counterclaim or otherwise) from any third party (other than pursuant to a policy of insurance referred to in paragraph 6) any sum in respect of any loss, damage or liability which has been, is or may become the subject of a Relevant Claim (other than a Tax Claim),

in each case a **Third Party Claim** (including for the avoidance of doubt, a prospective claim), then the following provisions of this paragraph 7 shall apply.

- 7.2 The Buyer shall, as soon as is reasonably practicable upon any member of the Buyer's Group becoming aware of a Third Party Claim, give notice (containing reasonable details of the Third Party Claim) to the Seller of the matter and shall, at the Seller's cost, consult with the Seller with respect to that Third Party Claim and keep the Seller reasonably informed of all material developments in relation to that Third Party Claim.
- 7.3 The Buyer shall not, and shall procure that no member of the Buyer's Group shall, make any admission of liability in respect of a Third Party Claim, or agree, compromise or settle a Third Party Claim, without the prior written consent of the Seller (such consent not to be unreasonably withheld, conditioned or delayed).

- 7.4 Subject to paragraph 7.8(d), the Buyer shall provide, and shall procure that each member of the Buyer's Group shall provide, at the Seller's cost, the Seller and the Seller's professional advisers with access at reasonable times (and on reasonable prior notice) to premises and personnel and to all relevant documents, records, correspondence, accounts and other information within the power, possession or control of any member of the Buyer's Group for the purpose of investigating the Third Party Claim and/or ascertaining whether any member of the Buyer's Group has a right of recovery against any person other than the Seller and/or enabling the Seller to take such action as is referred to in paragraph 7.7.
- 7.5 Upon any member of the Buyer's Group becoming aware of a Third Party Claim, the Buyer shall, and shall procure that each member of the Buyer's Group shall, retain and preserve all relevant documents, records, correspondence, accounts and other information within the power, possession or control of any such persons which are or could reasonably be considered relevant in connection with the Third Party Claim for so long as any such Third Party Claim remains outstanding.
- 7.6 The Seller shall be entitled, at its sole cost, to copies of any of the documents, records and information referred to in paragraph 7.5 or, in the event that the Seller wishes to insure against its liabilities in respect of any actual or prospective Relevant Claim, any information that a prospective insurer may reasonably require before effecting such insurance.
- 7.7 The Buyer shall, and shall procure that each member of the Buyer's Group shall:
 - (a) take such action and institute such proceedings, and give such information and assistance, as the Seller may request to dispute, resist, appeal, compromise, defend, remedy or mitigate the Third Party Claim or to enforce against any person (other than the Seller) the rights of any member of the Buyer's Group in relation to the Third Party Claim (including the assignment to the Seller or such other third party as the Seller may direct of any rights of action which any member of the Buyer's Group may have), and more generally co-operate fully and promptly with the Seller and its professional advisers; and
 - (b) in connection with any proceedings related to the Third Party Claim (other than against the Seller) use professional advisers nominated by the Seller and, if the Seller so request, permit the Seller to have exclusive conduct of the negotiations and/or proceedings,

and the Seller shall indemnify the Buyer (or the relevant member of the Buyer's Group) in respect of all costs, losses, expenses and/or other amounts (including interest) properly and reasonably incurred by the Buyer and/or any other member of the Buyer's Group pursuant to its obligations under this paragraph 7.7.

- 7.8 Notwithstanding any other provision of this Agreement:
 - (a) subject to the Seller indemnifying the Buyer to the Buyer's reasonable satisfaction in respect of all costs, expenses and/or other amounts (including interest) that the Buyer or relevant member of the Buyer's Group incurs in connection with the following actions, the Seller shall be entitled at any stage and at its sole discretion to settle any such Third Party Claim or direct the Buyer to settle such Third Party Claim as applicable and the Seller shall, where practicable, notify the Buyer in advance of its decision to do so;
 - (b) if the Buyer declines or fails to comply with a direction to settle within ten Business Days of a written direction to do so, the Seller shall thereupon cease to have any obligation to indemnify the Buyer in relation to such Third Party Claim pursuant to paragraph 7.7;
 - (c) the Seller shall have no obligation to make payment to the Buyer in respect of any Relevant Claim until the Third Party Claim related thereto (if any) has been agreed by the Buyer and the Seller, determined by an non-appealable court decision or is withdrawn by the claimant in question; and
 - (d) the Buyer shall not be required to provide or procure the provision of any document in respect of which legal privilege can properly be claimed by the Buyer where to do so would cause the loss of such privilege in respect of the advice contained in such document.
- 7.9 For the avoidance of doubt, the provisions of this paragraph 7 shall not apply to any Tax Claim.

8 Calculating the liability of the Seller

- 8.1 If, after the Seller has made any payment in respect of a Relevant Claim (other than a Tax Claim), the Buyer or a member of the Buyer's Group recovers or becomes entitled to recover from a third party (including any Tax Authority) (whether by payment, discount, credit, relief or otherwise) a sum which is referable to that payment (a Recovery Amount), then the Buyer shall be obliged to procure the pursuit of such Recovery Amount (if not already received) and shall forthwith following recovery repay (or procure the repayment) to the Seller of an amount equal to the lesser of any amount actually recovered (net of Tax and after the deduction of reasonable costs incurred in achieving such recovery) and the sum paid by the Seller.
- 8.2 In calculating the liability of the Seller, there shall be taken into account the amount by which any Tax, for which any member of the Buyer's Group is now or would in the future be accountable or liable to be assessed, is or will be reduced or extinguished as a result of the matter or circumstance giving rise to the Relevant Claim in question, such Tax to be calculated on the basis of rates of Tax prevailing at the time of such Relevant Claim.

9 Consequential loss

Notwithstanding anything to the contrary in this Agreement, the Seller shall not in any circumstances be liable to the Buyer, whether in contract, tort or breach of statutory duty or otherwise for any consequential loss, loss of profit or loss of reputation.

10 Fraud

Nothing in this Schedule, nor any other provision of this Agreement purporting to limit or exclude the Seller's liability, shall apply to any claim to the extent that it (or the delay in the discovery of it) is the result of, or is increased as a result of any fraud or fraudulent misrepresentation on the part of the Seller or any other member of the Seller's Group or any officer of the Seller or any other member of the Seller's Group.

11 Duty to mitigate

The Buyer shall (and shall procure that each member of the Buyer's Group shall) mitigate any loss suffered by it (or such member of the Buyer's Group) which has resulted in or could result in a Relevant Claim (other than a Tax Covenant Claim).

Schedule 7

Pre-Completion Conduct and Undertakings

- 1 Pending Completion, the Seller shall, and shall procure that the other members of the Seller's Group shall, to the extent permitted under applicable laws carry on the Business as a going concern in the way carried on prior to the date of this Agreement.
- Pending Completion, the Seller shall not, and shall procure that no other member of the Seller's Group shall, do or agree to do any of the following in relation to the Business:
 - enter into, terminate, materially amend or materially vary any Material Contract or any contract, transaction or arrangement which would be a Material Contract if it subsisted at the date of this Agreement;
 - (b) sell, licence, dispose of, terminate the right to use, fail to take any action to defend or preserve or permit the lapse, abandonment, surrender or, for applications, nonprosecution, of any rights related to material Intellectual Property Rights which:
 - (i) are owned or used by any Target Company; or
 - (ii) will be so owned by any Target Company upon completion of the Hive Down or pursuant to the IP Assignment Agreements;
 - (iii) will be licensed to a Target Company by a member of the Seller's Group pursuant to the Meat Free Substitute Technology Licence and Brand Licences;
 - enter into any agreement (including any transfer or licence) relating to, or grant, modify, change or terminate any rights relating to, the material Intellectual Property Rights set out in (b)(i) to (b)(iii)) above;
 - (d) allow any Target Company to:
 - acquire or agree to acquire any share, shares or other interest in any company, partnership or other venture or incorporate any new subsidiary;
 - give any Guarantee in respect of the obligations of any third party other than for amounts which are not material and are in the ordinary course of business;
 - (iii) enter into any borrowing, factoring or other financing commitments (other than the use of overdraft facilities agreed before the date of this Agreement) or lend any money to, or enter into any lending commitments with, any person (other than a

- member of the Seller's Group), being in each case commitments which are outside the ordinary course of its business;
- (iv) declare, authorise or pay any dividend or other distribution (whether in cash, stock or in kind) or reduce, purchase or redeem of any part of the paid-up share capital of any Target Company;
- (v) create, allot or issue, or grant any option over or other right to subscribe for any share or other security of any Target Company; and/or
- (vi) enter into any transaction with any member(s) of the Seller's Group other than on terms which are equivalent to arm's length terms;
- (e) create any Encumbrance over the material assets of any Target Company or dispose of any material assets of any Target Company, in each case other than in the ordinary course of business;
- commence or instigate any litigation, arbitration, prosecution or other legal proceedings against any person, body or authority (other than in respect of ordinary course debt collection for amounts which are not material);
- (g) make any admission of liability, or any agreement, settlement or compromise of any claims, disputes or litigation with any person, body or authority in relation to the Business;
- (h) terminate the employment contract of any Senior Employee other than validly for gross misconduct pursuant to the terms of their employment contract;
- enter into any new arrangement in respect of, or grant, any transaction or retention bonuses (whether payable in cash or otherwise) to any Employee in connection with implementation of the transactions set out in this Agreement;
- (j) make any change in the terms of employment (including pension fund commitments) of Employees other than those required by law, which could increase in aggregate the total staff costs of the Target Companies by more than three per cent. per annum;
- (k) except to replace employees on substantially the same terms or to effect the movement of the Central Services Employees to the Target Companies, employ or agree to employ any new persons fully or part time where the total staff costs of the Target Companies would be increased in aggregate by more than three per cent. per annum or dismiss any existing employees (except for incompetence or gross misconduct or other reasonable cause

- justifiable by law) where the total staff costs of the Target Companies would be reduced in aggregate by more than three per cent. per annum;
- enter into or undertake to enter into or materially amend any collective bargaining agreement, or other commitment, undertaking, site guarantee or agreement (including in relation to wage increases) with or towards any union, works council or other employee representative body;
- (m) do or fail to do anything which would be reasonably likely to result in the termination, revocation, suspension or non-renewal of any Approval;
- (n) in connection with the Leased Real Estate, terminate or serve notice to terminate, materially
 alter or materially vary, surrender or accept any surrender of or waiver of any material term
 of any lease;
- (o) change the jurisdiction of residence for Tax purposes of any Target Company; create any place of business of a Target Company in any jurisdiction in which that Target Company is not, as at the date of this Agreement, resident for Tax purposes; amend any material Tax return of a Target Company or amend or withdraw any material claim, election, surrender, notice or consent of any Target Company for Tax purposes; or cause any Target Company to surrender or allocate to or claim from any entity (other than another Target Company) any Relief, disallowance or other allocation for Tax purposes, other than in each case as assumed in the Accounts;
- (p) materially alter, amend or vary the accounting policies of any Target Company unless such alteration, amendment or variation is required by law or relevant accounting requirements and standards:
- take steps to effect a winding up, liquidation, administration, receivership or other analogous insolvency proceedings of any Target Company; and/or
- (r) enter into any agreement (conditional or otherwise) to do any of the foregoing.
- 3 Paragraphs 1 and 2 above shall not apply in respect of, and shall not operate so as to restrict or prevent:
 - (a) any matter reasonably undertaken by any member of the Seller's Group in an emergency
 or disaster situation with the intention of and to the extent only of those matters reasonably
 required with a view to minimising any adverse effect thereof (and of which the Buyer will
 be notified as soon as reasonably practicable);

- (b) any matter expressly permitted by, or necessary for the performance of, this Agreement, the other Transaction Documents and the Hive Down Agreements, or necessary for the Hive Down or Completion;
- (c) the entering into of any Transaction Document;
- (d) any matter undertaken with the prior written consent of the Buyer (not to be unreasonably withheld, conditioned or delayed);
- (e) any matter to the extent required by law; and/or
- (f) any steps required for Target Companies to be able to (or to be funded to) repay or recover any Inter-Company Balances or pay up excess cash including by way of dividend, capital contribution or share issuance, provided such payment, repayment or recovery does not give rise to a liability to Tax of such Target Company (except to the extent such liability has been or will be provided for in the Completion Statements or otherwise taken into account as a liability (other than a provision for deferred taxation) in the preparation of the Completion Statements, or to the extent that the Buyer is indemnified in respect of such liability to its reasonable satisfaction by the Seller).
- 4 The parties acknowledge that the undertakings in paragraphs 1 and 2 have been included with a view to preserving the value of the Business and in particular that the occurrence of the matters set out in paragraph 2 may adversely affect the value of the Business.
- To the extent permissible under applicable antitrust laws, from the date of this Agreement until Completion:
 - (a) the Seller shall, in order to provide the Buyer with sufficient visibility on the financial performance of the Target Companies with a view to preserving their value, keep the Buyer reasonably informed of the Target Companies' overall financial performance prior to Completion by:
 - at the request of the Buyer, providing the Buyer with aggregated summary financial information in relation to the Target Companies within ten Business Days of the end of each quarter; and
 - (ii) at the reasonable request of the Buyer, meeting with the Buyer's representatives no more frequently than once each quarter to discuss such financial information;

- (b) each of the Seller and the Buyer shall, and shall procure that members of the Seller's Group and Buyer's Group respectively shall, commence the migration planning activities envisaged by clauses 2.1, 2.2 and 5.5 to 5.7 of the Transitional Services Agreement; and
- (c) the Seller shall co-operate with the Buyer and facilitate engagement between the Buyer, the Seller and the EPA with regard to putting in place replacement EPA Financial Provisions relative to the IED Licences for the ROI Sites (if deemed necessary by the EPA).

Schedule 8 Completion

Part A: Seller's obligations

- 1 At Completion the Seller shall:
 - repay, or procure the repayment of, the Inter-Company Payables and the Inter-Company Receivables in accordance with clauses 9.1(b) and 9.1(c);
 - (b) terminate or procure the termination of the Related Party Agreements pursuant to clause 11.1;
 - (c) shall deliver (or procure the delivery) to the Buyer:
 - (i) transfers in respect of the Sale Shares duly executed and completed in favour of the relevant Designated Buyer, together with the certificate(s) for the Sale Shares (or an indemnity for lost share certificates, in the agreed form) and copies of the duly executed powers of attorney or other authorities under which any of the transfers have been executed:
 - voting powers of attorney in the agreed form duly executed by each of the Designated Sellers in favour of the relevant Designated Buyers;
 - (iii) letters of resignation (executed as deeds and in customary form) from all the directors and the secretary of each Target Company (other than any director or secretary whom the Buyer wishes to continue in office) resigning their offices with such resignations to take effect from Completion;
 - (iv) (if not previously so delivered) a copy of the minutes of a duly convened board meeting of the Seller and each Designated Seller as appropriate, or a written resolution of the directors of the Seller and each Designated Seller, authorising the sale of the relevant Sale Shares and the execution and performance of its obligations under this Agreement and such other Transaction Documents to which it is or is to be a party;
 - a copy of the minutes of each duly convened meeting, or written resolution, of the directors, referred to in paragraph 2 below;
 - (vi) by making such documents available at relevant Target Company's registered office, the statutory books of each Target Company;

- (vii) where relevant, the email address, security code and authentication code used by each Target Company for making electronic filings with the Registrar of Companies in the UK and Northern Ireland or the Companies Registration Office in Ireland;
- (viii) a counterpart of the Transitional Services Agreement duly executed by the Seller;
- (ix) a counterpart of the Van Sales Distribution Agreement duly executed by the Seller or such member of the Seller's Group as the Seller shall nominate;
- the Meat Free Substitute Technology Licence in the agreed form duly executed by Kerry Luxembourg SARL and Northfield Foods Limited;
- (xi) the Everwell Licence in the agreed form duly executed by Kerry Luxembourg SARL and such member of the Seller's Group as the Seller shall nominate;
- (xii) the Galtee Licence in the agreed form duly executed by Duffy Meats Limited and Breeo Foods Limited;
- (xiii) the Raw Material Supply Agreement in the agreed form duly executed by Kerry Ingredients & Flavours Limited and Northfield Foods Limited;
- (xiv) the Galtee IP Assignment Agreements duly executed by all parties;
- (xv) a counterpart of the Tax Covenant duly executed by the Seller;
- (xvi) (if not previously so delivered) a counterpart of the Disclosure Letter duly executed by the Seller and record of the materials stored within the Data Room; and
- (xvii) the Tax reference number of the Designated Seller(s) of the Irish Sale Shares for the purposes of the Stamp Duty (E-stamping of Instruments and Self-Assessment) Regulations 2012 of Ireland.
- At Completion, the Seller shall procure that each Target Company holds a duly convened meeting of its directors, or passes a written resolution of its directors, at or pursuant to which it is resolved that (in each case with effect from Completion):
 - the transfers mentioned in paragraph 1(c)(i) above be registered (subject to them being duly stamped) notwithstanding any provision to the contrary in the articles of association of that Target Company;
 - (b) each of the persons nominated by the Buyer be validly appointed as directors and/or secretary of that Target Company;

- (c) the resignations of the directors and secretaries of that Target Company, referred to in paragraph 1(c)(iii) above, be tendered and accepted;
- the registered office of that Target Company be changed to such place as the Buyer may nominate; and
- (e) all existing instructions and authorities to bankers are revoked.

Part B: Buyer's obligations

1 At Completion:

- each Designated Buyer shall pay to the Seller in accordance with clause 29, the Initial Purchase Price allocated to the Sale Shares in respect of which it is the Designated Buyer (as set out in Schedule 1); and
- (b) the Buyer shall deliver to the Seller:
 - (i) (if not previously so delivered) a copy of the minutes of a duly convened board meeting of the Buyer and each Designated Buyer, or a written resolution of the directors of the Buyer and each Designated Buyer, authorising the purchase of the relevant Sale Shares and the execution of, and the performance of its obligations under, this Agreement and such other Transaction Documents to which it is or will be a party;
 - (ii) (if not previously so delivered) a counterpart of the Disclosure Letter duly executed by the Buyer;
 - (iii) a counterpart of the Transitional Services Agreement duly executed by the Buyer;
 - (iv) a counterpart of the Van Sales Distribution Agreement duly executed by the Buyer or such member of the Buyer's Group as the Buyer shall nominate; and
 - (v) a counterpart of the Tax Covenant duly executed by the Buyer.

Schedule 9

Completion Statements

Part A - The Preparation of the Completion Statements

Preparation of draft Completion Statements

- As soon as reasonably practicable, and in any event by no later than 45 Business Days, following Completion, the Seller shall prepare and deliver to the Buyer a draft of each of the Completion Statements substantially in the form set out in Parts C and D of this Schedule (the **Draft Completion Statements**), such Draft Completion Statements to include or be accompanied by:
 - (a) a statement of Actual Net Indebtedness;
 - (b) a statement of Actual Working Capital; and
 - (c) a statement as to the amount to be paid by either the Buyer or the Seller pursuant to clause4.11.
- 2 In preparing the Draft Completion Statements:
 - reference shall be made to the definitions of the relevant items to be included in the Draft Completion Statements as set out in this Agreement;
 - (b) in applying such definitions and determining the items and amounts to be included in the Draft Completion Statements, the following hierarchy of accounting treatment shall be applied:
 - (i) first, the specific polices and accounting treatments set out in Part B of this Schedule;
 - (ii) secondly, and to the extent not covered by paragraph (i) above (provided they are not inconsistent with IFRS), the same accounting principles, policies, evaluations, categorisations, procedures, techniques, methods, bases, practices and estimation

- techniques as were used in the preparation of the Carve-Out Accounts, consistent with historical practice applied in preparing the Accounts; and
- (iii) thirdly, to the extent not covered by paragraphs (i) and (ii) above, on a going concern basis in accordance with IFRS in force and applicable to the Target Companies at the Accounts Date.

An illustrative example of Net Indebtedness and Working Capital as at 31 December 2020 is set out in column (B) of Part C and Part D respectively.

Notification of Disputed Items and proposed adjustments

- 3 As soon as reasonably practicable, and in any event within 35 Business Days, following the delivery of the Draft Completion Statements to the Buyer in accordance with paragraph 1 above, (the Review Period), the Buyer shall review the Draft Completion Statements and shall give notice in writing to the Seller either:
 - (a) specifying any item in the Draft Completion Statements that it wishes to dispute (a Disputed Item) including details of:
 - (i) its reasons for disputing each Disputed Item; and
 - the adjustments that it believes need to be made to the Draft Completion Statements in respect of each Disputed Item,

(a Dispute Notice); or

(b) that there is no Disputed Item and that the Buyer does not believe that any adjustments need to be made to the Draft Completion Statements (an **Agreement Notice**).

Finalisation of the Completion Statements where there are no Disputed Items

- 4 If:
 - (a) during the Review Period, the Buyer gives an Agreement Notice to the Seller; and/or
 - (b) by the expiry of the Review Period, no Dispute Notice has been given to the Seller,

the Draft Completion Statements shall constitute the Completion Statements for the purposes of this Agreement, and the date falling on the Business Day immediately after: (i) expiry of the Review Period; or (ii) if earlier, receipt by the Seller of an Agreement Notice, shall be the date of finalisation of the Completion Statements for all purposes of this Agreement.

Finalisation of the Completion Statements where there are Disputed Items

- 5 If the Buyer gives the Seller a Dispute Notice within the Review Period:
 - except for any Disputed Item, the Buyer shall be deemed to have agreed all other items in the Draft Completion Statements;
 - (b) the Seller and the Buyer shall discuss any Disputed Item and endeavour to agree in writing the adjustments (if any) required to be made to the Draft Completion Statements in respect of each such Disputed Item;
 - (c) if, within 20 Business Days after the end of the Review Period (the Resolution Period), the Seller and the Buyer agree in writing upon all adjustments required to be made to the Draft Completion Statements in respect of each Disputed Item (or agree that no such adjustments are required), the Draft Completion Statements as so agreed in writing between the Seller and the Buyer shall constitute the Completion Statements for the purposes of this Agreement, and the date on which the Seller and the Buyer agree in writing on the last of the adjustments required to be made to the Draft Completion Statements in respect of each Disputed Item (or agree that no such adjustments or no further adjustments are required) shall be the date of finalisation of the Completion Statements for all purposes of this Agreement; and
 - (d) if, by the expiry of the Resolution Period, there remains any Disputed Item in respect of which the Seller and the Buyer have not agreed in writing what adjustments are required to be made to the Draft Completion Statements (a **Remaining Item**), then:
 - (i) each such Remaining Item (but no other matters) shall, upon written notice from either the Buyer or the Seller to the other (a Referral Notice), be referred to the Independent Accountants who shall determine, in accordance with paragraphs 7 to 9 (inclusive) below, what adjustments (if any) are required to be made to the Draft Completion Statements in respect of each such Remaining Item;
 - (ii) the Draft Completion Statements adjusted:
 - (A) in respect of any Disputed Item, as agreed in writing between the Seller and the Buyer in accordance with paragraph 5(b) above (if, and to the extent that, the Seller and the Buyer agree any such adjustments); and
 - (B) in respect of any Remaining Item, as determined by the Independent Accountants in accordance with paragraph 5(d)(i) above (if, and to the extent

that, the Independent Accountants determine that any such adjustments are required),

shall constitute the Completion Statements for the purposes of this Agreement; and

(iii) the date on which the Independent Accountants determine the last of the adjustments required to be made to the Draft Completion Statements in respect of each Remaining Item (or determine that no such adjustments or no further adjustments are required) in accordance with paragraph 5(d)(i) shall be the date of finalisation of the Completion Statements for all purposes of this Agreement.

Finalised Completion Statements to be binding

When the Completion Statements have been finalised in accordance with the preceding paragraphs of this Schedule, then the Completion Statements shall be final and binding on the Parties and shall not be subject to question on any ground whatsoever.

Independent Accountants

- If and whenever any matter falls to be referred, in accordance with this Schedule, to Independent Accountants for determination, it shall be referred to such firm of independent chartered accountants: (i) as the Seller and the Buyer may agree in writing within 10 Business Days after the service of a Referral Notice; or (ii) failing such agreement, as the President for the time being of the Institute of Chartered Accountants in England and Wales shall nominate on the application of either the Seller or the Buyer (the Independent Accountants).
- 8 The Independent Accountants shall act on the following basis:
 - the Independent Accountants shall be instructed to notify the Seller and the Buyer of their determination of any such matter within 20 Business Days of such referral;
 - (b) the Independent Accountants shall be instructed to determine any dispute by reference to the accounting treatment specified in paragraph 2 of this Schedule and, in making any such determination, the Independent Accountants shall have exclusive jurisdiction to determine the proper construction of paragraph 2;
 - (c) the Seller and the Buyer shall be entitled to make written submissions to the Independent Accountants, but subject thereto the Independent Accountants shall have power, after having consulted with the Buyer and the Seller, to determine the procedure to be followed in relation to their determination;

- (d) any submissions to and the determination of the Independent Accountants shall be in the English language and any oral hearings shall be conducted in English;
- in making such submissions, the Seller and the Buyer shall state their respective best estimates of monetary amounts of the matters referred for determination;
- (f) in making their determination, the Independent Accountants shall act as experts and not as arbitrators, their decision as to any matter referred to them for determination shall be final and binding in all respects on the Parties, save in the case of manifest error or fraud;
- (g) no amendment may be made by the Independent Accountants to any items or amounts which are not Disputed Items; and
- (h) the fees, costs and expenses of the Independent Accountants shall be borne and paid as the Independent Accountants shall direct and, failing such direction, by the Seller and the Buyer equally.
- The Seller and the Buyer shall enter into an appropriate form of appointment of the Independent Accountants as soon as reasonably practicable (and in any event within 10 Business Days) following the selection of the Independent Accountants in accordance with paragraph 7 above and the Buyer and the Seller shall act reasonably in agreeing the terms and conditions of such appointment, including, without limitation, in respect of fees and any exclusions and limitations of liability where it can be reasonably demonstrated that such terms and conditions reflect market standard provisions for such appointments. In the event of any failure to agree such terms and conditions, the Seller shall be entitled to appoint the Independent Accountants on behalf of both the Buyer and the Seller and to agree such terms and conditions with the Independent Accountants as it shall determine, provided that it can be reasonably demonstrated that such terms and conditions broadly reflect market standard provisions for such appointments.

Provision of Information

- 10 Until the Completion Statements have been finalised in accordance with this Schedule:
 - (a) the Seller shall, and shall procure that each member of the Seller's Group shall, provide the Buyer and its professional advisers with access to employees of the Seller and other members of the Seller's Group and all information, assistance and access to books and records of account, documents, files, papers and information stored electronically which the Buyer may reasonably require for the purposes of this Schedule and permit the Buyer (at the Buyer's cost) to take copies thereof;

- (b) the Buyer shall, and shall procure that other members of the Buyer's Group shall, provide the Seller and its professional advisers with access to employees of the Buyer and other members of the Buyer's Group (including the Target Companies) and all information, assistance and access to books and records of account, documents, files, papers and information stored electronically which the Seller may reasonably require for the purposes of this Schedule and permit the Seller (at the Seller's cost) to take copies thereof; and
- (c) in the event that Independent Accountants are appointed, the Seller and the Buyer shall each provide, and shall procure that the members of the Seller's Group or the Buyer's Group (as applicable) shall provide, the Independent Accountants promptly with access to employees of the Seller (and other members of the Seller's Group) and the Buyer (and other members of the Buyer's Group) and all information, assistance and access to books and records of account, documents, files, papers and information stored electronically which they may reasonably require for the purposes of this Schedule, and the Independent Accountants shall be entitled (to the extent they consider it appropriate) to base their determination on such information and on the accounting and other records of the Target Companies.

Costs of respective accountants and other expenses

- 11 The fees, costs and expenses of any accountants (other than the Independent Accountants) used by the Buyer or the Seller in respect of the matters dealt with by this Schedule shall be for the account of the Buyer and the Seller respectively.
- 12 Each of the Buyer and the Seller shall bear its own fees, costs and expenses with respect to the finalisation of the Completion Statements.

Part B - Specific policies and accounting treatments to be used in preparing the Completion Statements

- Except as otherwise expressly specified, no account shall be taken of events taking place after Completion, and regard shall only be had to information available to the Parties up to the date that the Draft Completion Statements are delivered by the Seller to the Buyer.
- 2 For the purposes of the Completion Statements, Completion shall be treated as the end of a financial accounting and tax period.
- 3 The Completion Statements will be prepared in GBP. Assets and liabilities in the Completion Statements denominated in a currency other than GBP shall be converted at the Conversion Rate on the Completion Date.
- 4 The provisions of this Schedule shall be interpreted so as to avoid double counting (whether positive or negative) of any item to be included in the Completion Statements.
- The Completion Statements shall be prepared by reference to draft completion balance sheets at the Completion Date for the Target Companies, in substantially the same format as the Carve-Out Accounts (but including separate line items as necessary for the Completion Statements and to facilitate the calculation of any adjustments required pursuant to clauses 4.9 to 4.11 (such that the Parties to calculate Actual Net Indebtedness and Actual Working Capital in respect of each Target Company)) and on a consolidated basis, drawn up from the individual general ledgers of each of the Target Companies (including by reference to individual operating sites therein).
- Trade debtors, trade creditors, VAT and accruals (other than holiday pay accruals for direct labour) in respect of the Target Companies (other than Oakhouse and Rollover) shall remain with the Seller's Group where provided for in accordance with this Agreement.
- 7 For the avoidance of doubt, and save as set out as express inclusions in the definition of Indebtedness, other than in relation to Oakhouse and Rollover, the Working Capital Statement shall exclude all trade debtors, other debtors, trade creditors, and other creditors and accruals.
- 8 Prepayments in respect of all Target Companies shall remain with the Target Companies.
 Prepayment balances shall be included within the Working Capital Statement for all Target Companies.
- 9 In relation to the Target Companies other than Oakhouse and Rollover, trade debtors and trade creditors are stated net of any credit or debit balances payable on such trade debtors or creditors as the case may be. Trade debtors shall include VAT receivables from HMRC or the Irish Revenue

Commissioners and trade creditors shall include VAT payables to HMRC or the Irish Revenue Commissioners and capital purchase creditors.

- 10 In relation to Oakhouse and Rollover, trade debtors and trade creditors within the Working Capital Statement are stated net of any credit or debit balances payable on such trade debtors or creditors as the case may be. Trade debtors shall include VAT receivables from HMRC or the Irish Revenue Commissioners and trade creditors shall include VAT payables to HMRC or the Irish Revenue Commissioners and capital purchase creditors.
- 11 The Working Capital Statement shall include full provision for holiday pay accruals for all Target Companies for direct labour, calculated as at the Completion Date in line with historical practice.
- 12 In relation to the Target Companies other than Oakhouse and Rollover, in respect of GRN (goods received notes) accruals, accruals (other than holiday pay accruals for direct labour), which for the avoidance of doubt in respect of such Target Companies shall remain with the Seller's Group, are made for all goods and services received but not yet invoiced at the Completion Date, based on the quantity of goods received multiplied by the order price, including costs of freight and any other transportation costs and duties associated with these goods and services.
- 13 In relation to Oakhouse and Rollover, in respect of GRN (goods received notes) accruals, accruals within the Working Capital Statement shall be made for all goods and services received but not yet invoiced at the Completion Date, based on the quantity of goods received multiplied by the order price, including costs of freight and any other transportation costs and duties associated with these goods and services.
- 14 Lease liability amounts for any capitalised right of use leased assets are as calculated by Kerry's SAP REEX module
- Fixed asset, leased and non-leased, capital additions between the date of this Agreement and the Completion Date shall be determined by the capitalisation dates on the fixed asset registers.
- 16 The Completion Statements shall include £800,000 for lease dilapidations.
- 17 Deposits and sundry receivables will be included in the Completion Statements at their recoverable amount calculated in accordance with IFRS, consistent with historical practice.
- The value of Stock shall be such amount as calculated by the Seller in accordance with clause 4.4 of this Agreement. Any third party logistics providers holding Stock will be requested to issue stock confirmations to the Seller for the purposes of the Stock Take.
- 19 For the purposes of the valuation of the Stock:

- (a) all raw materials and packaging materials stock shall be valued at the lower of weighted average cost and net realisable value consistent with historical practice;
- (b) work in progress shall be valued at the latest production cost in line with historical practice (raw materials, packaging);
- (c) finished goods shall be valued at the lower of the latest actual monthly production cost (raw materials, packaging and direct labour) and attributable overheads in line with historical practice and net realisable value;
- (d) all raw materials, packaging, work in progress and finished goods will be reviewed to determine obsolescence. Obsolescence arises as follows:
 - stock items beyond their "use by / use before" date or which a customer will not accept on the basis of the remaining level of life in the product shall be considered obsolete;
 - (ii) stock items that have been contaminated shall be considered obsolete; and
 - (iii) work in progress stock which will not be converted to finished product within its shelflife shall be considered obsolete,

in each case consistent with historical practice;

- appropriate provision shall be made for obsolete, slow moving, defective or damaged raw materials, packaging, work in progress and finished goods, consistent with historical practice;
- (f) any amount required to discharge any retention of title (or other similar or analogous right) against the Stock shall be deducted from the value of the Stock, consistent with historical practice;
- (g) engineering spares and consumables shall be valued at cost with an appropriate provision for obsolete engineering spares consistent with historical practice.
- 20 If any insured event occurs after the date of this Agreement but before Completion in relation to any asset of a Target Company or the Business which needs to be replaced or restored in order for the relevant business to continue to be conducted in the ordinary course, then, to the extent that a member of the Seller's Group recovers any proceeds or is entitled to a receivable under a policy but the relevant asset is not replaced or restored before Completion, any such proceeds

shall be deducted from Cash and any such receivable shall not be included in Working Capital and, accordingly, shall not, in each case, be included in the Completion Statements.

- 21 The calculation of tax assets/liabilities for the purposes of the Net Debt Statement and Working Capital Statement will be undertaken in accordance with the following provisions:
 - (a) provisions for all current and non-current Taxes payable or liable to be paid, but unpaid, by any Target Company in respect of the period up to the Completion Date shall be calculated by reference to income, profit, or gains or losses earned, accrued, received or realised, remuneration paid and supplies made or received, as the case may be, in the period up to and including the Completion Date and income, profits, or gains or losses deemed to have been earned, accrued, received or realised, remuneration deemed to be paid and supplies deemed to be made or received, as the case may be, for Tax purposes in the period up to and including the Completion Date;
 - (b) the Net Debt Statement and Working Capital Statement shall include all liabilities to Tax of each Target Company arising by reason of that or any other Target Company leaving the Seller's Group for any Tax purpose or as a result of any deemed disposal or realisation by any Target Company of any asset or liability for any Tax purposes as a result of Completion (including, without limitation, any stamp duty land tax liability arising as a result of the withdrawal of any relief claimed in connection with the Hive Down);
 - (c) the Working Capital Statement shall include provision for all payroll taxes (including social security contributions) and (in accordance with paragraphs 5 to 7 above) VAT payables owed by, or any VAT receivables owed to, any Target Company in respect of any period up to and including the Completion Date;
 - (d) the Net Debt Statement shall include provision for all other Tax or amounts in respect of Tax (including, without limitation, corporation tax but excluding payroll taxes, social security contributions and VAT) payable by any Target Company in respect of any period up to and including the Completion Date;
 - (e) a provision for all Tax or amounts in respect of Tax (including, without limitation, corporation tax and/or payroll taxes and/or social security contributions) payable by any Target Company in respect of the period up to and including the Completion Date shall be included in the Net Debt Statement and/or the Working Capital Statement;
 - (f) for the purposes only of determining whether Taxes are payable or liable to be paid in respect of the period up to and including the Completion Date and whether a Relief has arisen in or relates to the period up to and including the Completion Date, an accounting

period of each Target Company shall be deemed to have ended on the Completion Date;

- (g) for the avoidance of doubt, no liability and no asset shall be included in the Completion Statements in respect of deferred taxation.
- 22 The Completion Statements shall include no amounts in respect of contingent liabilities (as defined in accordance with IFRS).
- 23 The Completion Statements shall include no amounts in relation to Excluded Assets, Transaction Costs, Excluded Contracts or Excluded Liabilities.

Part C - Form of Net Debt Statement

[***]

Part D - Form of Working Capital Statement

[***]

Schedule 10 Agency provisions

Part A: Agency structure

- The Seller is entering into this Agreement on the basis that any undertaking or agreement given by the Seller under this Agreement (including any Seller Obligation) is given, and any undertaking or agreement given by the Buyer under this Agreement (including any Buyer Obligation) is received, by the Seller as principal and, to the extent that the relevant undertaking or agreement relates to a particular Set of Sale Shares, as agent for the Designated Seller which is selling that particular Set of Sale Shares.
- 2 The Buyer is entering into this Agreement on the basis that any undertaking or agreement given by the Buyer under this Agreement (including any Buyer Obligation) is given, and any undertaking or agreement given by the Seller under this Agreement (including any Seller Obligation) is received, by the Buyer as principal and, to the extent that the relevant undertaking or agreement relates to a particular Set of Sale Shares, as agent for the Designated Buyer which is acquiring that particular Set of Sale Shares.
- To the extent that any payment is made by the Buyer to the Seller under this Agreement in respect of an Initial Purchase Price or by way of adjustment to the price of a particular Set of Sale Shares (including in respect of any Buyer Obligation), such payment is (i) made by the Buyer as agent for the Designated Buyer acquiring the relevant Set of Sale Shares (unless the Buyer is itself the relevant Designated Buyer, in which case it is made by the Buyer as principal) and (ii) received by the Seller as agent for the Designated Seller selling the relevant Set of Sale Shares (unless the Seller is itself the relevant Designated Seller, in which case it is received by the Seller as principal).
- To the extent that any payment is made by the Seller to the Buyer under this Agreement by way of or adjustment to the price of a particular Set of Sale Shares (including in respect of any Seller Obligation), such payment is (i) made by the Seller as agent for the Designated Seller selling the relevant Sale Shares under this Agreement (unless the Seller is itself the relevant Designated Seller, in which case it is made by the Seller as principal) and (ii) received by the Buyer as agent for the Designated Buyer acquiring the relevant Sale Shares under this Agreement (unless the Buyer is itself the relevant Designated Buyer, in which case it is received by the Buyer as principal).
- For the purposes of this Schedule, an undertaking or agreement or any Buyer Obligation or Seller Obligation relates to a particular Set of Sale Shares if the relevant provision relates to or applies

in respect of that Set of Sale Shares regardless of whether or not the relevant provision refers to it or them by name or applies in respect of more than one Set of Sale Shares.

If any undertaking under this Agreement is expressed to be an obligation of more than one Designated Seller or more than one Designated Buyer, it shall be construed as a separate obligation of each such Designated Seller or Designated Buyer (as the case may be) in relation only to the Sale Shares which that Designated Seller is selling or which that Designated Buyer is purchasing under this Agreement, and the liability of each such Designated Seller or Designated Buyer (as the case may be) shall be several and not joint.

Part B: Appointment of Buyer as agent for UK Designated Buyer and Irish Designated Buyer

- 1 Each of the UK Designated Buyer and the Irish Designated Buyer undertakes with the Buyer to comply with its obligations as a Designated Buyer under the this Agreement.
- Each of the UK Designated Buyer and the Irish Designated Buyer hereby irrevocably appoints and instructs the Buyer as its sole agent:
 - to enter into and perform this Agreement as its agent on the terms set out in this Schedule and to enter into and perform any other Transaction Documents;
 - (b) with the exception of any amounts payable by that Designated Buyer pursuant to clause 4, to receive or pay (as the case may be) any amounts owed to or paid or to be paid by or to that Designated Buyer pursuant to any of the provisions of this Agreement;
 - to initiate, receive, conduct and settle any claims and satisfy all judgments in respect of all matters arising under this Agreement; and
 - (d) to give and receive such representations, warranties, undertakings and indemnities, and assume such other rights and obligations as this Agreement provides are to be given, assumed, received, made or done by the Buyer on behalf of or as agent for that Designated Buyer (including in respect of the Warranties, the Tax Covenant and the indemnities given by the Seller and the Buyer as set out in this Agreement)

and the Buyer hereby accepts such appointment and instruction.

- 3 Each of the UK Designated Buyer and the Irish Designated Buyer undertakes to the Buyer:
 - (a) to provide the Buyer with funds to pay any amounts to be paid by that Designated Buyer pursuant to any of the provisions of this Agreement; and

- (b) to indemnify the Buyer against any and all Losses suffered or incurred by the Buyer in relation to any claim under this Agreement paid by the Buyer as agent of that Designated Buyer.
- Each of the UK Designated Buyer and Irish Designated Buyer agrees with the Buyer that the rights and benefits held and received (including the Warranties, the Tax Covenant and the indemnities) by that Designated Buyer through the Buyer's agency under this Agreement shall only be enforceable by that Designated Buyer through such agency and upon the terms of this Agreement. Any liability in respect of any breach of such rights and benefits (including in respect of the Warranties, the Tax Covenant and any indemnities) shall be determined solely in accordance with the terms of this Agreement.

Part C: Supplemental Transaction Agency Agreements

- The Seller and the Buyer agree that the Buyer may add or substitute a Designated Buyer in respect of any particular Set of Sale Shares or Sets of Sale Shares by Supplemental Transaction Agency Agreements (in substantially the same form as the Standard Transaction Agency Agreement) entered into after the date of this Agreement and prior to Completion, provided that:

 (a) any such additional or substitute Designated Buyer is and will remain a member of the Buyer's Group at Completion and (b) to the extent not incorporated in the UK or ROI, the identity of such additional or substitute Designated Buyer shall be subject to approval by the Seller (such approval not to be unreasonably withheld or delayed and provided that no such approval shall be required for the Buyer to be a Designated Buyer). Any such additional or substitute Designated Buyer (together with the Buyer, the Seller and each of the other Designated Sellers and Designated Buyer which entered into Standard Transaction Agency Agreements on or before the date of this Agreement and have not been so substituted) shall have such rights against, and obligations to, each other under this Agreement as they would have had if:
 - (a) such additional or substitute Designated Buyer had been specified as a Designated Buyer of the relevant Set of Sale Shares in Schedule 1 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price) of this Agreement; and
 - (b) the Buyer has been validly appointed as the agent of such additional or substitute Designated Buyer pursuant to a Standard Transaction Agency Agreement on or before the date of this Agreement,

and this Agreement shall be read and construed accordingly (including the terms "Designated Buyer" and "agent").

- 2 The Buyer undertakes that, if the relevant entity which is to purchase Sale Shares under this Agreement does not exist and/or has not entered into a Standard Transaction Agency Agreement at the date of this Agreement, such entity will as soon as practicable (and in any case before Completion) enter into a Supplemental Transaction Agency Agreement (and thereby become a Designated Seller or, as the case may be, a Designated Buyer if it is not one already).
- Each of the Seller and the Buyer shall negotiate in good faith with a view to agreeing before the Completion Date the final form of any Supplemental Transaction Agency Agreement pursuant to this Part B. Such document(s) shall be in all material respects in the agreed form of the supplemental agency agreement template (except solely to the extent that it is necessary to modify it to comply with the laws and regulations of the jurisdiction in which the relevant transfer of Sale Shares is taking place or to reflect the formalities for execution in its jurisdiction of incorporation by any entity which is to be party to such document).

Part D: Claims

- If the Buyer or any of its Related Undertakings wishes to make a claim under this Agreement against the Seller or any of its Related Undertakings, the Buyer shall procure that any such claim shall only be made by the Buyer (for itself and/or as agent for its relevant Related Undertakings) against the Seller (for itself and/or as agent for its relevant Related Undertakings) in accordance with the dispute resolution provisions in this Agreement. A party who is a Designated Buyer who ceases to be a Related Undertaking of the Buyer shall cease to have any rights under, and shall not be entitled to bring any claim in respect of, this Agreement.
- 2 If the Seller or its Related Undertakings wishes to make a claim under this Agreement against the Buyer or any of its Related Undertakings, the Seller shall procure that any such claim shall only be made by the Seller (for itself and/or as agent for its relevant Related Undertakings) against the Buyer (for itself and/or as agent for its relevant Related Undertakings) in accordance with the dispute resolution provisions in this Agreement. A party who is a Designated Seller who ceases to be a Related Undertaking of the Seller shall cease to have any rights under, and shall not be entitled to bring any claim in respect of, this Agreement.
- For the purposes of this Part C of this Schedule, whichever of the Seller or the Buyer is bringing a claim (whether as agent or otherwise) in accordance with paragraph 1 or paragraph 2 of this Part D, as the case may be, is termed the Claimant Party and the other of them is termed the Defendant Party. The parties agree that:
 - (a) the Defendant Party shall not raise any defence or objection to any such claim on the basis that it is made in the name of the Claimant Party acting as agent and/or made against the Defendant Party acting as agent pursuant to the provisions of this Schedule and shall be

- deemed to have waived the right to raise and to be stopped from raising any such defence or objection;
- (b) where a claim cannot, as a matter of law, be made by the Claimant Party in its own name as agent for a relevant Related Undertaking pursuant to this Schedule, any such claim may and shall be assigned by the Related Undertaking to the Claimant Party (provided that the liability of the person claimed against shall be no greater and no less than such liability would have been if the assignment had not occurred);
- (c) the Claimant Party undertakes to indemnify the Defendant Party and its Related Undertakings against any and all Losses suffered or incurred by any of them as a result of a Related Undertaking of the Claimant Party making any claim other than in accordance with Part C of this Schedule and the Claimant Party shall procure that any such claim made by a Related Undertaking of the Claimant Party is discontinued and withdrawn with immediate effect;
- (d) where a claim is made by the Claimant Party against the Defendant Party and the claim results in a payment being required to be made to the Claimant Party, the payment shall be made by the Defendant Party (as principal and/or, if applicable, as agent for its relevant Related Undertakings to the Claimant Party (as principal and/or, if applicable, as agent for its relevant Related Undertakings).

Schedule 11

Definitions and Interpretation

1 In this Agreement:

Accounts means the annual accounts of each Existing Company, including the notes to those accounts and the associated directors' and auditor's reports, for the financial year ended on the Accounts Date;

Accounting Standards means the accounting standards in force and applicable to the relevant Target Company as at the Accounts Date;

Accounts Date means 31 December 2020;

Actual Net Indebtedness means the Indebtedness less the Cash of the Target Companies at Completion, as ascertained after Completion in accordance with Schedule 9 (*Completion Statements*) and as set out in the Completion Statements;

Actual Working Capital means the aggregate Working Capital of the Target Companies at Completion, as ascertained after Completion in accordance with Schedule 9 (*Completion Statements*) and as set out in the Completion Statements;

Agency Agreements means the undisclosed agency agreements entered into between Kerry Foods Limited or Henry Denny & Sons (Ireland) Limited (as agent) and the Target Companies (as manufacturer);

Apportionable Items means any Customer Agreements and Supplier Agreements which contain provisions requiring apportionment over a period of time commencing prior to Completion (including promotions, rebates and overrider payments);

Approvals has the meaning given to it in paragraph 5.2 of Schedule 4 (Seller Warranties);

Asbestos Indemnity Claim means any claim under the environmental indemnity at clause 20.2(a) and 20.2(b);

Asbestos Reports means the two asbestos management survey reports dated 29 January 2020 and 13 January 2020 in respect of the main factory building and main administration building at the Hyde Property, prepared by Bureau Veritas UK, copies of which are included in the Data Room with reference numbers 1.3.2.2.14 and 1.3.2.2.15 respectively;

Associated Person means, in relation to a company, a person (including any director, manager, officer, secretary, employee, agent or other intermediary or subsidiary) who performs services for or on behalf of that company or who holds shares of capital stock, partnership interests, limited liability company membership interests and units, shares, interest and other participations in that company (in each case when performing such services or acting in such capacity);

Authorisations means all material regulatory or environmental licences, registrations, consents, permits (including Environmental Permits) certifications, approvals and other material authorisations (public and private) that are necessary for conducting the Business;

Brand Licences means the Galtee Licence and Everwell Licence;

Burton Property means the property at Century Court, First Avenue, Centrum 100, Burton on Trent, to the extent used prior to the date of this Agreement in relation to the Business;

Business means:

- the manufacture, processing, marketing, supply, distribution or sale of branded and private label chilled, frozen and ready to cook ready meals for sale into the consumer foods retail market and the direct to consumer market (the Ready Meals Business);
- (b) the manufacture, processing, marketing, supply, distribution or sale or any other form of commercialisation of branded or private label: (i) sausages and bacon; (ii) cooked meats; (iii) meat snacks; and (iv) meat alternative snacks, sausages, bacon, burgers and mince, in each case for sale into the consumer foods retail market (the Meats Business);
- the vans sales trading operations and services in Northern Ireland and ROI (the Van Sales Business); and
- (d) the business operating under the name 'Rollover' which supplies equipment and a range of hot food and snack options to business customers,

in each case as carried out by the Seller's Group as at Completion;

Business Day means a day other than a Saturday or Sunday on which banks are ordinarily open for the transaction of normal banking business in London and Dublin;

Business Payables means all amounts payable or owing by a member of the Seller's Group (other than Oakhouse and Rollover) for goods and/or services supplied to the Business or otherwise owing by a member of the Seller's Group (other than Oakhouse and Rollover) in respect of the Business, in each case prior to the Completion Date (whether or not invoiced);

Business Receivables means all amounts receivable by or owing to a member of the Seller's Group (other than Oakhouse and Rollover) for goods and/or services supplied by the Business

or otherwise owing to a member of the Seller's Group (other than Oakhouse and Rollover) in respect of the Business, in each case prior to the Completion Date (whether or not invoiced);

Buyer Obligation means any representation, warranty, covenant to pay or undertaking to indemnify given by the Buyer to the Seller under this Agreement or the Tax Covenant;

Buyer Warranties means the warranties set out in Schedule 5 (*Buyer Warranties*) and **Buyer Warranty** means any one of them;

Buyer's Assignee has the meaning given to it in clause 35.2(b);

Buyer's Bank Account means the account of the Buyer as notified to the Seller in writing from time to time;

Buyer's Group means the Buyer and each company which is for the time being (whether on or after the date of this Agreement) a Related Undertaking of the Buyer, including with effect from Completion the Target Companies;

CA 2006 means the UK Companies Act 2006;

CA 2014 means the ROI Companies Act 2014;

Carrickmacross Site means the site and facilities operated by a Target Company at Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan, Ireland which are shown outlined in red on Map 1 annexed to the Declaration of Identity of Peadar Stack dated 31 May 2021 located at 1.8.2.4.1 of the Data Room and the Occupied Real Estate set out at item (i) of paragraph 10.11 of Schedule 4

Carve-Out Accounts means the pro-forma income statement and pro-forma balance sheet of Business for the three consecutive financial years ended on the Accounts Date as set out in the Deloitte VFDD Report;

Cash means the aggregate of the cash (whether in hand or credited to any account with any banking, financial, acceptance credit, lending or other similar institution or organisation) and cash equivalents, including all interest accrued thereon, of the Target Companies as at Completion, but excluding, for the avoidance of doubt, any item or amount to the extent it is taken into account in calculating the Actual Working Capital;

CCPC means the ROI Competition and Consumer Protection Commission;

Central Services Employees means [***];

CMA has the meaning given to it in clause 3.1(a);

Commitments has the meaning given to it in clause 3.4(e);

Competition Act means the ROI Competition Acts 2002 to 2017;

Competition Authority has the meaning given to it in clause 3.4(a);

Completion means completion of the sale and purchase of the Sale Shares by the performance by the Parties of their respective obligations under clause 11 and Schedule 8;

Completion Statements means the Net Debt Statement and the Working Capital Statement;

Completion Date means the date on which Completion occurs in accordance with this Agreement:

Conditions has the meaning given to it in clause 3.1;

Confidential Information means trade secrets and information equivalent to them (including formulae, processes, methods, knowledge and Know-how) in connection with the Business or relating to the products manufactured, produced, distributed or sold and/or the services supplied by the Seller's Group in connection with the Business or the customers and suppliers of the Business and which are for the time being confidential to any member of the Seller's Group;

Continuing Related Party Agreements means the Meat Free Substitute Technology Licence and the Brand Licences the Van Sales Distribution Agreement and the Raw Material Supply Agreement;

Conversion Rate means with respect to a particular currency for a particular day the spot rate of exchange (the closing mid-point) for that currency into sterling on such date as published in the London edition of the Financial Times first published thereafter or, where no such rate is published in respect of that currency for such date, at the rate quoted Barclays Bank plc as at the close of business in London as at such date;

CTA 2010 means the Corporation Tax Act 2010;

Customer Agreements means all agreements, contracts, engagements, arrangements, licences and other commitments (whether in writing or orally) entered into by the Business or by the Seller's Group relating to the Business (including those which the Seller's Group transferred, or purported to transfer, to a Target Company under the Hive Down Agreements) in each case with third party customers, including (without limitation) those contracts set out in the Data Room reference 1.9.2, 1.4.1 (clean team) and 1.4.2 (clean team);

Data Protection Legislation means all laws and regulations applicable to the processing of Personal Data, including:

- (a) Regulation (EU) 2016/679 of the European Parliament and of the Council of 27 April 2016 (General Data Protection Regulation) (GDPR) (including as amended and transposed into the laws of the United Kingdom pursuant to the European Union (Withdrawal) Act 2018 and the European Union (Withdrawal Agreement) Act 2020) and other laws and regulations of the European Union, its member states and the United Kingdom relating to the processing of Personal Data, privacy and interception of communications to the extent applicable in the United Kingdom or Ireland, including the Irish Data Protection Act 2018;
- the corresponding laws and regulations of each jurisdiction in which any Target Company operates; and
- (c) where applicable, the related guidance and codes of practice issued by the United Kingdom's Information Commissioner, the Irish Data Protection Commission or any other regulator of each jurisdiction in which any Target Company operates;

Data Room means all documents, correspondence and other information made available for inspection by the Buyer and its advisers in the electronic data room and clean room, each hosted by Intralinks as the same appeared in respect of the electronic data room known as 'Project Verona' at 14:30pm on 17 June 2021 and in respect of the clean room known as 'Project Verona Clean Room' at 23:56pm on 16 June 2021 and as is listed in the Data Room Index;

Data Room Index means the indexes detailing the contents of the Data Room, in the agreed form:

Data Subject, Personal Data, Personal Data Breach and Supervisory Authority have the meanings given to them in the Data Protection Legislation;

Designated Buyers means the members of the Buyer's Group set out in column (3) of the table contained in Schedule 1 (*Designated Sellers*, *Designated Buyers and Apportionment of Purchase Price*) and/or any member of the Buyer's Group established as a Designated Buyer of any shares pursuant to a Supplemental Transaction Agency Agreement and provided that they shall only remain a Designated Buyer so long as they remain a Related Undertaking of the Buyer, and **Designated Buyer** means any of them;

Designated Sellers means the members of the Seller's Group set out in column (1) of the table contained in Schedule 1 (*Designated Sellers*, *Designated Buyers and Apportionment of Purchase Price*) and/or any member of the Seller's Group established as a Designated Seller of any shares pursuant to a Supplemental Transaction Agency Agreement and provided that they shall only

remain a Designated Seller so long as they remain a Related Undertaking of the Seller, and **Designated Seller** means any of them;

Disclosed means fairly disclosed in such manner and in such detail as to enable the Buyer to make a reasonably informed assessment of the nature, scope, consequences and significance of the matter concerned;

Disclosure Letter means the letter of the same date as this Agreement from the Seller to the Buyer disclosing certain matters in relation to the Seller Warranties, together with all documents attached to it or listed in any schedule to it;

Due Diligence Reports means:

- (a) the Project Verona vendor financial due diligence report prepared by Deloitte LLP, dated
 31 March 2021 (the **Deloitte VFDD Report**);
- the Project Verona tax financial due diligence report prepared by Deloitte LLP, dated 31 March 2021;
- (c) the Project Verona legal due diligence report prepared by Norton Rose Fulbright LLP, dated
 31 March 2021; and
- (d) the Project Verona environmental due diligence reports prepared by ERM Group Inc. in respect of certain Properties (being Enniskillen, Hyde, Attleborough, Shillelagh, Carrickmacross, Windmill Lane, Dean Way, Collett Way, Spurway), together with the final summary report, dated 25 and 29 March 2021;

Employees means the employees of the Target Companies and the Central Services Employees from time to time:

Employment Contracts means the contracts or other terms and conditions of employment under which the Employees are for the time being employed by the Seller or any member of the Seller's Group;

Encumbrance means any mortgage, charge, rent-charge, pledge, lien, option, restriction, right of first refusal, right of pre-emption, claim, right, interest or preference granted to any third party, or any other encumbrance or security interest of any kind (or an agreement or commitment to create any of the same);

Environment means any or all living organisms including the ecological systems of which they form part and the following media (alone or in combination): air (including the air within the

buildings and the air within other natural or man-made structures whether above or below ground); water (including water under or within land or in drains or sewers and surface, ground, coastal and inland waters); land (including surface land, sub-surface strata, land under water and natural and manmade structures); and in the case of man includes his senses and his property;

Environmental Authority means any person (including any Governmental Department or Government Agency or body, the Environment Agency, local authorities or the Health & Safety Executive or their respective equivalents in any jurisdiction in question) having regulatory powers and/or authority at law and/or any court of law or tribunal in relation to the Environment or Environmental Laws:

Environmental Claim means any claim or demand or any civil or administrative litigation, arbitration, dispute resolution proceedings, suit, action, notice or other enforcement process or any voluntary action approved by any Environmental Authority, or any enquiry or investigation by any Environmental Authority, official warning, abatement or other order or notice (conditional or otherwise) relating to the pollution or protection of the Environment, or harm or protection of the health of humans, animals or plants under any Environmental Permits or Environmental Laws;

Environmental Expert shall have the meaning set out in clause 20.15;

Environmental Instrument means any credit, allowance or certificate required to be held for the direct or indirect emission or potential emission of greenhouse gases under a regime established for that purpose by Environmental Laws;

Environmental Laws means all applicable laws (including all or any of statute, common law, rule, regulation, treaty, directive, direction, decision of the court, bye-law, code of practice, circular, guidance note, statutory guidance, order, notice, demand or official guideline of any Environmental Authority) in force in any relevant jurisdiction (including the European Union) at any time up to and including the date of this Agreement to the extent that they relate to or concern the protection of human health or the Environment, energy efficiency, climate change and/or the conditions of the workplace and worker health and safety or they regulate, control or prohibit the generation, use, handling, emission, transportation, storage, treatment or disposal of any substance (including waste) or any noise, vibration, odour, light or radioactivity or have as a purpose or effect the provision of remedies or compensation for harm or damage to the Environment or any loss arising therefrom or the condition, protection, maintenance, remediation, reinstatement, restoration of the Environment or any part of it;

Environmental Permits means any agreement, permission, permit, licence, authorisation, consent, registration, notification, exemption or other approval required pursuant to any

Environmental Laws by any Target Company (including any condition or requirement thereof as modified from time to time) in order to carry out its operations;

EPA means the Environmental Protection Agency of the Republic of Ireland;

EPA Financial Provisions means any secured fund, guarantee, insurance, cash deposit, letter of credit, bond, charge, security or any other provision either individually, or in combination, which is in place and agreed with the EPA for the purposes of covering the cost of any environmental liability that may occur during and after the lifetime of the ROI Sites;

Estimated Net Indebtedness means the Seller's reasonable estimate of Actual Net Indebtedness as notified by the Seller to the Buyer in accordance with clause 4.7;

Estimated Working Capital means the Seller's reasonable estimate of Actual Working Capital as notified by the Seller to the Buyer in accordance with clause 4.7;

Everwell Licence means the licence relating to the 'Everwell' brand from Kerry Luxembourg SARL to such member of the Seller's Group as the Seller shall nominate, to be entered into in the agreed form;

Excluded Assets, **Excluded Contracts** and **Excluded Liabilities** have the meanings given to them the relevant Hive Down Agreement;

Excluded Events means:

- (a) changes in general political, economic, financial, regulatory or market conditions;
- acts of war, hostilities, military activity, terrorism, sanction, embargo or other calamity or crisis;
- epidemics, pandemics (including Covid-19), earthquakes, floods, tsunamis, hurricanes, volcanos, fires, tornadoes or other natural disasters;
- (d) changes in law or regulation;
- (e) acts or omissions of any member of the Seller's Group or any Target Company at the written request or with the written consent of the Buyer; and
- (f) transactions contemplated by this Agreement or any Transaction Document including changes in control resulting from any such transaction, Events which are the subject of an indemnity in this Agreement or a Transaction Document and any matter that is Disclosed;

Existing Companies means the companies specified in Part A of Schedule 2 (*Target Companies*) and **Existing Company** means any of them;

Financial Crime Laws means all applicable laws, rules and regulations of any jurisdiction relating to the prevention of bribery, corruption, money laundering, terrorist financing, facilitation of tax evasion, fraud or similar or related activities or relating to financial sanctions, including the Foreign Corrupt Practices Act 1977 of the USA, the ROI Criminal Justice (Corruption Offences) Act, 2018, the ROI Criminal Justice (Theft and Fraud Offences) Act, 2001, the Terrorism Act 2010, the Proceeds of Crime Act 2002, the Bribery Act 2010, the Money Laundering Regulations 2017 and the Criminal Finances Act 2017, and Financial Crime Law means any of them;

Food Laws means any law or other requirement having legal effect in any jurisdiction to which the Business and/or the Existing Companies are subject, concerning meat, meat production and preparation, meat products and other foodstuffs intended for human consumption including without limitation food safety, hygiene, packaging, information, labelling, advertising, storage, distribution, transportation, importation, exportation, sale and supply of meat, meat products and other foodstuffs;

Fundamental Warranties means those Seller Warranties set out at paragraphs 1, 2, 3. 4.1 to 4.4 and 19 of Schedule 4 and **Fundamental Warranty** means any one of them;

Fundamental Warranty Claim means any claim, demand, action, proceeding or suit by the Buyer under or in connection with any of the Fundamental Warranties;

Galtee IP Assignment Agreements means:

- the three assignment agreements to be entered into between Breeo Brands Limited (as assignor) and Duffy Meats Limited (as assignee);
- the assignment agreement to be entered into between Kerry Luxembourg Sarl (as assignor) and Duffy Meats Limited (as assignee); and
- (c) the assignment agreement between the Seller (as assignor) and Duffy Meats Limited (as assignee),

in each case substantially in the form (subject to insertion of details of the consideration payable under them where applicable) of the execution copies of such agreements disclosed at Data Room reference 1.7.3.7.13 to 1.7.3.7.17;

Galtee Licence means the licence relating to the 'Galtee' brand from Duffy Meats Limited to Breeo Foods Limited to be entered into in the agreed form;

Groundwater Monitoring Scope means the scope and methodology of any groundwater monitoring required pursuant to the Hyde Environmental Permit as identified scoped and agreed between the Seller and the Buyer acting reasonably;

Groundwater Survey means a survey to be completed by an environmental consultant for the characterisation and reporting of the condition of the groundwater, in accordance with the applicable guidance issued by the Environment Agency from time to time, required to the Hyde Environmental Permit and undertaken in accordance with the Groundwater Monitoring Scope;

Groundwater Indemnity Claim means any claim under the environmental indemnity at clause 20.2(c);

Guarantee means any guarantee, indemnity, suretyship, letter of comfort or other assurance, security or right of set-off given or undertaken by a person to secure or support the obligations (actual or contingent) of any other person and whether given directly or by way of counter-indemnity to any other person who has provided a Guarantee;

Headline Price means £680,000,000;

Hive Down means the hive down of certain assets, rights and liabilities in respect of the Business from members of the Seller's Group to the Newly-Trading Companies pursuant to the terms of the Hive Down Agreements;

Hive Down Agreements means:

- the business transfer agreement relating to certain assets, rights and liabilities of the Ready Meals Business between Kerry Foods Limited (as seller) and Attleborough Foods Limited (as buyer), dated 30 May 2021;
- the business transfer agreement relating to certain assets, rights and liabilities of the Meats Business between Kerry Foods Limited (as seller) and Northfield Foods Limited (as buyer), dated 30 May 2021;
- (c) the business transfer agreement relating to the Van Sales Business in Northern Ireland between Kerry Foods Limited (as seller) and Consumer Van Sales Limited (as buyer), dated 30 May 2021;
- (d) the business transfer agreement relating to the Van Sales Business in the Republic of Ireland between Henry Denny & Sons (Ireland) Limited (as seller) and Icehall Limited (now known as Consumer Foods Van Sales (Ireland) Limited) (as buyer), dated 1 May 2020;

- (e) the Hive Down IP Assignment Agreements;
- (f) the Property Transfers; and
- (g) all other agreements entered into in connection with the foregoing

Hive Down IP Assignment Agreements means:

- the assignment agreement between the Seller (as assignor) and Northfield Foods Limited (as assignee), dated 30 May 2021;
- the assignment agreement between Kerry Luxembourg Sarl (as assignor) and Attleborough Foods Limited (as assignee), dated 30 May 2021 and
- the assignment agreement between Kerry Luxembourg Sarl (as assignor) and Northfield Foods Limited (as assignee), dated 30 May 2021;

holding company means a holding company (as defined by section 1159 CA 2006) or a parent undertaking (as defined by section 1162 CA 2006);

HMRC means HM Revenue and Customs;

Hyde Environmental Permit means the environmental permit issued to Kerry Foods Limited dated 29 October 2018 permit number EPR/BO5357IE in relation to activities carried out at the Hyde Property;

Hyde Property means the site and facilities operated by the Target Company situated at Godley Hill Road, Godley, Hyde, Cheshire SK14 3BR;

IED Licences means EPA Licence P0806-01 (in respect of the ROI Site at Carrickmacross) and EPA Licence P0804-02 (in respect of the ROI Site at Shillelagh), each as amended or replaced, and granted by the EPA under the Republic of Ireland Environmental Protection Agency Act 1992 (as amended) for the carrying out of certain industrial and agricultural activities;

IFRS means, the body of pronouncements issued by the International Accounting Standards Board (IASB), as adopted for use in the European Union further to the IAS regulation (EC 1606/2002), including International Financial Reporting Standards and interpretations approved by the IASB, International Accounting Standards and Standing Interpretations Committee interpretations approved by the predecessor International Accounting Standards Committee;

Indebtedness means the aggregate indebtedness of the Target Companies, including all declared but unpaid dividends, borrowings and other indebtedness by way of overdraft, acceptance credit or similar facilities, loan stocks, bonds, debentures, notes, debt or inventory financing, finance leases or sale and lease back arrangements or any other arrangements the purpose of which is to borrow money, together with forex, interest rate or other swaps, hedging obligations, bills of exchange, recourse obligations on factored debts and obligations under other derivative instruments, being for these purposes only those line items set out in Part C of Schedule 9 (Completion Statements) but excluding, for the avoidance of doubt, any item or amount to the extent it is taken into account in calculating the Actual Working Capital;

Indemnity Claim means any indemnity claim made by the Buyer pursuant to clauses 5.1, 5.2(b), 6.1(b), 12, 13 and 20;

Independent Accountants has the meaning given to it in paragraph 6 of Schedule 13;

Information Memorandum means the Project Verona confidential information memorandum prepared by Bank of America Corporation, dated February 2021;

Information Technology means information and communications technology infrastructure (including, without limitation, hardware, software, (in both object and source code) firmware, networks and connecting media);

Initial Purchase Price means an amount equal to:

- (a) the Headline Price;
- (b) minus the Estimated Net Indebtedness;
- (c) minus an amount equal to the difference between the Estimated Working Capital and the Target Working Capital (if the Estimated Working Capital is less than the Target Working Capital) or adding an amount equal to the difference between the Estimated Working Capital and the Target Working Capital (if the Estimated Working Capital is greater than the Target Working Capital);

Intellectual Property Rights means copyrights, (including, without limitation, rights in computer software), patents, trade marks, trade names, service marks, business names (including, without limitation, internet domain names), design rights, database rights, semi-conductor topography rights, rights in undisclosed or confidential information (such as Know-how, trade secrets and inventions (whether patentable or not)), and all other intellectual property or similar proprietary rights of whatever nature (whether registered or not and including, without limitation, applications

to register or rights to apply for registration) which may now or in the future subsist anywhere in the world;

Inter-Company Balances means Inter-Company Payables and Inter-Company Receivables;

Inter-Company Payables means amounts payable by the Business to the Seller's Group (other than the Target Companies);

Inter-Company Receivables means amounts receivable by the Business from the Seller's Group (other than the Target Companies);

IP Assignment Agreements means:

- (a) the four assignment agreements between Kerry Luxembourg SARL (as Assignor) and Duffy Meats Limited (as Assignee) dated 11 June 2021;
- (b) the assignment agreement between Kerry Luxembourg SARL (as Assignor) and Noon Products Limited (as Assignee) dated 11 June 2021;
- (c) the assignment agreement between Kerry Luxembourg SARL (as Assignor) and Oakhouse
 Foods Limited (as Assignee) dated 11 June 2021;
- (d) the assignment agreement between Kerry Luxembourg SARL (as Assignor) and Rollover
 Limited (as Assignee) dated 11 June 2021;
- the assignment agreement between Kerry Luxembourg SARL (as Assignor) and Northfield
 Foods Limited (as Assignee) dated 11 June 2021;
- (f) the assignment agreement between Kerry Luxembourg SARL (as Assignor) and Rye Valley Foods Limited (as Assignee) dated 11 June 2021;
- (g) the assignment agreement between Kerry Group plc (as Assignor) and Duffy Meats Limited (as Assignee) dated 11 June 2021;
- the assignment agreement between Kerry Group plc (as Assignor) and Noon Products Limited (as Assignee) dated 11 June 2021; and
- (i) the Galtee IP Assignment Agreements;

Irish Sale Shares means the entire issued share capital of each of the Target Companies that is incorporated in ROI;

IT and Data means:

- (a) the Information Technology; and
- (b) data (both Personal Data and other data);

Know-how means all information not publicly known, owned by the Seller or any member of the Seller's Group and exclusively used, or required to be used, in, or in connection with, the Business, held in any form (including that comprised in or derived from design drawings, prototypes, models, discoveries, improvements, data, formulae, specifications, component lists, instructions, manufacturing technology manuals, brochures, catalogues, processes, process descriptions and all other technical information and materials);

Leased Real Estate means the leasehold property described in Part B of Schedule 3 (*Properties*);

Longstop Date means 23:59 (London time) on the date falling 12 months after the date of this Agreement;

Losses means all claims, demands, liabilities, damages, compensation, tribunal awards or orders, fines, losses, costs (including legal costs and experts' and consultants' fees) and expenses;

Material Adverse Change means any food safety, food hygiene or food contamination event, circumstance or occurrence or any combination of them which arises or occurs after the date of this Agreement (whether or not foreseeable on or before the date of this Agreement) (an **Event**) which:

- (a) is not an Excluded Event; and
- (b) which (i) has resulted in one or more manufacturing or processing sites or facilities operated by the Target Companies in respect of the Ready Meals Business and/or the Meats Business (excluding sites or facilities relating to Oakhouse and Rollover) substantially ceasing to operate for at least 30 consecutive days; and (ii) has, or is reasonably likely to have a material adverse effect on the business, operations, assets, liabilities (including contingent liabilities), properties, business or financial condition, results or prospects of the Target Companies taken as a whole;

Material Contract shall have the meaning set out in paragraph 9 of Schedule 4 (Seller Warranties);

Meat Free Substitute Technology Licence means the technology licence relating to the production process for meat-free substitutes (bacon and ham) from Kerry Luxembourg SARL to Northfield Foods Limited to be entered into in the agreed form;

Net Debt Statement means the statement of Actual Net Indebtedness at Completion prepared in accordance with the provisions of Schedule 9 (*Completion Statements*) and in the form set out in Part C of Schedule 9 (*Completion Statements*);

Newly-Trading Companies means the companies specified in Part B of Schedule 2 (*Target Companies*) and **Newly-Trading Company** means any of them;

Noon Products means Noon Products Limited (company registration number 02166664), further details of which are set out in Part A of Schedule 2 (*Target Companies*);

Oakhouse means Oakhouse Foods Limited (company registration number 04226390), further details of which are set out in Part A of Schedule 2 (*Target Companies*);

Occupied Lands has the meaning given to it in clause 21.1;

Occupied Real Estate has the meaning given to it in paragraph 10.12 of Schedule 4;

Owned Real Estate means the freehold property described in Part A of Schedule 3 (Properties);

Phase 2 CMA Reference means a reference by the United Kingdom Competition and Markets Authority to its chair for the constitution of a group under Schedule 4 to the Enterprise and Regulatory Reform Act 2013;

PHI Arrangements means the ROI PHI Arrangement and UK income protection policy with Canada Life;

Planning Acts means:

- (a) the Planning and Development Acts 2000-2021, the Local Government (Sanitary Services) Act 1878-2011, the Building Control Acts 1999-2014, the Fire Service Acts 1981 and 2003, and the Safety Health and Welfare at Work Acts 2005;
- (b) the Planning (Northern Ireland) Order 1991 (as amended) (where applicable) and the Planning Act (Northern Ireland) 2011, the Building Regulations (Northern Ireland) Order 1979 (as amended) (the Building Control Acts), the Fire Safety Regulations (Northern Ireland) 2010 (as amended), and the Health and Safety at Work (Northern Ireland) Order 1978 (as amended); and

(c) the Town and Country Planning Act 1990, the Planning (Listed Building and Conservation Areas) Act 1990, the Planning (Hazardous Substances) Act 1990, the Planning (Consequential Provisions) Act 1990, the Planning and Compensation Act 1991, the Planning and Compulsory Purchase Act 2004, the Planning Act 2008, the Localism Act 2011, the Growth and Infrastructure Act 2013, the Housing and Planning Act 2016, the Neighbourhood Planning Act 2017,

together with any extension, amendment, modification or re-enactment of any such acts, regulations and orders, and any rules, regulations, orders or instruments under any of the said Acts (before or after the date of this Agreement) and for the time being in force together with any lawful requirements of any competent authority relating to building and construction compliance;

Portadown Property means the property at Suite 6, The Exchange, Goban Street, Portadown, Co. Armagh, to the extent used prior to the date of this Agreement in relation to the Business;

Pre-Completion Undertakings means the undertakings as detailed in Schedule 7 (*Pre-Completion Undertakings*);

Properties means the Owned Real Estate and the Leased Real Estate;

Property Transfer has the meaning given to it in the relevant Hive Down Agreement;

Proposed Transaction has the meaning given to it in clause 3.1(a)(i);

Purchase Price means an amount equal to:

- (a) the Headline Price;
- (b) minus the Actual Net Indebtedness:
- (c) minus an amount equal to the difference between the Actual Working Capital and the Estimated Working Capital (if the Actual Working Capital is less than the Estimated Working Capital) or adding an amount equal to the difference between the Actual Working Capital and the Estimated Working Capital (if the Actual Working Capital is greater than the Estimated Working Capital);

Q&A Responses means the replies to pre-contractual enquiries raised by the Buyer and/or its Representatives and contained in the Data Room;

Raw Material Supply Agreement means the supply agreement relating to the provisions of food ingredients by the Retained Kerry Group to the Business to be entered into between Kerry Ingredients & Flavours Limited and Northfield Foods Limited in the agreed form;

Related Party Agreements means all agreements, contracts, understandings or undertakings entered into in connection with the Business (whether in writing or orally) with a member of the Seller's Group, including without limitation the Agency Agreements, but excluding the Transaction Documents and the Inter-Company Payables, excluding the Continuing Related Party Agreements;

Related Undertaking in relation to any company means any subsidiary or holding company of that company or any subsidiary of any such holding company, in each case from time to time;

Relevant Claim means any claim, demand, action, proceeding or suit by the Buyer under or in connection with this Agreement and/or the Tax Covenant, whether for damages, compensation or any other relief, other than a Restrictive Covenant Claim;

Relevant Named Contract means the contracts defined in the Hive Down Agreements as 'Named Contracts' which are not assignable without obtaining the consent of the relevant counterparty;

Relevant Period from (and including) 1 January 2019 to the date of this Agreement;

Relief means any loss, relief, allowance, exemption, set-off, deduction, credit, repayment or other relief relating to any Tax or to the computation of income, profits or gains for the purposes of any Tax;

Remedial Works means the investigation, removal, cure, containment, neutralisation, clean-up or remediation, including use restrictions and natural attenuations, of any contamination or hazardous substances, required to achieve compliance with requirements under Environmental Laws;

Representatives means, in relation to any person, its directors, officers, employees, agents, professional advisers, accountants and consultants;

Restricted Products means products of the type being manufactured and sold by the Business at Completion;

Restrictive Covenant Claim means any claim, demand, action, proceeding or suit by the Buyer in connection with a breach of clause 17.1 or 17.2;

Retained Business means the businesses of the Seller's Group from time to time, other than the Business;

Retained Kerry Group means the Seller's Group other than the Target Companies.

ROI means Republic of Ireland;

ROI DB Scheme means the Kerry Group plc Defined Benefit Plan (registered with the Pensions Authority under number PB2678);

ROI DC Plan means that Kerry Group Plc Defined Contribution Pension Plan;

ROI Pension Schemes means each and all of the ROI DC Plan and the ROI DB Scheme;

ROI PHI Arrangement means the disability benefits arrangement insured with Irish Life providing disability benefits to Employees;

ROI Sites means the sites and facilities operated by Target Companies at (a) Carrickmacross Industrial Estate, Castleblayney Road, Carrickmacross, Co. Monaghan, Ireland; and (b) Coolattin, Shillelagh, Co. Wicklow, Ireland, and "each of the ROI Sites" shall be construed accordingly;

Rollover means Rollover Limited (company registration number 02606878), further details of which are set out in Part A of Schedule 2 (*Target Companies*);

Sale Shares means the entire issued share capital of each of the Target Companies set out opposite the names of the Designated Sellers in column (2) of the table contained in Schedule 1 (Designated Sellers, Designated Buyers and Apportionment of Purchase Price);

Section 357 Guarantees means the guarantees provided by the Seller in respect of Duffy Meats Limited, Rye Valley Foods Limited and Consumer Foods Van Sales (Ireland) Limited, in each case in respect of the financial year ending 31 December 2020, under section 357 of CA 2014;

Seller Obligation means any representation, warranty, covenant to pay or undertaking to indemnify given by the Seller to the Buyer under this Agreement or the Tax Covenant;

Seller Warranties means the warranties set out in Schedule 4 and Seller Warranty means any one of them;

Seller Warranty Claim means any claim, demand, action, proceeding or suit by the Buyer under or in connection with any of the Seller Warranties;

Seller's Assignee has the meaning given to it in clause 35.2(a);

Seller's Bank Account means the account of the Seller as notified to the Buyer in writing from time to time;

Seller's Group means the Seller and each company which is for the time being (whether on or after the date of this Agreement) a Related Undertaking of the Seller, other than after Completion the Target Companies;

Seller's Group Insurance Policies has the meaning given to it in clause 17.1;

Senior Employees means those Employees who are employed by the Seller's Group at Managing Director level or at the level immediately below Managing Director level;

Set of Sale Shares means, in relation to a Designated Seller, the shares comprising issued share capital of any particular Target Company which are to be sold by that Designated Seller under this Agreement;

Shared Agreement means any Customer Agreement or Supply Agreement that relates to both the Business and the retained business of the Seller's Group;

Standard Transaction Agency Agreements means the agreements between the Seller and a Designated Seller or between the Buyer and a Designated Buyer entered into prior to the execution of this Agreement, appointing the Seller or the Buyer as the Designated Seller's or Designated Buyer's agent respectively on the terms set out therein, and Standard Transaction Agency Agreement means any one of them;

Stock means the following, wherever situated, as at the Completion Date:

- (a) all finished products held by the Target Companies as trading stock for supply exclusively in the course of the Business;
- (b) all stocks of meat, ingredients and other raw materials held by the Target Companies exclusively in the course of the Business;
- all semi-produced or partly finished products which are in the course of production by the Target Companies into finished products for supply in the course of the Business; and
- (d) all stocks of packaging materials and of maintenance or other engineering parts, components and accessories in use or held by the Target Companies for use or supply exclusively in the course of the Business;

Stock Take has the meaning given to it in clause 4.4;

subsidiary means a subsidiary undertaking (as defined by section 1162 CA 2006) or a subsidiary (as defined by section 1159 CA 2006);

Supplemental Transaction Agency Agreements means the agreements between the Seller, the Buyer and either a Designated Seller or Designated Buyer to be entered into in the circumstances referred to in Part B of Schedule 10 (Agency provisions) after the date of this Agreement (but in any case prior to or at Completion), in substantially the same form as the Standard Transaction Agency Agreement and Supplemental Transaction Agency Agreement means any of them;

Supply Agreements means the agreements, contracts, engagements, arrangements and other commitments (whether in writing or orally) entered into by the Business or by the Seller's Group for use in the Business (including those which the Seller's Group transferred, or purported to transfer, to a Target Company under the Hive Down Agreements) in each case for the purchase of raw materials, goods and services from time to time, but excluding the Related Party Agreements;

Surviving Provisions means clause 22 (*Entire Agreement*), 24 (*Remedies and waivers*) to 33 (*Agency structure*) and 36 (*Rights of third parties*) to 40 (*Service of process*), Schedule 10 (*Agency provisions*) and Schedule 11 (*Definitions and interpretation*);

Target Companies means the Existing Companies and the Newly-Trading Companies and **Target Company** means any one of them;

Target Working Capital means £14,100,000, being the Parties' agreed target amount of Working Capital as at Completion;

Tax means:

- (a) all forms of tax, levy, duty, charge, impost, withholding or other amount whenever created
 or imposed and whether of the United Kingdom or elsewhere, payable to or imposed by
 any Tax Authority; and
- (b) all charges, interest, penalties and fines incidental or relating to any Tax falling within (a) above or which arise as a result of the failure to pay any Tax on the due date or to comply with any obligation relating to Tax;

Tax Authority means any revenue, customs, fiscal, governmental, statutory, state or provincial authority, body or person, whether of the United Kingdom or the Republic of Ireland or elsewhere;

Tax Claims means Tax Warranty Claims and Tax Covenant Claims;

Tax Covenant means the tax covenant, in the agreed form, to be entered into between the Seller and the Buyer;

Tax Covenant Claim means any claim, demand, action, proceeding or suit by the Buyer pursuant to clauses 2.1(a), 2.1(b) or 2.1(d) of the Tax Covenant;

Tax Warranties means the Seller Warranties contained in paragraphs 3, 5.3 and 5.4 (in each case, insofar as the matter relates to or is in connection with Tax) and paragraph 20 of Schedule 4 (*Seller Warranties*);

Tax Warranty Claim means any claim, demand, action, proceeding or suit by the Buyer under or in connection with any of the Tax Warranties;

TCA means the Taxes Consolidation Act 1997 of ROI:

Transaction Agency Agreements means the Standard Transaction Agency Agreements and the Supplemental Transaction Agency Agreements entered into in connection with this Agreement, and **Transaction Agency Agreement** means any one of them;

Transaction Costs has the meaning given to it in clause 4.14;

Transaction Documents means this Agreement, the Disclosure Letter, the IP Assignment Agreements, the Standard Transaction Agency Agreements, the Supplemental Transaction Agency Agreements and all documents expressed to be in the agreed form, and all other documents to be entered into pursuant to this Agreement or expressly designated by the Buyer and the Seller to be a Transaction Document;

Transfer Regulations means the Transfer of Undertakings (Protection of Employment) Regulations 2006 as amended and the European Communities (Protection of Employees on Transfer of Undertakings) Regulations 2003;

Transfer Taxes means stamp duty, stamp duty land tax and all other applicable transfer, stamp, documentary or similar taxes including any interest or penalties on any of the same;

Transitional Services Agreement means the agreement between the Seller and the UK Designated Buyer, in the agreed form, setting out certain services to be provided for a transitional period after Completion from the Seller to the UK Designated Buyer and the Target Companies;

UK DB Section means the defined benefit section of the Kerry Retirement Benefits Scheme;

UK DC Section means the defined contribution section of the Kerry Retirement Benefits Scheme;

UK Life Assurance Arrangement means the life cover provided for UK and NI employees via the Kerry Retirement Benefits Scheme for employees who are members of the UK DC Section

and via the Kerry Life Assurance Scheme for employees who are not members of the UK DC Section.

UK Pensions Regulator means the "Regulator" as defined in the Pensions Act 2004;

Valid VAT Invoice means an invoice prepared and issued in accordance with VATA or VATCA (as applicable);

Van Sales Distribution Agreement means the agreement between Kerry Foods Limited, Henry Denny & Sons (Ireland) Limited, Consumer Foods Van Sales Limited and Consumer Foods Van Sales (Ireland) Limited, in the agreed form, in relation to the distribution by the Buyer of certain products of the Retained Business for a transitional period after Completion;

VAT means the tax called Value-Added Tax chargeable, leviable and payable as provided in VATA or VATCA (as applicable) and any other tax of a similar nature;

VATA means the Value Added Tax Act 1994;

VATCA means the Value-Added Tax Consolidation Act 2010 of ROI;

Warranties means the Buyer Warranties and the Seller Warranties and Warranty means any one of them;

Working Capital means the aggregate working capital of the Target Companies, being for these purposes only those line items in respect of the relevant Target Companies as set out in Part D of Schedule 9 (Completion Statements) (which, for the avoidance of doubt, includes all Stock, prepayments, trade debtors, trade creditors and accruals) and determined in accordance with specific polices and accounting treatments set out in Part B of Schedule 9 (Completion Statements), but excluding, for the avoidance of doubt, any item or amount to the extent it is taken into account in calculating the Actual Net Indebtedness; and

Working Capital Statement means the statement of the Actual Working Capital at Completion prepared in accordance with the provisions of Schedule 9 (Completion Statements) and in the form set out in Part D of Schedule 9 (Completion Statements).

- 2 In this Agreement unless the context requires otherwise:
 - a document expressed to be in the agreed form means a document in a form which has been agreed in writing by the Parties (or on their behalf) on or before the execution of this Agreement;

- (b) the table of contents and the headings are inserted for convenience only and do not affect the interpretation of this Agreement;
- (c) references to clauses and Schedules are to clauses of, and schedules to, this Agreement, references to this Agreement include its Schedules and references to a Part or paragraph are to a Part or paragraph of a Schedule to this Agreement;
- (d) references to this Agreement or any other document or to any specified provision of this Agreement or any other document are to this Agreement, that document or that provision as from time to time amended in accordance with the terms of this Agreement or that document or, as the case may be, with the agreement of the relevant parties;
- (e) words importing the singular include the plural and vice versa, words importing a gender include every gender;
- (f) references to a **person** include an individual, corporation, partnership, any unincorporated body of persons and any government entity;
- (g) references to any English legal term for any action, remedy, method of judicial proceeding, legal document, legal status, Court, official or any legal concept or thing shall in respect of any jurisdiction other than England be deemed to include what most nearly approximates in that jurisdiction to the English legal term;
- (h) references to time are to London time;
- references to **sterling** or £ or **GBP** are to the lawful currency from time to time of the United Kingdom;
- (j) references to making or calculating any payment to a person in respect of any Losses shall be construed as requiring that the calculation of the amount payable shall take into account the amount of any relief, credit or allowance available for tax purposes to the person suffering such Losses and any reduction of tax otherwise payable by such person which, in each case, arises as a result of the matter giving rise to such payment;
- (k) the rule known as the ejusdem generis rule shall not apply, and accordingly words introduced by words and phrases such as include, including, other and in particular shall not be given a restrictive meaning or limit the generality of any preceding words or be construed as being limited to the same class as the preceding words where a wider construction is possible;

- (I) the word company shall be deemed to include any partnership, undertaking or other body of persons, whether incorporated or not incorporated and whether now existing or formed after the date of this Agreement;
- (m) reference to a person having control of another person, or being controlled by another person, or being under common control with another person shall be construed as referring to control within the meaning of any of sections 450 and 707 CTA 2010 and sections 995(1) to 995(3) (inclusive) ITA 2007; and
- (n) the phrase "to the extent" shall mean, "if, but only to the extent".
- In this Agreement, unless the context requires otherwise, a reference to any statute or statutory provision (whether of the United Kingdom or elsewhere) includes:
 - any subordinate legislation (as defined by section 21(1) Interpretation Act 1978) made under it; and
 - (b) any provision which it has superseded or re-enacted (with or without modification), and any provision superseding it or re-enacting it (with or without modification), before or on the date of this Agreement, or after the date of this Agreement except to the extent that the liability of any Party is thereby increased or extended,

and any such statute, statutory provision or subordinate legislation as is in force at the date of this Agreement shall be interpreted at the date of this Agreement and no account shall be taken of any change in the interpretation of any of the foregoing by any court of law or tribunal made after the date of this Agreement.

EXECUTED by)		
as			
for and on behalf of)		
KERRY GROUP PLC)		
EXECUTED by)		
as			
for and on behalf of)		
PILGRIM'S PRIDE CORPORATION)		
)		
EXECUTED by			
as			
for and on behalf of)		
ONIX INVESTMENTS UK LIMITED)		
EXECUTED by)		
as			
for and on behalf of)		
ARKOSE INVESTMENTS ULC)		

EXHIBIT 31.1 CERTIFICATION BY PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Fabio Sandri, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q for the fiscal quarter ended June 27, 2021, of Pilgrim's Pride Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- . The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 28, 2021

/s/ Fabio Sandri Fabio Sandri

Principal Executive Officer

EXHIBIT 31.2 CERTIFICATION BY PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Matthew Galvanoni, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q for the fiscal quarter ended June 27, 2021, of Pilgrim's Pride Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a. Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c. Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
 - The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 28, 2021

/s/ Matthew Galvanoni Matthew Galvanoni Principal Financial Officer

EXHIBIT 32.1 CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. § 1350 ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), the undersigned officer of Pilgrim's Pride Corporation (the "Company"), does hereby certify, to such officer's knowledge, that:

The quarterly report on Form 10-Q for the quarter ended June 27, 2021 (the "Form 10-Q") of the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934, and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company. Date: July 28, 2021 /s/ Fabio Sandri

Fabio Sandri

Principal Executive Officer

EXHIBIT 32.2 CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. § 1350 ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

Pursuant to section 906 of the Sarbanes-Oxley Act of 2002 (subsections (a) and (b) of section 1350, chapter 63 of title 18, United States Code), the undersigned officer of Pilgrim's Pride Corporation (the "Company"), does hereby certify, to such officer's knowledge, that:

The quarterly report on Form 10-Q for the quarter ended June 27, 2021 (the "Form 10-Q") of the Company fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934, and information contained in the Form 10-Q fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 28, 2021 /s/ Matthew Galvanoni

Matthew Galvanoni Principal Financial Officer